University of Alberta

Studies on the mechanisms of immune evasion in *Trypanosoma carassii* infections of the goldfish (*Carassius auratus* L.)

by

Ayoola Oladiran

A thesis submitted to the Faculty of Graduate Studies and Research in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

in

Physiology, Cell and Developmental Biology

Department of Biological Sciences

©Ayoola Oladiran

Spring 2012

Edmonton, Alberta

Permission is hereby granted to the University of Alberta Libraries to reproduce single copies of this thesis and to lend or sell such copies for private, scholarly or scientific research purposes only. Where the thesis is converted to, or otherwise made available in digital form, the University of Alberta will advise potential users of the thesis of these terms.

The author reserves all other publication and other rights in association with the copyright in the thesis and, except as herein before provided, neither the thesis nor any substantial portion thereof may be printed or otherwise reproduced in any material form whatsoever without the author's prior written permission.

ABSTRACT

Parasites possess variety of mechanisms to modulate or evade host defence systems to maintain chronic infection and ensure their transmission. The protozoan parasite *Trypanosoma carassii* is infective to a number of freshwater fish species and can cause significant mortality in aquaculture. *T. carassii* shares similarities with both *T. brucei*, the causative agent of sleeping sickness and *T. cruzi*, which causes Chagas' disease. It is not known how *T. carassii* escapes host immune responses.

My doctoral research focused on the interactions between *T. carassii* and its fish host with emphasis on the strategies used by the parasite to evade fish immune host defense.. and establish long lasting infections. I identified at the molecular level the key antigens of *T. carassii* present in excretory/secretory (ES) products and surface protein fractions of the parasite. As a first step toward understanding the importance of select antigens of *T. carassii*, I produced the recombinant proteins and characterized their roles in immune evasion.

Like heat shock protein 70 of other pathogens, I found that *T. carassii* hsp70 was immunogenic and was present in both ES products and surface protein fraction of the parasite. Recombinant parasite hsp70 significantly increased expression of pro-inflammatory genes and enhanced inflammatory response of goldfish macrophages. In contrast, another parasite surface molecule, glycoprotein 63 (Gp63) down-regulated both pathogen and cytokine-induced inflammatory responses of goldfish monocytes and macrophages. Parasite gp63 was associated with macrophages and appeared to interfere with signalling mechanisms. Since

complement-mediated lysis is one of the main host defence responses against trypanosomes, I cloned and characterized parasite surface molecule called calreticulin. Recombinant *T. carassii* calreticulin bound to first component of complement, C1q, of not only goldfish but also humans. Further, recombinant *T. carassii* calreticulin inhibited C1q-dependent hemolysis.

T. carassii infection of goldfish induced increased expression of pro- and anti-inflammatory cytokines. Increased cytokine mRNA levels were observed during the acute phase of infection, and then they returned to normal levels or were down-regulated during the elimination phase of the infection. These findings demonstrate that parasite surface molecules and those found in ES fraction have the capacity to manipulate host inflammatory and antimicrobial responses, thereby ensuring persistence of *T. carassii* in its host.

ACKNOWLEDGEMENTS

I cannot find words to express my appreciation and gratitude to my supervisor Dr. Miodrag Belosevic, for the guidance, supervision and support I received during my course of study. I am honoured to have worked under your supervision; I know I have improved tremendously as a scientist and as an individual. Your words of encouragement and constructive criticisms provided the inspiration for completion of this thesis.

My profound appreciation goes to my PhD supervisory committee members Dr. John Chang and Dr Catherine Field, who despite their busy schedule found time to meet with me. I thank you for your thoughtful advice and encouragement.

I am greatly indebted to my parents Mr and Mrs Oladiran, my brothers Busayo and Ayorinde, and my sister Bola for their unflinching moral support, understanding and encouragement. Dad and Mum, I am particularly grateful to you for laying the foundation for my education. Thank you for the advice and discipline instilled in me. To all other family members, I am very grateful for your love and support.

To my other parents Elder and Mrs Ake-johnson, you have been very supportive of me. I am grateful for your unwavering love, patience and understanding.

To my beautiful wife Oluwatoyin, words are not enough to describe how grateful I am to you for being there always. Your love, patience, understanding and support kept me going. I am blessed to have you by my side. Thank you very much for encouraging me and for accepting to go on this journey with me. To my PhD candidacy and defense members Dr. Ted Allison, Dr. Reuben Kaufman, Dr Norm Neuman, Dr. James Stafford and Dr. Andre Buret; thank you for your guidance and support. Special thanks to Dr. A. Shostak for his advice and for helping me improve on my statistical ability.

To my colleagues Barbara Katzenback, David Beauparlant, Dr. Erick Garcia, Jonathan Pun, Jordan Hodgkinson, Dr. Leon Grayfer, Mariel Hagen, Dr. Rebecca Baldwin, Steve Hitchen, thank you for being part of my academic experience and for making my stay here a memorable one. I have learnt from each of you and I would always remember the time we shared together.

To Dr. Daniel Barreda, Dr. James Stafford, Dr. Patrick Hanington, other past and present members of the Belosevic Lab, you have motivated and inspired me in different ways. Thank you for your encouragement and support always.

To my friends in Nigeria and Canada too numerous to mention, I am truly thankful for your continued friendship. Ayo Odebunmi, Ayo Olajide, Abigail Adebusuyi, Emmanuel Pila, Collins Obiosa, Idowu Ohioze, Joseph Ayeni, Kingsley Ezeagwula, Sabinus Okafor, Samuel Oladapo and Seyi Akande deserve special mention. Thank you all for your friendship, encouragement and support.

To the staff of biological sciences aquatic facility, animal facility, MBSU and administrative offices, I am exceedingly grateful for your assistance and support. The work described in this thesis would not have been possible without your help.

Finally, I would like to acknowledge the financial support provided by Natural Sciences and Engineering Research Council, NSERC (grant to Dr. Belosevic), Department of Biological Sciences (Teaching and Research Assistantship), Faculty of Graduate Studies and Research, FGSR (Provost Doctoral Scholarship and Queen Elizabeth Doctoral Scholarship).

Thank you all for making a doctor out of me!

TABLE OF CONTENTS

	Page
CHAPTER 1: GENERAL INTRODUCTION AND OBJECTIVES	1
1.0. General introduction	1
1.1. Objectives and specific aims of the thesis	5
1.2. References	7
CHAPTER 2: LITERATURE REVIEW	9
2.0. Introduction	9
2.1. African trypanosomes and tsetse fly	13
2.1.1. Tsetse fly defense against trypanosomes	13
2.1.2. T. brucei development in the tsetse fly	15
2.2. T. cruzi and its insect vectors	18
2.2.1. T. cruzi interaction with triatomine vectors	18
2.2.2. T. cruzi development in triatomines	20
2.3. T. carassii and the leech vector	23
2.3.1. T. carassii development in the leech vector	23
2.4. Immune evasion strategies of trypanosomes in mammalian hosts	25
2.4.1. Trypanosome-induced immunosuppression	25
2.5. Immune evasion mechanisms of <i>T. brucei</i>	28
2.5.1. The variant surface glycoprotein	28
2.5.2. Modulation of host immunity	30
2.5.3. Modulation of host cytokine gene expression	31
2.5.4. Evasion of complement-mediated lysis	32
2.6. Immune evasion of <i>T. cruzi</i>	33
2.6.1. Role of mucin glycoproteins	33

2.6.2. Complement regulatory proteins	34
2.6.3. Gp63 proteases	37
2.6.4. Interference with cytokines and phagocytic cells	38
2.7. Immune evasion mechanisms of T. carassii	38
2.7.1. T. carassii infection of goldfish	38
2.7.2. Evasion of humoral immunity	39
2.7.3. The search for complement regulatory proteins in T. carassii	40
2.7.4. Modulation of host immunity	41
2.7.5. Possible role of proteases	43
2.8. Summary	44
2.9. References	47
CHAPTER 3: MATERIALS AND METHODS	75
3.0. General procedures	75
3.1. Fish and parasite culture conditions	75
3.1.1. Fish	75
3.1.2. Fish blood and serum preparation	75
3.1.3. Parasites	76
3.1.4. Purification of parasites from blood	76
3.1.5. Infection of fish	77
3.1.6. Assessment of infection	77
3.1.7. Preparation of parasite antigens	78
3.1.7.1. Excretory/secretory (ES) products	78
3.1.7.2. Whole cell lysates	78
3.1.8. Detergent solubilization of membrane proteins	79
3.1.9. Anti-carp IgM	79

3.2. Goldfish kidney-derived monocyte/macrophage cultures	80
3.2.1. Culture medium	80
3.2.2. Isolation of goldfish kidney leukocytes	80
3.2.3. Preparation of cell-conditioned medium (CCM)	81
3.3. Molecular procedures	81
3.3.1. Isolation of total RNA	81
3.3.2. Cloning of T. carassii genes	82
3.3.2.1. T. carassii hsp70 (Tcahsp70)	82
3.3.2.2. T. carassii calreticulin (TcaCRT)	82
3.3.2.3. T. carassii glycoprotein 63 (Tcagp63)	82
3.3.2.4. In silico analysis of T. carassii genes	83
3.4. Using prokaryotic expression system for production of <i>T. carassii</i>	
recombinant proteins	84
3.4.1. Cloning of T. carassii genes into pET SUMO expression vector	rs 84
3.4.2. Pilot expression studies of recombinant Tcahsp70, TcaCRT,	
Tcagp63	84
3.4.3. Scaled up expression of recombinant Tcahsp70 and TcaCRT	
in E. coli	85
3.4.4. Scaled up expression of recombinant Tcagp63 in E.coli	87
3.5. Quantitative expression analysis (Q-PCR) of goldfish immune genes	88
3.5.1. Q-PCR of monocytes and macrophages genes	88
3.5.1.1. Pro-inflammatory cytokine expression in goldfish	
macrophages treated with recombinant Tcahsp70	89
3.5.1.2. Immune gene expression in macrophages exposed to	
rTcagp63 and A. salmonicida	90

3.5.1.3. Immune gene expression in macrophages treated with	
rTcagp63	91
3.5.2. Quantitative expression analysis of immune genes in goldfish	
tissues	92
3.6. Measurement of the production of reactive oxygen and nitrogen	
intermediates: nitric oxide and respiratory burst assays	92
3.7. Functional analysis of the effects of <i>T. carassii</i> proteins on antimicrobial	
responses of goldfish monocytes and macrophages	94
3.7.1. Assessment of the ability of rTcahsp70 to induce nitric oxide	
response of goldfish macrophages	94
3.7.2. Assessment of the effects of rTcagp63 on A. salmonicida-	
induced nitric oxide response of goldfish macrophages	95
3.7.3. Assessment of the effects of rTcagp63 on A. salmonicida	
induced respiratory burst response of goldfish monocytes	96
3.8. SDS-PAGE and western blot	96
3.8.1. SDS-PAGE	96
3.8.2. Western blot	97
3.8.2.1. Analysis of excretory/secretory proteins using SDS-	
PAGE and western blot	97
3.8.2.2. SDS-PAGE and western blot analysis of rTcahsp70	
rTcaCRT and rTcagp63	99
3.8.2.3. SDS-PAGE and western blot analysis of carp and	
goldfish first complement component C1q	100
3.8.2.4. Far western blot	101
3.9. Investigation of the effects of rTcagp63 on goldfish macrophages	102

3.9.1. The interaction between rTcagp63 and goldfish macrophages	102
3.9.2. Assessment of the effects of rTcagp63 on phospho-tyrosine	
protein pattern of goldfish macrophages	103
3.10. Production and purification of polyclonal rabbit anti-recombinant	
Tcahsp70, TcaCRT and Tcagp63	104
3.10.1. Production of polyclonal rabbit anti-recombinant Tcahsp70,	
TcaCRT and rTcagp63 antibodies	104
3.10.2. Affinity purification of anti-recombinant Tcahsp70,	
TcaCRT and Tcagp63 IgG from rabbit serum	104
3.11. Functional assessment of TcaCRT and investigation of its interaction	
with goldfish complement component C1q	105
3.11.1. Assessment of interaction of goldfish complement	
component C1q with immobilized rTcaCRT using affinity	
chromatography	105
3.11.2. Assessment of interaction of goldfish complement	
component C1q and rTcaCRT using co-immunoprecipitation	106
3.11.3. Assessment of binding of human complement component	
C1q to immobilized rTcaCRT	106
3.11.4. Assessment of the effects of rTcaCRT on classical	
complement pathway-mediated haemolysis	107
3.11.5. Assessment of surface localization of TcaCRT using	
immunofluorescence and confocal microscopy	108
3.12. Proteomic analysis of <i>T. carassii</i> surface antigens	109
3.12.1. Two-dimensional gel electrophoresis (2-DE) of T. carassii	
Surface protein fraction	109

3.12.2. Mass spectrometry analysis	110
3.13. Idnetification of <i>T. carassii</i> genes up-regulated in vivo using	
suppression subtraction subtraction hybridization	111
3.13.1 In silico analysis of differentially expressed gene fragments	
identified in suppression subtraction hybridization	112
3.14. Statistical analysis	113
3.15. References	124
CHAPTER 4: THE COURSE OF INFECTION IN Trypanosoma carassii-	
INFECTED GOLDFISH AND EXPRESSION ANALYSIS OF	
INFLAMMATORY AND ANTIMICROBIAL GENES IN THE GOLDFISH	
(Carassius auratus L.) INFECTED WITH Trypanosoma carassii	127
4.0. Introduction	127
4.1. Experimental design	130
4.1.1. Assessment of T. carassii growth in vitro	130
4.1.2. Course of infection in goldfish infected with T. carassii	130
4.1.3. Assessment of the effects of <i>T. carassii</i> infection on expression	1
of inflammatory and antimicrobial genes of goldfish	131
4.2. Results	132
4.2.1. The course of <i>T. carassii</i> infection in the goldfish	132
4.2.2. T. carassii induces differential cytokine gene expression	
during the acute and elimination phases of the infection in	
the goldfish	132
4.2.3. Infection of goldfish with T. carassii induces delayed	
expression of iNOS genes	134
4.3. Discussion	135

4.4. References		148
CHAPTER 5: IDENTIFICATIO	N OF HEAT SHOCK PROTEINS AND	
OTHER ANTIGENS IN EXCR	ETOTY/SECRETORY PRODUCTS AND	
SURFACE PROTEIN FRACTIO	ON FROM Trypanosoma carassii	155
5.0. Introduction		155
5.1. Experimental design		158
5.1.1. Proteomic analysi	s of ES products and surface protein fraction	158
5.1.2. Molecular cloning	g and expression of T. carassii hsp70	159
5.1.3. Assessment of eff	ects of recombinant T. carassii hsp70 on	
cytokine gene exp	pression by goldfish macrophages	159
5.1.4. Assessment of eff	ects of recombinant T. carassii hsp70 on	
antimicrobial resp	onse of goldfish macrophages	160
5.2. Results		161
5.2.1. Identification of 7	. carassii antigens in ES products	161
5.2.2. Proteomic analysi	s of T. carassii surface protein fraction	161
5.2.3. <i>T. carassii</i> hsp70	(Tcahsp70) is immunogenic	162
5.2.4. <i>T. carassii</i> hsp70	is present in surface protein fraction	163
5.2.5. Recombinant para	site hsp70 induces expression of pro-	
inflammatory gen	es in goldfish macrophages	163
5.2.6. Recombinant para	site hsp70 induces nitric oxide response of	
goldfish macroph	ages	164
5.3. Discussion		165
5.4. References		180

CHAPTER 6: GLYCOPROTEIN 63 (GP63) OF Trypanosoma carassii

SUPPRESSES ANTIMICROBIAL RESPONSES OF GOLDFISH

(Carass	ius auratus L.)	186
6.0. Intr	oduction	186
6.1. Exp	perimental design	189
	6.1.1. Molecular cloning and expression of Tcagp63	189
	6.1.2. Assessment of the effect of Tcagp63 on nitric oxide response	
	of goldfish macrophages	189
	6.1.3. Assessment of the effect of rTcagp63 on respiratory burst	
	response of goldfish monocytes	190
	6.1.4. Assessment of effect of rTcagp63 on immune gene expression	
	of goldfish macrophages activated with A. salmoncida	191
	6.1.5. Assessment of effect of rTcagp63 on immune gene expression	
	of goldfish macrophages	192
	6.1.6. Assessment of interaction of rTcagp63 with goldfish	
	macrophages	192
6.2. Res	sults	193
	6.2.1. Tcagp63 is similar to gp63 of other trypanosomes	193
	6.2.2. rTcagp63 is unstable and undergoes rapid degradation	194
	6.2.3. rTcagp63 abrogated A. salmonicida-induced nitric oxide	
	response of goldfish macrophages	195
	6.2.4. rTcagp63 abrogated A. salmonicida-induced respiratory	
	burst response of goldfish monocytes	196
	6.2.5. rTcagp63 exerts its effects partly through entry into immune	
	cells	196

6.2.6. Pre-incubation of cells with rTcag63 altered immune gene	
expression	197
6.2.7. rTcagp63 bound and was internalised by goldfish macrophages	3
and altered phosphotyrosine protein patterns of macrophages	198
6.3. Discussion	199
6.4. References	220
CHAPTER 7: Trypanosoma carassii CALRETICULIN BINDS HOST	
COMPLEMENT COMPONENT C1q AND INHIBITS CLASSICAL	
COMPLEMENT PATHWAY-MEDIATED LYSIS	228
7.0. Introduction	228
7.1. Experimental design	231
7.1.1. Molecular cloning of <i>T. carassii</i> calreticulin	231
7.1.2. Assessment of TcaCRT localization using immunofluorescence	е
and confocal microscopy	231
7.1.3. Examination of interaction of goldfish complement component	-
C1q with immobilized rTcaCRT	232
7.1.4. Haemolysis assays	234
7.2. Results	234
7.2.1. TcaCRT is present in membrane protein fraction and may be	
secreted	234
7.2.2. TcaCRT is expressed on parasite surface	235
7.2.3. Anti-carp C1q-A antibody cross-reacts with goldfish C1q	236
7.2.4. rTcaCRT binds goldfish C1q in vitro	236
7.2.5. rTcaCRT binds human C1q	238
7.2.6. rTcaCRT inhibits C1q-dependent complement lysis	238

7.3. Discussion	239
7.4. References	256
CHAPTER 8: GENERAL DISCUSSION	261
8.0. Evasion of host immune responses by parasites	261
8.1. Future direction	270
8.2. Proposed model of <i>T. carassii</i> evasion mechanisms	275
8.3. Summary	276
8.4. References	279

LIST OF TABLES

	Page
Table 3.1. Components of Trypanosoma carassii TDL-15 Medium	115
Table 3.2. Components of MGFL-15 Medium	116
Table 3.3. Primers used for molecular cloning of Trypanosoma carassii	
genes (3'- 5')	117
Table 3.4. Quantitative real time primer sequences	119
Table 3.5. <i>Trypanosoma carassii</i> antigens identified in ES products by MS	121
Table 3.6. Trypanosoma carassii proteins identified in surface protein	
fraction by MS	122
Table 3.7. Differentially expressed Trypanosoma carassii genes identified	
by SSH	123
Table 4.1. Parasitemia in goldfish infected with Trypanosoma carassii	141

LIST OF FIGURES

	Page
Figure 2.1. The complement system	46
Figure 4.1. Image of <i>in vitro</i> cultivated <i>T. carrasii</i>	140
Figure 4.2. Growth of <i>T. carassii in vitro</i> in medium supplemented with	
10% heat-inactivated goldfish serum	142
Figure 4.3. Quantitative expression analysis of goldfish immune genes in	
the kidney of T. carassii-infected and non-infected fish	143
Figure 4.4. Quantitative expression analysis of goldfish immune	
genes in the spleen of T. carassii-infected and non-infected fish	144
Figure 4.5. Quantitative expression analysis of goldfish immune genes in	
the liver of T. carassii-infected and non-infected fish	145
Figure 4.6. Quantitative expression analysis of goldfish iNOS genes in	
T. carassii-infected and non-infected fish	146
Figure 4.7. Quantitative expression analysis of goldfish NADPH oxidase	
component genes in kidney of T. carassii-infected (day 21 pi)	
and non-infected fish	147
Figure 5.1. Two-dimensional gel electrophoresis of <i>T. carassii</i> surface	
proteins	169
Figure 5.2. Immunogenicity of T. carassii hsp70	170
Figure 5.3. Amino acid alignment of <i>T. carassii</i> hsp70 sequence and those	
from other trypanosomatids (T. cruzi, T. brucei and L. major) as	
well as sequences from mammalian (Homo sapiens) and non-	
mammalian (Carassius auratus and Cyprinus carpio) vertebrates	171
Figure 5.4. Schematic of protein structure of T. carassii hsp70	172

Figure 5.5. <i>T. carassii</i> hsp70 is present in membrane surface protein fraction	173
Figure 5.6. Analysis of pro-inflammatory cytokine expression of goldfish	
macrophages treated with recombinant T. carassii hsp70	174
Figure 5.7. Analysis of chemokine gene expression in goldfish macrophages	
treated with rTcahsp70	175
Figure 5.8. Analysis of immune gene expression in goldfish macrophages pre-	
treated with pronase before incubation with rTcahsp70	176
Figure 5.9. T. carassii hsp70 up-regulates iNOS gene expression of goldfish	
macrophages	177
Figure 5.10. T. carassii hsp70 enhances nitric oxide response of goldfish	
macrophages	178
Figure 5.11. Nitric oxide response of goldfish macrophages is due to	
Tcahsp70	179
Figure 6.1. Amino acid alignment of <i>T. carassii</i> gp63 (Tcagp63) sequence	
and those from T. cruzi, T. brucei and Leishmania major	205
Figure 6.2. Schematic of protein structure of T. carassii gp63	206
Figure 6.3. Phylogenetic analysis of the <i>T. carassii</i> gp63 demonstrated closest	
relationship to T. cruzi and T. brucei gp63 sequences	207
Figure 6.4. Western blot analysis of rTcagp63	208
Figure 6.5. Recombinant T. carassii gp63 abrogated A. salmonicida-induced	
NO response of goldfish macrophages	209
Figure 6.6. Abrogation of A. salmonicida-induced NO response of goldfish	
macrophages is due to rTcagp63	210
Figure 6.7. Recombinant T. carassii gp63 abrogated A. salmonicida-induced	
ROI response of goldfish macrophages	211

Figure 6.8.	Recombinant <i>T. carassii</i> gp63 abrogated rgTNFα2-induced ROI	
	response of goldfish monocytes	212
Figure 6.9.	Pre-incubation of cells with rTcagp63 and subsequent washing	
	failed to reverse A. salmonicida and rgTNF α 2-induced ROI	
	response of goldfish monocytes	213
Figure 6.10	. Analysis of immune gene expression of goldfish macrophages	
	pre-treated with rTcagp63 before stimulation with A. salmonicida	214
Figure 6.11	. Analysis of IL-10 and NADPH oxidase components gene	
	expression in goldfish macrophages and monocytes pre-treated	
	with rTcagp63 before stimulation with A. salmonicida	215
Figure 6.12	2. Analysis of immune gene expression of goldfish treated with	
	rTcagp63	216
Figure 6.13	8. Recombinant T. carassii gp63 was bound to goldfish	
	macrophages	217
Figure 6.14	. Recombinant T. carassii gp63 associates with goldfish	
	macrophages	218
Figure 6.15	. Recombinant T. carassii gp63 altered phospho-tyrosine	
	protein pattern of goldfish macrophages	219
Figure 7.1.	A representative gel of two-dimensional gel electrophoresis of	
	T. carassii membrane proteins	245
Figure 7.2.	Amino acid alignment of Trypanosoma carassii calreticulin	
	(TcaCRT) sequence and those from <i>T. cruzi</i> and <i>T. brucei</i>	246
Figure 7.3.	Schematic of protein structure of T. carassii CRT	247
Figure 7.4.	Rabbit anti-recombinant TcaCRT affinity purified rabbit IgG	
	polyclonal antibody recognizes TcaCRT in ES products and	

whole cell lysate of <i>T. carassii</i>	248
Figure 7.5. T. carassii CRT is expressed on the parasite surface	249
Figure 7.6. Rabbit anti-carp C1q-A antiserum recognizes goldfish C1q	250
Figure 7.7. Goldfish C1q binds immobilized rTcaCRT	251
Figure 7.8. The far western blot showing that goldfish C1q interacted with	
rTcaCRT	252
Figure 7.9. Goldfish C1q interacted with rTcaCRT	253
Figure 7.10. T. carassii CRT binds human C1q	254
Figure 7.11. Recombinant T. carassii CRT inhibited classical complement	
pathway-mediated haemolysis	255
Figure 8.1. Proposed mechanisms of T. carassii immune evasion.	278

LIST OF ABBREVIATIONS

ANOVA-analysis of variance

BCIP-5-bromo-4-chloro-3-indolyl phosphate

BSA-bovine serum albumin

cDNA-complementary DNA

CRP-complement regulatory proteins

DIC -differential interference contrast

DMSO-dimethyl sulfoxide

DNA -deoxyribonucleic acid

ES-excretory/secretory

FACS-fluorescence activated cell sorter

FCA-Freund's complete adjuvant

FIA-Freund's incomplete adjuvant

FITC-fluorescein isothiocyanate

GPI-glycosylphosphatidylinositol

HSP-heat shock protein

IFN-interferon

IgG-immunoglobulin G

IgM-immunoglobulin M

IL-interleukin

iNOS-inducible nitric oxide synthase

IP-intraperitoneal

IPTG-isopropyl-beta-D-galactopyranoside

kDa-kiloDalton

LAL-Limulus amebocyte lysate

LB-Luria-Bertani

LPS-lipopolysacharride

MAC-membrane attack complex

MS-mass spectrometry

MBL-mannose binding lectin

NBT-nitro blue tetrazolium

NFkB-nuclear factor kappa B

NO-nitric oxide

OD-optical density

PAMP-pathogen associated molecular pattern

PBS-phosphate buffered saline

PCR-polymerase chain reaction

PEG-polyethyleneglycol

PMSF-phenylmethylsulfonylchloride

Q-PCR-quantitative-polymerase chain reaction

RACE-rapid amplification of cDNA ends

RNA-ribonucleic acid

ROI-reactive oxygen intermediates

RT-PCR-reverse transcriptase polymerase chain reaction

SDS-PAGE-sodium dodecyl sulphate polyacrilamide gel electrophoresis

SEM-standard error of the mean

SRA-serum resistance-associated gene

SSH-selective subtractive hybridization

TBS-tris buffered saline

TGF-transforming growth factor

Th1/Th2-T-helper cell

TLTF-trypanosome lymphocyte triggering factor

TNF-tumor necrosis factor

TTBS-TBS+Tween 20

VSG-variable surface glycoprotein

VSP-variant surface protein

CHAPTER 1

GENERAL INTRODUCTION AND OBJECTIVES

1.0 General introduction

Protozoan parasites have evolved remarkable immune modulation and evasion strategies to subvert host defence systems. These strategies enable their survival and persistence in various hosts and ensure their transmission. Parasite immune evasion processes are in large part dependent on their ability to manipulate host molecular mechanisms that constitute and regulate the immune response or the normal functioning of host cells. The study of these immune evasion processes has furthered our understanding of the functioning of the immune system [15]. The different parasites use several mechanisms either simultaneously or sequentially at different stages of the infection. Also, parasites produce molecules deployed in different ways for the purpose of evading host responses.

Generally, parasites immune evasion mechanisms can be classified as passive evasion where parasites hide from immune system by adopting an intracellular existence or by invading immune privileged tissue [2]. Parasites can become invisible to the immune system for example, by shielding surface components following recognition [3], or change their surface identity during the course of infection as in the case of antigenic variation [17]. Also, parasites can become temporarily inactive (hypobiosis and/or encystation) so as to escape immune system. Active modulation and interference is a mechanism where parasite actively interfere with hosts immune responses especially the regulatory network such as cytokines responsible for generation of potent immune defence strategies. Parasites can also interfere with host cell functions by producing molecules that modulate immune response and cellular functions. By using a combination of the strategies mentioned above parasites can evade each step of the immune host defense.

Trypanosomes are digenetic protozoa that infect humans, domestic and wild animals. These parasites cause important medical and veterinary diseases making them a major public health concern. There are many species of trypanosomes and they infect hosts belonging to all vertebrate genera. Parasites also cycle between insect or leech vectors and vertebrate host where they undergo biochemical and morphological changes [10, 11, 16]. Trypanosomes have received attention in the last two decades because of the diseases they cause and their remarkable immune evasion mechanisms. The completed genome sequences of mammalian trypanosomes have revealed the extensive diversity of molecules that contribute to immune evasion by these parasites. The different species interact uniquely with their vertebrate hosts using an array of evasion strategies. Some of the most fascinating immune evasion mechanisms such as antigenic variation were first described in the trypanosomes.

African trypanosomes such as *T. brucei* have evolved an array of immune evasion strategies. The most important of these is switching of the variant surface glycoproteins (VSG) molecules. The VSGs are highly immunogenic and highly abundant molecules and their controlled expression allows parasites to keep one step ahead of host immune responses [6]. The *T. brucei* genome contains about

2

1000 different VSG genes and up to 20 VSG expression sites. An allelic exclusion mechanism ensures that only one VSG gene is expressed at any one time allowing the parasite to generate antigenically distinct glycoproteins and forcing the host to generate successive waves of antibodies [4]. Although the controlled expression of VSGs contributes to resistance of African trypanosomes to complement mediated lysis (CML), it is also believed that a major surface protease (MSP or GP63) plays a major role in parasite protection from CML [6, 7]. Additionally, these parasites secrete/excrete molecules known as trypanokines (cysteine proteases), that modulate host cytokine networks for parasites benefit.

A significant portion of the surface of *T. cruzi* is composed of mucin glycoproteins encoded by approximately 850 genes [5]. The mucin genes contain variable regions and their controlled expression protects the parasite from host immune responses. *T. cruzi* also expresses complement regulatory proteins (CRPs) on its surface. These molecules protect parasites from CML by restricting formation of active C3 convertase [12], and preventing assembly of membrane attack complex that kills parasites. Additionally, excreted/secreted CRPs may bind host complement molecules away from parasites surface.

There is increasing interest in fish parasitoses and defence mechanisms against these infections, because of their economic importance in aquaculture and fisheries. *Trypanosoma carassii* is thought to be a non-pathogenic trypanosome of fish in natural population. These parasites live in the vascular system of many economically important fish species. It is thought that the immunocompetent state of the host is responsible for their ability to control *T. carassii* infection, although

3

the prevalence of infection can be as high as 100% in aquaculture setting [9, 13]. In immunocompromised state, which can be caused by stress imposed by high population density, healthy fish can succumb to infection [13]. Mortality caused by *T. carassii* can be significant although it may not be due to high parasitemia but rather parasite-induced immunosuppression.

The goldfish-*T. carassii* model system represents a natural host-parasite system to dissect trypanosome evasion mechanisms. The investigation of the host-parasite interactions of the more studied trypanosomes *T. brucei* and *T. cruzi* are mostly carried out in mice that are not the natural hosts for these parasites. In this case, comparison of the results is usually difficult because different strains of hosts and parasites are used. The advantages of goldfish-*T. carassii* model system are as follows: (a) it is a natural host-parasite association which allows the study of protective immune responses of primitive ectothermic vertebrates to protozoan infection; (b) goldfish are hardy, relatively inexpensive and can be maintained easily; (c) *T. carassii* can be grown *in vitro* [16, 20]; and (d) goldfish can be readily infected with *T. carassii* [8, 18, 19].

This thesis focuses on some of the mechanisms used by *T. carassii* to evade goldfish immune responses. There are 8 chapters in this thesis. Chapter 2 provides a detailed review on trypanosome biology, interaction with their hosts and immune evasion strategies with emphasis on mammalian trypanosomes *T. brucei* and *T. cruzi*, and fish trypanosome *T. carassii*. Chapter 3 contains materials and methods used for the research described in this thesis while Chapter 4 focuses on the description of the course of infection in goldfish infected with *T*. *carassii*, and examination of changes in immune gene expression in *T. carassii*infected fish. Cloning of *T. carassii* heat shock protein 70 (Tcahsp70), the assessment of the effects of Tcahsp70 on goldfish immune cells, proteomics analysis of *T. carassii* surface proteins and excretory/secretory antigens were reported in Chapter 5. In Chapter 6, cloning of *T. carassii* glycoprotein 63 (Tcagp63), the assessment of the effects of Tcagp63 on goldfish monocyte and macrophages, and role of Th1/Th2 balance in trypanosome-induced immunosuppression were examined. Chapter 7 describes cloning of *T. carassii* calreticulin (TcaCRT), the assessment of TcaCRT effects on the function of classical pathway of complement as well the role of complement system in controlling trypanosome infection. General discussion of the parasite evasion mechanisms, the summary and implications of the results obtained and recommendations for future research are presented in Chapter 8.

1.1 Objectives and Specific Aims of the Thesis

The overall goal of my thesis research was to investigate the interactions between *T. carassii* and its fish host. The central objective of my thesis was to identify at the molecular level the key antigens of *T. carassii* and characterize the importance of these antigens to immune evasion strategies used by the parasite. Specific aims of my thesis were:

1. To investigate goldfish-*T. carassii* interaction by examining changes in the expression of goldfish immune genes during the course of infection.

- 2. To determine key antigens of *T. carrassii* in excretory/secretory products and surface protein extract using genomic and proteomic approaches.
- 3. To determine and clone the genes encoding these antigens and produce the recombinant protein for further characterization.
- 4. To functionally characterize the key antigens of *T. carrassii* and determine their contribution to evasion or modulation of goldfish immune responses.
- 5. To define the molecular mechanisms used by parasite to persist in the host by subverting or modulating host defence during the course of the infection.

1.2 References

- Barrett, M. P., R. J. Burchmore, A. Stich, J. O. Lazzari, A. C. Frasch, J. J. Cazzulo and S. Krishna. 2003. The trypanosomiases. Lancet **362**: 1469-1480.
- Bhopale, G. M. 2003. Pathogenesis of toxoplasmosis. Comparative Immunology, Microbiology and Infectious Diseases 26: 213–222.
- Bloom, B. R. 1979. Games parasites play: how parasites evade immune surveillance. Nature 279: 21–26.
- 4. Borst, P. 2002. Antigenic variation and allelic exclusion. Cell 109: 5-8.
- Buscaglia, C. A., V. A. Campo, A. C. Frasch, and J. M. Di Noia. 2006. *Trypanosoma cruzi* surface mucins: host-dependent coat diversity. Nature Review Microbiology 4: 229-236.
- Donelson, J. E., K. L. Hill, and N. M. A. El-Sayed. 1998. Multiple mechanisms of immune evasion by African trypanosomes. Molecular and Biochemical Parasitology 91: 51-56.
- El-Sayed, N. M. A., and J. E. Donelson. 1997. African trypanosomes have differentially expressed genes encoding homologues of the *Leishmania* GP63 surface protease. Journal of Biological Chemistry 272: 26742-26748.
- Islam, A. K. M. N., and P. T. K. Woo. 1991. *Trypanosoma danilewskyi* in *Carassius auratus*: The nature of protective immunity in recovered goldfish. Journal of Parasitology 77: 258-262.
- Lom, J., and I. Dykova. 1992. Protozoan parasites of fishes. Developments in aquaculture and fisheries science. Vol. 26, Amsterdam: Elsevier Science Publishers.
- Matthews, K. R., and K. Gull. 1994. Cycles within cycles: The interplay between differentiation and cell division in *Trypanosoma brucei*. Parasitology Today 10: 473-476.
- Matthews, K. R., J. R. Ellis, and A. Paterou. 2004. Molecular regulation of the life cycle of African trypanosomes. Trends in Parasitology 20: 40-47.

- Norris, K. A., J. E. Schrimpf, and M. J. Szabo. 1997. Identification of the gene family encoding the 160-kilodalton *Trypanosoma cruzi* complement regulatory protein. Infection and Immunity 65: 349-357.
- Overath, P., J. Haag, M. G. Mameza, and A. Lischke. 1999. Freshwater fish trypanosomes: Definition of two types, host control by antibodies and lack of antigenic variation. Parasitology **119**: 591-601.
- Overath, P., J. Ruoff, Y.D. Stierhof, J. Haag, H. Tichy, I. Dykova, and J. Lom. 1988. Cultivation of bloodstream forms of *Trypanosoma carassii*, a common parasite of freshwater fish. Parasitology Research 84: 343–347.
- Phillips, R. E. 2002. Immunology taught by Darwin. Nature Immunology
 3: 987–989
- Qadri, S. 1962. An experimental study of the life cycle of *Trypanosoma* danilewskyi in the leech, *Hemiclepsis marginata*. Journal of Protozoology 9: 254-258.
- 17. Turner, C. M. R. 2002. A perspective on clonal phenotypic (antigenic) variation in protozoan parasites. Parasitology **125:** S17–S32.
- Wang, R., and M. Belosevic. 1994. Cultivation of *Trypanosoma danilewskyi* (Laveran & Mesnil 1904) in serum-free medium and assessment of course of infection in goldfish, *Carassius auratus* (L.). Journal of Fish Disease 17: 47-56
- 19. Woo, P. T. 1981. Acquired immunity against *Trypanosoma danilewskyi* in goldfish, *Carassius auratus*. Parasitology **83**: 343–346.

CHAPTER 2

LITERATURE REVIEW¹

2.0 Introduction

The order Kinetoplastida includes genus *Trypanosoma* spp. and Leishmania spp. that infect hundreds of millions of humans and wild and domestic animals. These protozoans are characterized by the presence of kinetoplast a unique single mictochondrion, mitochondrial RNA editing, multicistronic transcription of nuclear genes and transplicing and metabolic compartmentalization [46, 213]. Trypanosomes have unique and extensive immune evasion strategies, which can be linked to their biology and life cycle. Their life cycle involves development of parasites that are morphologically and biochemically distinct within the vertebrate hosts and in the vectors. The development of different morphological forms during cell differentiation is orchestrated by the parasite cytoskeleton and is essential for the passage of parasites from vector to the host and vice versa. During development, these protozoans alternate between proliferative and non-proliferative forms adapted to a new environment, change gene expression patterns, translate specific proteins such as the variant surface antigens and generate a number of complement regulatory proteins [129, 128, 144].

¹A version of this chapter has been published: Oladiran, A., and M. Belosevic, 2011. A review: Immune evasion strategies of trypanosomes. Journal of Parasitology (*in press*)

Although many species of trypanosome cause important veterinary diseases, only two species (*T. brucei* and *T. cruzi*), cause significant human morbidity and mortality. Trypanosomes cause more than 20 million infections and 100,000 deaths per year [193].

Sleeping sickness is a re-emerging disease caused by African trypanosomes. The disease occurs as a chronic form caused by *T. brucei gambiense*, which is found in West and Central Africa, and an acute form caused by *T. brucei rhodesiense* found in Eastern and Southern Africa [220]. Sleeping sickness occurs in 36 sub-Saharan countries and is prevalent within the tsetse fly zone. It is believed that over 60 million people living in this region are at risk of contracting the disease with over 300,000 new cases every year [220, 221]. Chagas disease, also referred to as American trypanosomiasis is caused by *T. cruzi* and has been reported in all Latin America countries where it constitutes the most important parasitic infection based on socio economic impact and strain on public health systems. Chagas disease affects about 18 million people with about 100 million people living in high risk areas, and 300,000 new cases are reported every year including 21,000 deaths per year [219].

In mammals African trypanosomes, which include *T. brucei*, *T. congolense* and *T. vivax*, thrive extracellularly in the bloodstream and other tissue fluids. Some *T. brucei* subspecies cause sleeping sickness in humans whereas *T. congolense* and *T. vivax* are not infectious to immunocompetent humans but causes a disease called Nagana in cattle. The natural resistance of human to *T. brucei brucei*, which causes disease in cattle results from the presence of

trypanolytic factors in human serum causing the lysis of the parasites. On the other hand, humans are highly susceptible to *T. brucei gambiense* and *T. brucei rhodesiense* since these parasites are resistant to serum-mediated lysis [137, 188, 189]. Resistance of *T. brucei rhodesiense* has been linked to the presence of serum resistance associated gene (SRA) in this parasite [43, 224].

The infection with African trypanosomes is initiated by tsetse fly bite, resulting in the inoculation of non-dividing metacyclic trypanosomes into the host. Metacyclic trypanosomes differentiate into bloodstream parasites with heterogenous population of proliferative slender forms in the ascending phase, and non-proliferative stumpy forms during peak parasitemia. The stumpy trypanosome is pre-adapted for differentiation into procylic trypanosomes that proliferate in tsetse fly midgut [128]. The surface of the bloodstream form of the African trypanosome is covered with highly abundant variant surface glycoproteins (VSG), while the procyclic trypanosomes are covered with the invariant surface glycoprotein, procyclin [162, 167]. The VSGs are used to evade host immune responses by antigenic switching [50] and are discussed further below.

T. cruzi the causal agent of Chagas disease is primarily an intracellular parasite. Although this parasite causes disease in humans it can also infect many other mammalian species [2]. *T. cruzi* infection starts when the insect vector takes a bloodmeal from an infected host. Blood stage trypomastigotes are taken up and transform to epimastigotes in the insect digestive tract. Epimastigotes divide and transform into metacyclic promastigotes that are excreted in feces. These

metacyclic trypomastigotes enters mammalian host and eventually invade variety of cell types where they transform into dividing amastigotes. Amastigotes transform to bloodstream trypomastigotes, which are released by rupturing of the cells. These trypomastigotes thrive extracellulary in the bloodstream and can infect other host cells. They also possess complement regulatory proteins that enable parasites to avoid complement-mediated lysis [143, 144].

Non-mammalian trypanosomes have been relatively neglected and very little is known about these parasites. The fish trypanosome, T. carassii, is transmitted by blood sucking leeches from infected fish to naive hosts and is believed to be exclusively extracellular. The parasite infects freshwater fish including carp (Cyprinus carpio), eel (Anguilla spp.) and tench (Tinca tinca) in Europe and Asia [112]. Prevalence of infection and mortality caused by T. *carassii* can be as high as 100% in aquaculture setting [4]. Experimental infection with T. carassii has been demonstrated for a number of fish species including goldfish [113, 222]. We are using the goldfish model system to study hostparasite interaction with the aim of dissecting the evasion mechanisms of T. *carassii*, as well as understanding how fish respond to infection. Presently, not much is known about the life cycle and interactions of this parasite with the leech vector and the fish host. However, few authors [158, 164] have reported that the trypanosome undergo series of developmental changes within the leech. It is not known precisely how T. carassii avoids complement-mediated lysis in the fish host, although its surface coat is very similar to that of T. cruzi [4]. This suggests that *T carassii* like *T. cruzi* may have complement binding proteins on its surface.

In support of this, we have shown that trypsin treatment of *T. carassii* increases their susceptibility to lysis by the alternative complement pathway of the goldfish [156].

This review explores evasion mechanisms used by trypanosomes to evade or modulate host defences. The focus is on mammalian trypanosomes *T. brucei* and *T. cruzi* as well as non-mammalian trypanosome *T. carassii*. *T. carassii* shares fascinating similarities with both *T. brucei* and *T. cruzi*. These nonmammalian trypanosomes thrive extracellulary in the vascular system of the host like *T. brucei*, and possess a surface architecture similar to that of intracellular *T. cruzi*. Since trypanosome immune evasion strategies are closely linked to their surface coat, it is hypothesized that *T. carassii* will interact with the host defences like *T. cruzi*.

2.1 African trypanosomes and tsetse fly

2.1.1 *Tsetse fly defense against trypanosomes*

Tsetse flies (Diptera: Glossinidae) are known vector of African trypanosomes and there is evidence that 31 different species of tsetse flies are capable of transmitting trypanosomes [8]. Human infective *T. brucei rodesiense* and *T. brucei gambiense*, as well as *T. brucei brucei*, *T. congolense*, *T. vivax* and *T. simiae*, that infect domesticated animals, are all transmitted by tsetse flies [165]. With respect to fly infection and parasite establishment, the different species of trypanosomes develop in different organs of the fly. *T. vivax* develops exclusively in the mouthparts while *T. brucei* and *T. congolense* first colonise the
fly midgut following infection, and later migrate to the mouthparts where maturation occurs. *T. brucei* then migrate to the salivary gland [165]. In the tsetse fly host, trypanosomes are faced with myriads of immune challenges and have evolved a sophisticated system to cope with the fly host defences. In the tsetse fly midgut trypanosomes must deal with physical barriers, lectins, anti-microbial peptides and proteases. Variety of antimicrobial peptides including defensins, diptericins, attacins have been reported in tsetse fly midgut [75, 76, 78]. For example, the levels of attacin and defensin transcripts were found to increase by day 3 in the fat body of trypanosome-infected flies [76]. Also, targeted knockdown of attacin transcript increased trypanosome infection in the fly, indicating involvement of anti-microbial peptides in tsetse fly-trypanosome interaction [78]. It has also been established that recombinant attacin inhibits procyclic trypanosome proliferation *in vitro* [79].

Tsetse flies are refractory to trypanosomes and susceptibility to infection appears to be maternally inherited as demonstrated by breeding experiments [131, 134]. This pattern of inheritance was the focus of research leading to identification of midgut lectins, which were previously implicated in development of *T. cruzi* in triatomine bugs [155]. Lectin activity has been detected in tsetse fly midgut, and lectin inhibitory sugars fed to flies along with infective feed resulted in 100% infection [130]. Stimulation of fly immunity by bacteria before trypanosome infection led to significant decrease in trypanosome infection levels [76], suggesting that the innate immunity may be an important regulator of trypanosome infection in the flies [165]. It has been shown that not all tsetse flies in an area are capable of transmitting trypanosomes. Thus, the vector capacity of tsetse fly is governed by its ability to acquire infectious forms during a blood meal and to subsequently develop an infection and transmit the parasites to a new vertebrate host. Transmission of trypanosomes to vertebrate hosts also depends on the density of the tsetse fly population, their longevity, susceptibility of tsetse flies to infection, and human behaviour or activities that increase man-tsetse fly contacts [220].

2.1.2 T. brucei development in the tsetse fly

Tsetse flies are infected with a non-replicating short stumpy form taken up in blood meal from infected mammal. The stumpy forms survive the attack by proteases in the fly midgut and differentiate into a dividing procyclic forms [172]. It has been suggested that proteases in the fly posterior midgut could be one of the natural triggers of differentiation of stumpy trypanosome into procyclic forms [165], although additional signals have also been reported [194]. Proteases have been shown to induce differentiation of stumpy trypanosomes into procyclic forms in culture [172]. This *in vitro* differentiation can also be induced by addition of citrate or *cis*-aconitate to the culture medium [228]. In the tsetse fly midgut, the bloodstream parasites are exposed to proteases resulting in death of the slender proliferating bloodstream forms [172]. Trypanosomes that are transformed from stumpy into procyclic parasites then shed their VSG coat, they leave the midgut and migrate to the salivary gland [217]. Following differentiation into procyclic form, the bloodstream VSG is cleaved by a surface metalloprotease to allow synthesis of the procyclic stage-specific coat known as

procyclin within few hours [69, 71]. Procyclins are immunodominant surface glycoproteins rich in proline residues and are different from VSG [162]. Procyclins are encoded by multi gene family and may be EP procyclins containing Glu-Pro dipeptide repeats or GPEET procyclins with Gly-Pro-Glu-Glu-Thr repeats [167]. Both types of procyclins have several isoforms and have been extensively reviewed [166, 167]. All isoforms of procyclins are expressed at similar levels few hours following differentiation into procyclic parasites [211]. Procyclins are resistant to proteases and are thought to protect procyclic trypanosomes from destruction in the fly midgut [3, 107]. They are also required for establishment of the infection as demonstrated by procyclin gene deletion experiment [168].

Arrival of procyclic trypanosomes to the testse fly salivary gland is followed by differentiation into the metacyclic form pre-adapted for life in the mammalian host. Metacyclic trypanosomes are quiescent and have been reported to express metacyclic VSGs on their surface although their repertoire of VSG at this point in the life cycle is limited [38]. The metacyclic parasites are introduced into mammalian host during tsetse fly bite and express one of the 10-15 different VSGs for about 7 days before switching to expression of bloodstream VSGs [50]. It has been reported that metacyclic VSGs can occasionally be re-expressed as bloodstream VSGs late in the infection [50]. VSGs are very important to bloodstream form of African trypanosomes. Interference with GPI anchor synthesis [139] or down-regulation of the VSG mRNA by RNA interference [182] is lethal to trypanosomes, indicating intact VSG coat is essential for trypanosome survival. Interestingly, loss of procyclin is tolerated by insect stage parasites as shown by deletion of all procyclin genes and procyclin is neither essential for transmission of the parasite, nor required for establishment of the infection [138, 212]. However, procyclin appears to contribute to the efficiency of transmission, since genetically manipulated parasites bearing only one intact procyclin gene (GPEET) showed decreased infectivity [168].

A role for surface sialic acids in the protection of parasites in the tsetse fly midgut has been described. T. brucei cannot synthesize sialic acid, however, expression of surface glycosylphosphatidylinositols (GPIs)-anchored transsialidase enzyme by procyclic parasites has been described [57, 135]. Trypanosomes transfer sialic acids that they are unable to synthesize from the host's glycoconjugates to the GPIs on their surface using the trans-sialidase enzyme [138]. This process is important for the survival of trypanosomes in tsetse fly since parasites lacking sialic acids due to defect in synthesis of trans-sialidase are unable to survive unless sialylated by soluble trans-sialidase [138]. Interestingly, transfer of sialic acid from host's glycoconjugates onto mucins on the parasites surface has been described for both T. cruzi and T. carassii [4, 31, 63]. These parasites also use trans-sialidase enzyme, anchored to the parasite surface to transfer sialic acids. It has also been shown that trans-sialidase shed into the bloodstream binds receptors on the cell surface and influences defense mechanisms of the host [35, 199, 200].

Additional research is needed to further understand the mechanisms used by parasites to evade or modulate potentially harmful insect factors. The

completion of trypanosome genomes has revealed extensive diversity of genes that encode molecules that are differentially expressed in the different developmental stages of the parasite and are believed to be important for parasite defense strategies [219]. Progress in determining the array of insect molecules involved in the interaction with the parasites has been slow because of the limited genome information available for the insect vectors that is further complicated by the many different species of the vectors capable of transmitting trypanosomes. To understand the vector-parasite interaction on a genomic scale, the full genome sequence of insect vectors is necessary. Several tissue specific expressed sequence tags (ESTs) libraries have now been generated for tsetse fly *Glossina morsitans* morsitans. Also, there is information on the structure and size of the genome for different species, which is being used to decipher complete genome of *Glossina* [7]. EST library constructed from *Glossina* salivary gland have been sequenced to identify salivary proteins. More than 250 proteins were identified that may play a role in blood feeding, while new polypeptides families unique to Glossina were also identifed [9], although only few *Glossina* genes and proteins have been functionally characterized to date.

2.2 T. cruzi and its insect vectors

2.2.1 *T. cruzi* interaction with triatomine vectors

T. cruzi is present in the fecal droppings of infected insect vectors of the subfamily Triatominae and family Reduviidae, which are hematophagus bugs capable of transmitting the parasite. Their life cycle is characterized by five

18

nymph stages before sexual maturity, which may take up to six months depending on the species. Partly because of the control program employed against the domestic vectors, the normal pattern of transmission of *T. cruzi* (fecal route) is being replaced by secondary transmission patterns [179]. Thus, transmission by congenital transfer and blood transfusion are becoming increasingly common [219]. Recent report also indicated acquisition of infection after ingestion of food contaminated with metacyclic forms [226].

There are 140 species of Triatominae, all of which are potential vector of Chagas disease, although only *Rhodnius prolixus* and *Triatoma infestans* have been well studied [96, 178]. Adult insects can ingest blood of the host, 6-12 times their body weight, and store the blood in the anterior part of the midgut. These insects have saliva composed of various compounds to counteract hemostatic response of the mammalian host and overcome coagulation and platelet aggregation [160]. The blood is concentrated and moved into small intestine in barely digested form [64]. During this movement, the blood is prevented from clotting by anti coagulants from the salivary glands and stomach [49, 105]. Digestion and absorption of the blood occurs, and the remains of digestion are stored in the rectum and later removed by urine secreted rapidly after blood ingestion [64].

Following digestion of the blood, insects begin a starvation phase that may last for several months depending on species and climate [174]. Starvation is accompanied by shortage of nutrients in the insect gut that consequently affect parasite ability to multiply. Starvation does not appear to adversely affect spheromastigotes stage of *T. cruzi*, since an increase in proportion of this developmental stage during starvation has been reported [97]. This led to suggestion that the spheromastigote stage of the life cycle may be adapted to survive under adverse conditions [97]. About one to four weeks following infection of insect, parasite population increases rapidly in the small intestine, more than in the rectum. In the later stages of infection, parasite multiplication is more rapid in the rectum than the small intestine [173]. Preferential multiplication and habitation of rectum by *T. cruzi* has been suggested to result from ability to better attach to the rectum rather than the effects of digestive enzymes on the parasite in the small intestine [12, 66, 96]. There is evidence for attachment of the flagellated parasites to the lining of rectal cuticle with the rectal pads being the most preferred site of colonization [175].

2.2.2 T. cruzi development in Triatomines

T. cruzi life cycle is complex and involves many morphologically and physiologically distinct stages that occur along the digestive tract of the vector. The bloodstream trypomastigotes are ingested from infected mammalian host by the insect vector during a blood meal. Parasite then undergoes series of transformations in the lumen of the digestive tract of the insect vector. After few days, the bloodstream trypomastigotes develop into dividing and non-replicative amastigotes, spheromastigotes and dividing epimastigotes [13, 96]. The epimastigotes later differentiates into non-replicative and infective metacyclic trypomastigotes in the rectum. Metacyclic trypomastigotes are adapted for infecting vertebrate hosts and are passed out in feces and urine [13, 96]. It has

been reported that all stages of parasite development in the insect vector are passed out in feces and urine, however, only trypomastigotes are capable of infecting mammalian host [180]. Since *T. cruzi* cannot penetrate intact skin, parasites gain entry into the mammalian hosts through skin abrasions, punctures as well as mucous membranes [180].

In the insect gut, parasites are faced with digestive enzymes, lectins, proteases and other component of insect gut, as well as factors derived from digestion of blood meal, which may affect parasite establishment and multiplication [13, 94, 96]. It has been reported that parasite's ability to survive within the digestive tract of the vector depends on the nutritional state of the host, the strain of the parasite, trypanolytic compounds present, digestive enzymes, lectins, bacteria living in the gut and the endocrine system of the vector [65]. Some of the molecules in insect gut that interact with parasite and may interfere with parasite multiplication or survival have been identified. Trialysin, a poreforming molecule contained in saliva of T. infestans when ingested together in the blood meal has been shown to lyse the parasite [127]. Cathepsin B, the major digestive enzyme in the small intestine of triatomines, appears not to have any effect on the population density of T. cruzi [65]. However, elevated activity of cathepsin D has been observed in the small intestine of infected insects, one to three days after ingestion of blood meal containing T. cruzi compared to noninfected insects [26]. The haemolytic factors in the crop responsible for lysis of erythrocyte membrane to release free hemoglobin for digestion, as well as lectins, have been implicated shown to modulate parasite multiplication and

transformation [64]. It has been demonstrated that differential agglutinin and lysin activities exist in *R. prolixus* gut against different strains of *T. cruzi* [133].

There are also changes in temperature, osmolality, pH and nutrients, that parasite must cope with in the insect gut once it has been transferred from the mammalian host. In addition, microorganisms living in the gut of insect vectors may contribute significantly to parasite survival due to modulation of vector immunity [13]. Symbionts in the insect gut might reduce concentrations of the components of the innate immunity in the triatomines, resulting in an environment that is conducive for parasite growth [115]. Also, gut microorganisms themselves can produce factors that negatively impact parasite population density. *Serratia marcescens*, a haemolytic bacterium found in the stomach of *R. prolixus* [14] is capable of producing prodigiosin pigment that kills epimastigotes of certain *T. cruzi* strains [13, 14].

The parasites have adapted to survive attack by various elements of insect immunity. Antimicrobial peptides such as defensins are produced in the fat body, intestine and midgut of triatomines [65, 115]. Defensins are known to lyse Grampositive and Gram-negative bacteria, fungi and parasites [28, 32]. *T. cruzi* infection induces changes in defensin expression in *Rhodnius prolixus* with higher expression levels in the intestine and midgut of infected insects, although parasites appear to be protected from the high concentration of defensins produced [115]. Thus far, three isoforms of defensins have been identified in *R. prolixus* [115]. Furthermore, five cDNAs encoding lysozymes have been characterized in triatomines [11, 18, 95]. These genes are induced in response to experimental introduction of *T. cruzi* into the hemocoel of *R. prolixus* [11, 18], although the parasite does not invade hemocoel naturally. A change in transcriptional profile of lysozyme gene has been observed in *R. prolixus* following ingestion of blood containing *T. cruzi* [203]. In *T. infestans*, cysteine proteases and a cDNA encoding for lysozyme have been identified in the insect gut, although their involvement in the vector-parasite interactions is unknown [94, 95]. Lysozymes and defensins are the only antimicrobial proteins well characterized in triatomines, but there is also evidence for presence of other antimicrobial proteins in the insect vector [115, 215].

Nitric oxide generation is a conserved mechanism of innate immunity in invertebrates. In *R. prolixus, T. cruzi* induced variable expression of NOS genes according to the stage of infection. NOS gene expression was up-regulated in stomach and hemocytes during early stage of infections, ranging from one to two days post-infection, and corresponded with the period when differentiating epimastigotes inhabited the stomach. *T. cruzi* infection also induced nitrite production in the small intestine of *R. prolixus* during early stages of infection while a significant increase in nitrite levels were observed in the rectum of the insect at all stages of infection [218].

2.3 *T. carassii* and the leech vector.

2.3.1 *T. carassii* development in the leech vector

Progress on deciphering the development of *T. carassii* within the leech vector as well as interaction with the leech has been very slow. At present nothing

is known about parasite factors that contribute to survival in the leech as well as leech factors that hinder parasite growth. Transmission of fish trypanosomes by leeches has been reported [reviewed in 158]. The leech *Hemiclepis marginata* sucks blood from the fish host ingesting trypanosomes in the process. Following feeding, the leech becomes sluggish and motionless with a contracted body. Trypanosomes undergo series of morphological alterations in the leech crop as early as 3 - 5 hours after ingestion including repositioning of the kinetoplast. Parasites later differentiate into diving stumpy form one day after ingestion, then into crithidial form 2 days after ingestion. By day three, at least three distinct populations of trypanosomes were observed. The crithidial forms, the stumpy forms and a somewhat longer crithidial forms with narrow posterior and tapering anterior ends. These forms then undergo numerous alterations as the developmental cycle progresses. By day 7, tapering trypanosomes dominate the parasite population and by day 8, they become very long, slender flagellates that represent the final stage of development in the leech. These are the infective metacyclic trypanosomes some of which migrate to the proboscis sheath of the leech, to be introduced during the next blood meal. All stages of parasite development appear to occur in the leech crop [158], which is contrary to what has been observed in other trypanosomes where parasite development occurs in different regions of the insect gut.

Parasitic invertebrates such as leeches have evolved mechanisms to block host blood coagulation processes at many points during coagulation cascade. Inhibitors of factor Xa, platelets aggregation, thrombin and proteases have been identified in the leech [22, 37, 170]. *Hirudo medicinalis*, the medicinal leech, has been reported to produce many proteinase inhibitors including inhibitors of trypsin, plasmin, elastase, kallikrein, chymotrypsin, cathepsin, factor Xa, thrombin and complement component, C1s [22]. It is not known whether these anti-coagulants would affect *T. carassii* development in the leech.

2.4 Immune evasion strategies of trypanosomes in mammalian hosts

2.4.1 Trypanosome-induced immunosuppression

Trypanosomes are masters at elaborating variety of escape mechanisms to circumvent host immune responses, while inducing factors that promote their growth. Suppression of T and B cell responses to trypanosome-related and unrelated antigens is considered a major mechanism of evasion of the host immune response. Following infection with African trypanosomes, there is an increase in immunoglobulin (Ig) levels especially IgM due to activation of B cells by cytokines. These include trypanosome-specific antibodies as well as antibodies raised against auto antigens resulting from non-specific polyclonal B cell activation [214]. Surprisingly, some specific B cell and T cell responses are suppressed and several factors have been implicated in this immunosuppression. Early study on immunosuppression in trypanosome-infected cattle implicates polyclonal B cell activation and generation of suppressor T cells and suppressor macrophages as the cause of immunosuppression [83]. Several studies have reported an initial absence of specific host response towards the parasites, which was then followed by a transient proliferative T cell response to trypanosome

antigens in the first few days of infection [67]. In mice infected with *T. brucei*, T cell proliferation increased significantly in the spleen in the first few days of infection, and then the proliferative response abruptly stopped [214].

Suppression of T and B cell responses has been attributed to both suppressor T cells and suppressor macrophages. Endogenously produced IL-10 and IFNy have been shown to mediate the suppression of T cell responses in T. congolense-infected mice [204, 205]. Infection of highly susceptible BALB/c and relatively resistant C57BL/6 mice to T. congolense, showed that C57BL/6 mice controlled their parasitaemia and survived much longer, while BALB/c mice did not and succumbed to the infection within days. Susceptible BALB/c mice had higher plasma levels of IL-10, IL-4 and IFNy than the resistant C57BL/6 mice. Conversely, plasma TNF α levels were elevated significantly in the resistant relative to the susceptible mice. Concanavalin A (Con A)-induced secretions of IL-10, IL-4 and IFNy by splenocytes from infected BALB/c mice were significantly higher than those from C57BL/6 mice, indicating suppression of Con A-induced proliferation of splenocytes from infected BALB/c mice. Suppression of splenocyte proliferative response was due to production of IL-10 and IFN γ , since addition of neutralizing anti-IL-10 or anti-IFNy antibodies to the Con-Astimulated spleen cell cultures from infected BALB/c mice effectively reversed this suppression. Also, administration of anti-IFNy antibody to BALB/c mice early during infection dramatically shifted the phenotype of these susceptible mice to a more resistant-like phenotype by prolonging the survival period of the

hosts [204, 205]. It was later reported that CD4⁺ T cells mediate the immunosuppression in *T. congolense* infected-BALB/c mice [206].

Several other lines of experimental evidence exist for the induction of immunosuppression during trypanosome infection. Defects in antigen processing and inability of macrophages to present antigens to T cells have been suggested to lead to production of immunosuppressive factors such as nitric oxide and prostaglandins [214]. There is also an increase in production of TGF β , an immunosuppressive cytokine, which inhibits production of IL-4, IL-5 and IL-6 [58]. Some studies have reported deficiency in IL-2 production and IL-2 receptor expression, although prostaglandins and IFN γ produced by macrophages were thought to contribute to the suppression of IL-2 receptor expression [40, 184]. Macrophages are centrally involved in suppression of T-cell proliferative responses during *T. b. brucei* infection. T-cell proliferation is controlled by NO released by macrophages and suppressor macrophages have been shown to be responsible for suppression of T-cell responses to trypanosome antigens [70].

These suppressive effects are exerted through mechanisms that vary with tissue and stage of infection. In *T. b. brucei*-infected mice, T-cell proliferative responses, but not T-cell effector cell maturation and cytokine synthesis were impaired by prostaglandins and nitric oxide during early stages of infection [176]. It has also been suggested that NO preferentially targeted pathways important for proliferation, but not pathways required for cytokine synthesis, to account for this preferential suppressive effects [176]. Both IFN γ and TNF α appear to contribute to up-regulation of prostaglandin and NO synthesis during early stage of

infection, whereas T-cell proliferation seem to occur independent of prostaglandin and NO during late stage infection, although IFN γ produced by CD8⁺ T cells appears to be important [40]. These results support the observation that macrophages from animals infected with trypanosomes show characteristics of cellular activation probably due to contact with IFN γ , a potent activator of macrophages, produced during infection. These macrophages also exhibit suppressor cell activity [154].

The identity of parasite antigens responsible for induction of auto antibodies as well as the role of auto antibodies in trypanosome-induced immunosuppression deserves further investigation. Few investigators have reported on the specificity of auto antibodies produced during infection. Some of these antibodies were produced in response to nucleic acids [81, 93], red blood cells [93], as well as myelin [85]. These antibodies appear to be generated by a non-specific B cells stimulation or molecular mimicry [41, 136]. Identification of epitopes of parasite antigens responsible for such non-specific B cell stimulation or molecular mimicry should be the focus of future research.

2.5 Immune evasion mechanisms of *T. brucei*.

2.5.1 *The variant surface glycoprotein*

The bloodstream parasites are continuously exposed to many components of the host immune system and have multiple mechanisms to evade host immune responses. Antigenic variation of the VSG molecules is the most important immune evasion mechanisms of this parasite. These highly immunogenic and highly abundant glycoproteins form a dense cell surface coat that protects the invariant surface antigens from immune recognition. The VSG also protect the parasite from complement-mediated lysis [59, 197] and can induce production of important cytokines such as interleukin 1 (IL-1) and TNF α by macrophage [122, 194, 213]. VSG molecules can be cleaved from the parasite surface by a GPI-phospholipase C (GPI-PLC) a virulence factor that contributes to inflammatory response through action of soluble VSG and remnant GPI [124, 216]. Given the importance of VSG to African trypanosomes, soluble VSGs may have further roles in immune evasion that remain to be fully elucidated.

The *T. brucei* genome contains a repertoire of at least 1250 different VSG genes and up to 20 VSG expression sites [21, 23, 82, 209]. An allelic exclusion mechanism ensures that only one VSG gene is expressed at any one time, allowing the parasite to generate antigenically distinct glycoproteins and forcing the host to mount successive wave of VSG-specific antibodies [27]. When antibody production against a particular VSG has taken place, trypanosomes bearing this coat are lysed but a small number of parasites begin to express an antigenically distinct VSG, and eventually populate the vascular system of the host [201]. It has been reported that elimination of trypanosomes was due to phagocytosis and complement-mediated lysis that was mediated by anti-VSG antibodies [44, 119, 153]. The first class of anti-typanosomal antibody produced in infected mammal is the IgM generated against the specific VSG and this IgM fixes complement and aids in the assembly of complement components on the parasite surface [153]. The complement has been shown to efficiently lyse

parasites *in vitro* [72]. Phagocytosis of IgM opsonized parasites occurs by engagement of complement receptor CR3 (CD11b/CD18), and is facilitated by iC3b produced as a byproduct of antibody-induced complement activation and from activation of alternative pathway of complement [153]. In addition to IgM, IgG has also been reported to mediate phagocytosis of African trypanosomes [196]. To prevent anti-VSG antibody mediated phagocytosis, African trypanosomes restrict accumulation of VSG antibodies on their surface by endocytosis that occurs through the flagellar pocket of the parasites [56].

2.5.2 *Modulation of host immunity*

Presence of African trypanosomes in the bloodstream leads to polyclonal B cell activation resulting in increased production of IgM [20]. *In vivo* experiments have demonstrated the role of antibodies in clearance of trypanosome infections. IgG is involved in clearing *T. congolense* and *T. brucei* infections in mice [121, 153, 192]. In contrast, control of experimental *T. evansi* infection in mice appears to be IgM dependent, suggesting different antibody classes are involved in elimination of different trypanosome species [19].

Macrophages and other innate immune cells play important roles in host defense against trypanosomes. Macrophages are highly phagocytic and produce several pro-inflammatory cytokines, reactive oxygen and nitrogen intermediates upon appropriate activation. Macrophage activation has been reported in trypanosome infections. Interferon gamma (IFN- γ) produced by T-helper 1 (Th1) cells activated by parasite antigens is believed to serve as strong activation signal for macrophages. Complement receptor 3 (CR3)-dependent phagocytosis of IgM or IgG opsonised parasites by activated macrophages and Kupffer cells (liver macrophages) results in TNF α production and down-regulation of NO synthesis [77, 153, 183, 192]. Since NO is toxic to African trypanosomes [52], this provide parasite with means to evade destruction by reactive nitrogen intermediates, one of the major antimicrobial mechanism of macrophages. TNF α has been demonstrated to kill parasites [116, 123], although resistance of some parasites to TNF α has also been reported [92]. Other macrophage receptors involved in parasite recognition include members of the TLRs family that signal through MyD88-dependent pathway [51, 153].

2.5.3 Modulation of host cytokine gene expression

African trypanosomes are known for their ability to secrete molecules that modulate host immune responses. These molecules, which are collectively known as trypanokines, are trypanosome proteins that modulate host cytokines for parasite benefit. Secretion of trypanokines is an example mechanism that parasites have evolved to enable their persistence in the host [207]. Trypanosomes cysteine proteases and T lymphocyte triggering factors (TLTF) are examples of trypanokines [88, 131]. TLTF triggers CD8⁺ cells to produce IFN- γ , which stimulates parasites growth and production of TGF- β [15, 16, 17, 149, 150, 151]. TLTF is located in the flagellar pocket and interacts directly with the CD8 molecule on the surface of IFN γ -secreting cells to induce cytokine production [149, 207]. This molecule is produced by trypanosomes early during experimental infection, since the host is capable of generating neutralising antibodies that prevent TLTF function during later stages of infection [74].

Trypanosomes can also modulate host cytokines to enhance their survival in the host. It is well documented that parasites or parasite-derived molecules are capable of triggering immune response not required for their elimination [110, 124, 147, 154, 161, 181]. This strategy favours parasite establishment and persistence while diverting host responses away from parasite elimination. During the early stages of infection of mice with African trypanosomes higher levels of pro-inflammatory cytokines IFNy and TNF α were reported [39, 191]. Similarly, the infection of vervet monkeys with *T. brucei rhodesiense* induced higher levels of IFN γ , TNF α and soluble TNF receptor 1 in serum of the infected animals [125]. These findings suggest preferential parasite-orchestrated induction of a Th1-type immune response, which is known to be ineffective against extracellular parasites. The survival of the host is thought to be dependent on a shift from proinflammatory cytokines (IFN γ and TNF α), to the anti-inflammatory cytokines (TGF β and IL-10) production, enabling increased antibody response and eventual elimination of the parasites [140].

2.5.4 Evasion of complement-mediated lysis

T. brucei activates the alternative complement pathway in human hosts but it escapes lysis by preventing the complement cascade from proceeding beyond the association of C3 convertase with the parasite surface [45]. It was demonstrated that parasite incompletely activates the alternative pathway of complement without generating the terminal complex of the cascade resulting in inhibition of lysis [45]. Parasite VSG has been implicated in protection from complement-mediated lysis [197]. Trypanosomatid gp63 proteases have been suggested to be involved in host-parasite interaction and share sequence as well as structural similarities to the gp63 of their mammalian host and insect vectors. *T. brucei* homologues of *Leishmania* glycoprotein 63 (gp63) have been identified and well characterised. *Leishmania* gp63 is important for entry of promastigote into host macrophages and survival inside macrophages [34, 109, 190]. The protease activity of gp63 also contributes to the resistance of promastigotes to complement-mediated lysis [29], while immunization with recombinant protein partially protects animals in experimental infections [1, 36, 148, 225]. *T. brucei* appears to possess multiple genes encoding gp63 that are differentially expressed in the bloodstream parasites [55, 104]. It has been reported that gp63 may protect bloodstream trypanosomes from complement-mediated lysis by impairing a protein processing functions [55, 69, 104]. Gp63 is also required for removal of VSGs from the surface of bloodstream trypanosomes [69, 104].

2.6 Immune evasion mechanisms of *T. cruzi*

2.6.1 Role of mucin glycoproteins

T. cruzi amastigotes escape host humoral immune responses by adopting an intracellular existence. The parasite surface is covered in mucin glycoproteins that are essential for protection from recognition by host immune system. These mucins are acceptors of sialic acid residues, which are transferred from host glycoconjugates by the enzyme trans-sialidase. That *T. cruzi* genome contains ~ 850 mucin-encoding genes, which constitute ~ 1% of the genome and 6% of predicted genes, highlights the significance of these proteins [31]. The mucin glycoproteins are also found on the surface of *T. carassii* [108] and *Cryptosporidium parvum* [198]. These molecules act at the interface between the parasites and the mammalian and non-mammalian hosts and protect the parasites against their immune responses [31].

The mucin glycoproteins are differentially expressed during *T. cruzi* life cycle. Both the amastigotes and blood trypomastigotes express variable mucin belonging to *T. cruzi* mucins (TcMUC) family, while the epimastigotes express the invariant mucins of T. *cruzi*, small mucin-like gene (TcSMUG) family. It is unclear which types of mucins are expressed by metacyclic trypomastigotes, however, it has been proposed that these parasites express the invariant mucins of the TcSMUG family similar to epimastigotes. TcMUC is characterized by an internal tandem repeats of Thr8-Lys-Pro2 sequence flanked by an N-terminal signal peptide and a GPI-anchor signal. The TcSMUG have a shorter central region with similar flanking regions, as observed in TcMUC [31]. Gene duplication, point mutations, deletions and insertions contribute to extensive diversity of the mucin genes [47, 48].

2.6.2 Complement regulatory proteins

The major evasion strategy of *T. cruzi* is the possession of several complement regulatory molecules, which allow the parasite to evade complement-mediated lysis (CML) [143, 210]. The bloodstream trypomastigotes possess the ability to resist CML using several surface glycoproteins that prevent complement activation [62, 86, 144]. The most studied of the complement binding molecules is

the 160kDa complement regulatory protein (160kDa CRP), which is expressed on the parasite surface and restricts activation of both the alternative and classical pathway of complement. This molecule binds complement components C3b and C4b, thereby interfering with the assembly of proteolytically active C3 convertase formation [144]. Transfection of epimastigotes with cDNA encoding trypomastigote 160 kDa CRP results in expression of the CRP on the epimastigote surface leading to transformation of complement sensitive epimatigotes to complement resistant parasites [142]. As shown in Figure 2.1, T. cruzi CRPs prevent assembly of C3 convertase enzymes of both the classical and alternative pathways of complement. Lysis of T.cruzi by complement is dependent on lytic antibodies [30, 101, 144], however, the alternative complement pathway is primarily responsible for the demise of the parasites. The classical pathway is inefficient in parasite lysis but enhances lysis by the alternative pathway of complement since sera from individuals that are genetically deficient in classical pathway component protein C2 lysed 50% of IgG-coated parasites, and the addition of purified C2 to the deficient sera significantly enhanced lysis of the parasites lysis [99].

The inability of the classical pathway of complement to efficiently destroy the parasites may be partly due to calreticulin (CRT), a parasite molecule (TcCRT) known to inhibit classical pathway-mediated lysis [6, 61, 146, 208]. TcCRT binds human C1q and inhibit C1q-dependent lysis [61], indicating that the presence of TcCRT on parasite surface could inhibit C1q-mediated activation of classical complement pathway at the earliest stage of complement activation. Figure 2.1 indicates where trypanosome calreticulin acts to inhibit the classical pathway of complement. Additionally, secreted CRT molecules may perform other functions such as binding of host molecules away from the parasites, nonspecific stimulation of immune responses or immunosuppression [146]. Parasite CRTs have multiple C1q-binding motifs, which are thought to enhance specific interaction with host C1q molecules [141]. Thus, multiple C1q binding motifs in parasite CRTs may represent an evolutionary strategy to compensate for multipoint interaction between antibody and C1q, to ensure complete inactivation of the classical complement pathway [141]. CRTs bind both collagenous tail and globular domains of C1q [90, 91, 141] and can compete with antibody for binding to C1q, indicating CRTs and antibodies share C1q binding sites [141]. TcCRT prevents human classical complement pathway activation by competitively inhibiting binding of (C1r-C1s)₂ tetrameric complex to collagenous tails of C1q and functionally interfering with C1s activities without releasing the serine protease from the C1q recognition complex [208]. Since T. cruzi infected individuals produce variable levels of specific antibodies against TcCRT in their serum [126], these antibodies may inhibit TcCRT-human C1q interaction that may benefit the hosts. Antibody raised against TcCRT inhibits calreticulin interaction with human C1q in vitro [5].

Several lines of evidence support the notion that *T. cruzi* elicits severe autoimmune responses in the host during infection and this contributes significantly to pathogenesis of Chagas' disease. This infection-induced autoimmune disease is believed to cause severe cardiac pathology observed in about 30% of patients with chronic infection. TcCRT has high degree of homology (more than 50% identity) to human CRT (HuCRT), and is immunogenic in humans and mice. Anti-TcCRT antibodies generated in mice immunized with recombinant TcCRT (rTcCRT) reacted with rHuCRT suggesting that TcCRT may be involved in induction of autoimmunity in Chagas disease [159]. The anti-angiogenic and anti-tumor activities of TcCRT have also been described. TcCRT inhibits capillary morphogenesis, migration and proliferation of human endothelial cells, and growth of murine mammary tumor cell line. These observations, together with the ability to inhibit classical pathway of complement, suggest that TcCRT is an anti-inflammatory molecule that impairs anti-parasite immune responses [114].

2.6.3 Gp63 proteases

Annotation of *T. cruzi* genome revealed duplication of gp63 gene resulting in the division of the genes into four novel groups based on sequence features [118]. Complete genome and proteome analyses of *T. cruzi* have revealed at least 174 copies of *T. cruzi* gp63 (Tcgp63) that encode at least 29 different proteins. Many Tcgp63 genes are expressed in multiple life cycle stages, while others are stage specific. Antibodies raised against a Tcgp63 inhibit infection of cells by trypomastigotes as shown by neutralization assays [102]. These results suggest involvement of Tcgp63 in host-parasite interaction and evasion of phagocytic killing. Gp63 proteases of *T. cruzi* were extensively duplicated and contain several novel domains [118], suggesting Tcgp63 could play additional roles in immune evasion that are yet to be identified.

2.6.4 Interference with cytokines and phagocytic cells

Microbial pathogens have evolved strategies to benefit from immune regulatory mechanisms provided by cytokines [42]. T. cruzi induces the production of both Th1 and Th2 cytokines in infected individuals. Elevated expression of Th1 cytokines IFNy and IL2 as well as Th2 cytokines IL-4 and IL-10 were reported [171]. Level of IL-10 mRNA was consistently elevated in T. *cruzi*-infected individuals leading to suggestion that the elevated IL-10 level may contribute to parasite persistence, since IL-10 is known to inhibit the protective Th1 immune responses [2, 186, 187]. Induction of IL-10 may be an important strategy for parasite survival, since elimination of parasites is dependent on Th1 mechanism that result in production IL-12, IFN γ , TNF α and the induction of nitric oxide response by activated macrophages [33, 80, 185]. IFN γ produced by NK cells during T. cruzi infection activates phagocytic cells to destroy internalised parasites through production of reactive nitrogen intermediates [10, 33, 68]. However induction of cytokines that leads to both cell mediated and humoral response has been suggested to be important for development of protective immunity [227].

2.7 Immune evasion mechanisms of *T. carassii*

2.7.1 T. carassii infection of goldfish

Although the prevalence of infection in aquaculture setting can reach 100%, *T. carassii* is considered to be non-pathogenic under natural conditions [111, 152]. The dose of parasites determines the severity of infection and

mortality is higher when fish are injected with high number of parasites [84]. In experimental infection of goldfish, symptoms such as anaemia, anorexia and hsitopathological changes in hematopoietic organs have been reported [53, 84]. It is unclear how *T. carassii* is able to maintain long-term infections in the fish host. As discussed earlier, *T. brucei* employ antigenic variation of their surface glycoproteins,to escape host immune attack. *T. cruzi* has mucin glycoproteins as well as complement regulatory proteins to survive host immune insults. Although *T. carassii* have mucin glycoproteins on their surface, these proteins have not been experimentally demonstrated to aid in parasite persistence.

2.7.2 Evasion of humoral immunity

T. carassii is believed to be exclusively extracellular since there is currently no evidence for intracellular stage of this parasite, suggesting predisposition to direct contact with components of both cellular and humoral immunity. In experimental infection of goldfish with *T. carassii* recovered host become immune to re-infection, suggesting the presence of immunological memory. Injection of serum or purified IgM from recovered fish conferred protection to naïve hosts in passive transfer studies [111, 152, 223]. Immunization of goldfish with parasite excretory/secretory product induces protective antibodies [24], however it is unclear how these antibodies function to eliminate parasites. Antibody-dependent complement-mediated lysis may be one way by which these antibodies mediate parasite clearance. Although the parasite antigens that induce protective antibodies in the host are yet to be fully elucidated, it is tempting to speculate that *T. carassii* possess surface molecules including complement regulatory proteins similar to those of *T. cruzi*. Also, it is unclear whether *T. carassii* is able to endocytose surface bound antibodies for destruction, as observed for VSG-bound antibody in African trypanosomes.

2.7.3 The search for complement regulatory proteins in T. carassii

It is not known how T. carassii evades complement-mediated lysis in the fish host. The similarity in surface coat between T. cruzi and T. carassii has led to suggestion that T. carassii may also possess complement regulatory proteins on their surface for protection against complement lysis. This suggestion is supported by the observation that trypsin treated parasites were more susceptible to lysis by the alternative complement pathway of goldfish [156]. Parasites regained resistance to lysis 6-24 hrs after cultivation in the absence of trypsin and the resistance to lysis was abrogated after treatment of the parasites with protein synthesis inhibitor puromycin [156]. These results strongly suggest the existence of surface molecules that protect trypanosomes from lysis by fish complement cascades. Only few parasite molecules have been identified to date and none of these molecules is a major protective antigen. Parasite tubulin and cysteine proteinase have been reported as potential molecules that may participate in protection from complement-mediated lysis, but immunization of fish with the recombinant proteins only induced partial protection against a challenge infection [24, 87, 163]. These parasite molecules, although immunogenic, do not stimulate long lasting protection against re-infection. Immunization of mice with tubulinrich preparation from T. brucei, and recombinant beta-tubulin from T. evansi offers partial protection against experimental challenge with these parasites [117,

106]. Alternatively, immunization with recombinant proteins may fail to induce effective lytic or protective antibodies as observed for *T. cruzi* CRPs where DNA but not protein immunization was found to be protective [177].

Recently, I [146] reported on the identification of *T. carassii* calreticulin (TcaCRT) as a secreted or excreted (ES) molecule that is abundantly expressed on the parasite surface. TcaCRT binds goldfish C1q, the first component of the classical pathway of fish complement and inhibits classical complement pathway-mediated lysis [146]. The ability of trypanosome calreticulin to inhibit complement lysis at the earliest stage of complement activation (Fig. 2.1) was first observed for *T. cruzi* calreticulin [61]. Since complement lyses trypsinized *T. carassii*, TcaCRT may contribute to protection of *T. carassii* from complement-mediated lysis. These results suggest TcaCRT inhibition of classical pathway may be partly responsible for the ineffective nature of goldfish classical complement pathway to cause parasite lysis [146, 156]. The presence of multiple C1q-binding motifs on TcaCRT may allow this molecule to competitively inhibit antibody binding or induce conformational changes that effectively abrogate the activation of classical complement pathway to goldfish [146].

2.7.4 *Modulation of host immunity*

We previously reported that *T. carassii* ES molecules are immunogenic and identified parasite tubilin, hsp70 and calreticulin in this protein fraction [24, 146, 147]. Antibodies raised against parasite tubulin inhibit parasite growth *in vitro*, although immunization of fish with the recombinant molecule only partially reduced parasitemia [24, 87]. Because tubulins are highly abundant and immunogenic, they are targets of host immune response and antibody production during infection. These antibodies may benefit the host by eliminating the parasites. However, tubulins are also shed during *in vitro* cultivation of the parasites, indicating that they may be released from parasites *in vivo* and may direct immune responses away from the parasite surface. It is possible that parasite tubulin play other roles in immune evasion that are yet to be elucidated.

The ability of *T. carassii* hsp70 (Tchsp70) to enhance pro-inflammatory response of goldfish macrophages suggests the importance of this molecule in possible modulation of host immunity [147]. Tcahsp70 was identified in ES fraction as well as in membrane protein fraction of *in vitro* cultured parasites separated by 2-dimensional gel electrophoresis [147]. These results indicate a novel function of trypanosome hsp70 as a possible modulator of fish immune response. Recombinant Tcahsp70 (rTcahsp70) enhanced expression of proinflammatory cytokines and chemokines of goldfish macrophages. These findings suggest surface expression of Tcahsp70, and provide parasites with an excellent immune evasion strategy by preferentially inducing a Th1 type immune response to evade Th2 type response, which has been shown to play key role in parasite elimination [84, 87, 152, 223, 147]. The secreted hsp70 and calreticulin may have other roles in immune modulation. It is not clear at this point whether T. carassii suppresses host immunity or induces autoimmunity. Because of the similarity of T. carassii and T. cruzi surface coats, and the high degree of homology of calreticulins from both parasites [146], I speculate that extracellular TcaCRT may be involved in induction of autoimmunity, angiogenesis or immunosuppression.

I recently observed an increased expression of both pro-inflammatory and anti-inflammatory cytokines in *T. carassii* infected fish during the course of the infection [145]. I observed induction of INFγ and IL-10 during the acute phase of infection, suggesting a mixed Th1/Th2-type immune response [145]. An overlapping Th1/Th2 cytokine response can lead to generalized immunosuppression, which is a hallmark of African trypanosome infection [20, 204, 205, 206] although an overlapping Th1/Th2 cytokine patterns have also been observed in *T. cruzi* infected individuals [171]. Since increased expression of pro-inflammatory genes during the course of the infection paralleled the upregulation of anti-inflammatory cytokine genes, these results suggest Th1/Th2like responses may be important in the control of *T. carassii* infection in the goldfish. Furthermore, these findings indicate that IFNγ and IL-10 may play significant role in the control of fish trypanosomiasis.

2.7.5 *Possible role of proteases*

A cathepsin L-like proteinase (cysteine proteinase) of *T. carassii* has been cloned and characterised. This cysteine proteinase exhibits a high level of conservation within the Trypanosomatida in terms of structure and activity [163]. Immunization of carp with the recombinant proteinase induced an increase in proteinase-specific antibodies and slightly lowered parasitemia in experimentally challenged animals [163] suggesting that *T. carassii* cysteine proteinase is not a major protective antigen in carp. I have recently cloned *T. carassii* protein. *T. carassii* gp63 shares sequence similarity with those of other kinetoplastids and it appears there

are multiple genes encoding gp63 in this parasite. Trypanosome gp63 warrant further study to understand the significance of these molecules in host-parasite interactions, as well as in parasite elimination, since these molecules elicit different effects.

2.8 Summary

In this review, I have provided an exploratory view of the current knowledge on immune evasion or modulation strategies employed by trypanosomes in both endothermic and ectothermic hosts. Various aspects of trypanosome biology that contribute to their remarkable immune evasion strategies were discussed so as to better understand these essential survival mechanisms of these highly successful protozoa. To emphasise the significant role played by the vectors of trypanosomes, I have included detailed discussion of parasite interactions with the insect or leech vectors. These parasites are fascinating with respect to their biology. They cycle between vertebrate and invertebrate hosts and must survive a plethora of antimicrobial immune responses in both hosts. Additionally, parasites have evolved ways to cope with change in temperature and nutrients as they cycle between hosts. Many of the molecules used by mammalian trypanosomes in host-parasite interaction and immune evasion have been described. More progress in this area is expected with the deluge of genome information available as investigators move from gene discovery to protein characterization and function.

The goldfish-*T. carassii* model system provides an excellent opportunity to study host-pathogen interaction that occur at the natural host-parasite interface. This is an obvious advantage over *T. brucei* and *T. cruzi*, where investigation of host-parasite interactions has been largely carried out in mice that are not the natural hosts for these parasites.



Modified from Nonaka and Smith, 2000.

Figure 2.1. The complement system. The alternative, lectin and classical pathways of complement are activated differently. Trypanosomes have developed strategies to inhibit complement-mediated lysis. *T. cruzi* complement regulatory proteins prevent formation of C3 convertase enzymes of both alternative and classical pathways. Both *T. cruzi* and *T. carassii* calreticulin can inhibit the classical pathway at the earliest level of complement activationComplement molecules that have been isolated or cloned in fish are indicated.

2.9 References

- Abdelhak, S., H. Louzir, J. Timm, L. Blel, Z. Benlasfar, M. Lagranderie, M. Gheorghiu, K. Dellagi, B. and Gicquel. 1995. Recombinant BCG expressing the leishmania surface antigen Gp63 induces protective immunity against Leishmania major infection in BALB/c mice. Microbiology 141: 1585-1592.
- Abrahamsohn, I. A., and R. L. Coffman. 1996. *Trypanosoma cruzi*: IL-10, TNF, IFN-γ, and IL-12 regulate innate and acquired immunity to infection. Experimental Parasitology 84: 231-244.
- Acosta-serrano, A. E., E. Vassella, M. Liniger, C. Kunzrenggli, R. Brun, I. Roditi, and P. T. Englund. 2001. The surface coat of procyclic Trypanosoma brucei: programmed expression and proteolytic cleavage of procyclin in the tsetse fly. Proceedings of National Academy of Science U S A. 98: 1513-1518.
- Agüero, F., V. Campo, L. Cremona, A. Jäger, J.M Di Noia, P. Overath, D.O Sánchez, and A.C. Frasch. 2002. Gene discovery in the freshwater fish parasite Trypanosoma carassii: identification of transialidase-like and mucin-like genes. Infection and Immunity **70:** 7140-7144.
- Aguilar, L., G. Ramírez, C. Valck, M.C. Molina, A. Rojas, W. Schwaeble, V. Ferreira, and A. Ferreira. 2005. F(ab')2 antibody fragments against Trypanosoma cruzi calreticulin inhibit its interaction with the first component of human complement. Biological Research 38: 187-195.
- Aguillon, J. C., L. Ferreira, C. Perez, A. Colombo, M.C. Molina, A. Wallace, A. Solari, P. Carvallo, M. Galindo, N. Galanti et al. 2000. Tc45, a dimorphic Trypanosoma cruzi immunogen with variable chromosomal localization, is calreticulin. American Journal of Tropical Medicine and Hygiene 63: 306-312.
- Aksoy, S., M. Berriman, N. Hall, M. Hattori, W. Hide, and M. J. Lehane. 2005. A case for a Glossina genome project. Trends in Paraistology 21: 107-111.

- Aksoy, S., W. C. Gibson, and M. J. Lehane. 2003. Interactions between tsetse and trypanosomes with implications for the control of trypanosomiasis. Advances in parasitology 53: 1-83.
- Alves-silva, J., J. M. Ribeiro, J. Van Den Abbeele, G. Attardo, Z. Hao, L. R. Haines, M. B. Soares, M. Berriman, S. Aksoy, M. J. Lehane. 2010. An insight into the sialome of Glossina morsitans morsitans. BMC Genomics 11: 213.
- Antunez, M. I., and R. L Cardoni. 2000. IL-12 and IFN-γ production, and NK cell activity, in acute and chronic experimental *Trypanosoma cruzi* infections. Immunology letters **71**: 103-109.
- Araujo, C. A., P. J. Waniek, P. Stock, C. Mayer, A. M. Jansen, G. A. Schaub. 2006. Sequence characterization and expression patterns of defensin and lysozyme encoding genes from the gut of reduviid bug *Triatoma brasiliensis* Insect Biochemistry and Molecular Biology 36: 547-560.
- Asin, S., S. Catala, 1995. Development of *Trypanosoma cruzi* in Triatoma infestans: influence of temperature and blood consumption. The Journal of Parasitology 81: 1-7.
- Azambuja, P., E. S. Garcia, N. A. Ratcliffe. 2005. Gut microbiota and parasite transmission by insect vectors. Trends in Parasitology 21: 568-572.
- Azambuja, P., D. Feder, E. S. Garcia. 2004. Isolation of *Serratia marcescens* in the midgut of *Rhodnius prolixus*: impact on the establishment of the parasite, *Trypanosoma cruzi*, in the vector. Experimental Parasitology **107**: 89-96.
- Bakhiet, M., T. Olsson, J. Mhlanga, P. Buscher, N. Lycke, P.H. van der Meide, and K. Kristensson. 1996. Human and rodent IFN-γ as a growth factor for *Trypanosoma brucei brucei*. European Journal of Immunology 26: 1359-1364.
- Bakhiet, M., T. Olsson, C. Edlund, B. Hojeberg, K. Holmberg, J. Lorentzen, and K. Kristensson. 1993. *Trypanosoma brucei brucei* derived

factors that triggers CD8⁺ lymphocytes to interferon gamma secretion: purification, characterization and protective effects *in vivo* by treatment with monoclonal antibodies against the factor. Scandinavian Journal of Immunology **37:** 165-178

- Bakhiet, M., T. Olsson, P. van der Meide, and K. Kristensson. 1990. Depletion of CD8+ T cells suppresses growth of *Trypanosoma brucei brucei* and interferon-gamma production in infected rats. Clinical and Experimental Immunology 81: 195-199.
- Balczun, C., E. Knorr, H. Topal, C. K. Meiser, A. H. Kollien, G. A. Schaub. 2008. Sequence characterization of an unusual lysozyme gene expressed in the intestinal tract of the reduviid bug *Triatoma infestans*. Parasitology Research 102: 229-232
- Baral, T. N., P. De Baestselier, F. Brombacher, and S. Magez. 2007. Control of *Trypanosoma evansi* infection is IgM mediated and does not require a Type 1 inflammatory response. Journal of Infectious Disease 195: 1513-1520
- Barrett, M. P., R. J. Burchmore, A. Stich, J. O. Lazzari, A. C. Frasch, J. J. Cazzulo, and S. Krishna. 2003. The trypanosomiases. Lancet 362: 1469-1480.
- Barry, J. D., L. Marcello, L. J. Morrison, A. F. Read, K. Lythgoe, N. Jones, M. Carrington, G. Blandin, U. Bohme, E. Caler et al. 2005. What the genome sequence is revealing about trypanosome antigenic variation. Biochemical Society Transactions 33: 986-989
- 22. Baskova, I. P., and L. L. Zavalova. 2001. Proteinase inhibitors from the medicinal leech *Hirudo medicinalis*. Biochemistry **66**: 703-714.
- 23. Berriman, M., E. Ghedin, E., C. Hertz-Fowler, G. Blandin, H. Renauld, D. C. Bartholomeu, N. J. Lennard, E. Caler, N. E. Hamlin, B. Haas et al. 2005. The genome of the African trypanosome *Trypanosoma brucei*. Science **309**: 416-422.
- 24. Bienek, D. R., D. A. Plouffe, G. F. Wiegertjes, and M. Belosevic. 2002. Immunization of goldfish with excretory/secretory molecules of
Trypanosoma danilewskyi confers protection against infection. Developmental and Comparative Immunology **26:** 649-657.

- Black, S. J., P. Guirnalda, D. Frenkel, C. Haynes, V. Bockstal. 2010. Induction and regulation of *Trypanosoma brucei* VSG-specific antibody responses. Parasitology 137: 2041-2049.
- 26. Borges, E. C., E. M. Machado, E. S. Garcia, P. Azambuja. 2007. *Trypanosoma cruzi*: effects of infection on cathepsin D activity in the midgut of *Rhodnius peolixus*. Experimental Parasitology **112**: 130-133.
- 27. Borst, P. 2002. Antigenic variation and allelic exclusion. Cell 109: 5-8.
- Boulanger, N., P. Bulet, C. Lowenberger. 2006. Antimicrobial peptides in the interaction between insects and flagellate parasites. Trends in Parasitology 22: 262-268.
- 29. Brittingham, A., C. J. Morrison, W. R. McMaster, B. S. McGwire, K. P. Chang, and D. M. Mosser. 1995. Role of the *Leishmania* surface protease gp63 in complement fixation, cell adhesion, and resistance to complement-mediated lysis. Journal of Immunology **155**: 3102-3111.
- Budzko, D. B., M. C. Pizzimenti, and F. Kierszenbaum. 1975. Effects of complement depletion in experimental chagas disease: immune lysis of virulent blood forms of *Trypanosoma cruzi*. Infection and Immunity 11: 86-91.
- Buscaglia, C. A., V. A. Campo, A. C. Frasch, and J. M. Di Noia. 2006. *Trypanosoma cruzi* surface mucins: host-dependent coat diversity. Nature Review Microbiology 4: 229-236
- 32. Bulet, P., and R. Stocklin. 2005. Insect antimicrobial peptides: structures, properties and gene regulation. Protein and Peptide Letters **12**: 3-11.
- 33. Cardillo, F., J. C. Voltarelli, S. G. Reed, and J. S. Silva. 1996. Regulation of *Trypanosoma cruzi* infection in mice by gamma interferon and interleukin 10: role of NK cells. Infection and Immunity 64: 128-134.
- 34. Chang, C. S., and K. P.Chang. 1986. Monoclonal antibody affinity purification of a *Leishmania* membrane glycoprotein and its inhibition of

leishmania-macrophage binding. Proceedings of National Academy of Science USA **83:** 100-104.

- Chuenkova, M., and M. E. Pereira. 1995. *Trypanosoma cruzi trans*sialidase: enhancement of virulence in a murine model of Chagas' disease. Journal of Experimental Medicine **181**: 1693-1705.
- 36. Connell, N. D., E. Medina-Acosta, W. R. McMaster, B. R. Bloom, and D. G. Russell. 1993. Effective immunization against cutaneous leishmaniasis with recombinant bacille Calmette-Guerin expressing the *Leishmania* surface proteinase gp63. Proceedings of National Academy of Science USA **90**: 11473-11437.
- 37. Corral-Rodriguez, M. A., S. Macedo-Ribeiro, P. J. Pereira, and P. Fuentes-Prior. 2010. Leech-derived thrombin inhibitors: from structures to mechanisms to clinical applications. Journal of Medicinal Chemistry 53: 3847-3861.
- Crowe, J. S., J. D. Barry, A. G. Luckins, C. A. Ross, and K. Vickeman. 1983. All metacyclic variable antigen types of *Trypanosoma congolense* identified using monoclonal antibodies. Nature **306**: 389-391.
- 39. Darji, A., M. Beschin, M. Sileghem, H. Heremans, L. Brys, and P. De Baetselier. 1996. In vitro simulation of immunosuppression caused by *Trypanosome brucei*: active involvement of gamma interferon and tumor necrosis factor in the pathway of suppression. Infection and Immunity 64: 1937-1943.
- 40. Darji, A., M. Sileghem, H. Heremans, L. Brys, and P. De Baetselier. 1993.
 Inhibition of T-cell responsiveness during experimental infections with *Trypanosoma brucei*: active involvement of endogenous gamma interferon. Infection and Immunity **61**: 3098-3102.
- 41. Davies, J. M. 1997. Molecular mimicry: can epitope mimicry induce autoimmune disease?. 1997. Immunology and Cell Biology **75:** 113-126.
- 42. Damian, R. T. 1997. Parasite immune evasion and exploitation: Reflections and projections. Parasitology 115: 169-175.

- 43. De Greef, C., H. Imbrechts, G. Matthyssens, N. Van Meirvenne, and R. Hamers. 1989. A gene expressed only in serum-resistant variants of *Trypanosoma brucei rhodesiense*. Molecular and Biochemical Parasitology 36: 169-176.
- 44. Dempsey, W. L., and J. M. Mansfield. 1983. Lymphocyte functions in experimental African trypanosomiasis. V. Role of antibody and the mononuclear phagocyte system in variant-specific immunity. Journal of Immunology 130: 405-411.
- 45. Devine, D. V., R. J. Falk, and A. E. Balber. 1986. Restriction of the alternative pathway of human complement by intact *Trypanosoma brucei* subsp. *gambiense*. Infection and Immunity **52**: 223-229.
- 46. De Souza W. Basic cell biology of *Trypanosoma cruzi* Curr Pharm Des.2002. 8: 269-364
- 47. Di Noia, J. M., I. D'Orso, I. Aslund, D. O. Sanchez, and A. C. Frasch.
 1998. The *Trypanosoma cruzi* mucin family is transcribed from hundreds of genes having hypervariable regions. Journal Biological Chemistry 273: 10843-10850
- 48. Di Noia, J. M., G. D. Pollevick, M. T. Xavier, J. O. Previato, I. Mendoca-Previato, D. O. Sanchez, and A. C. Frasch. 1996. High diversity in mucin genes and mucin molecules in *Trypanosoma cruzi*. Journal of Biological Chemistry **271**: 32078-32083.
- 49. Dodt, J., M. Otte, K. H. Strube, and T. Friedrich. 1996. Thrombin inhibitors of bloodsucking animals. Seminars Thrombosis and Hemostasis
 22: 203-208
- Donelson, J. E., K. I. Hill, and N. M. A. El-Sayed. 1998. Multiple mechanisms of immune evasion by African trypanosomes. Molecular and Biochemical Parasitology 91: 51-56.
- Drennan, M. B., B. Stijlemans, V. Abbeele, V. J. Quesniaux, M.
 Barkhuizen, F. Brombacher, P. De Baestselier, B. Ryffel, and S. Magez.
 2005. The induction of a type 1 immune response following *Trypanosoma*

brucei infection is MyD88 dependent. Journal of Immunology **175:** 2501-2509.

- Duleu, S., P. Vincendeau, P. Courtois, S. Semballa, I. Lagroye, S. Daulouede, J. I. Boucher, K. T. Wilson, B. Veyret, and A. P. Gobert. 2004. Mouse strain susceptibility to trypanosome infection: an arginase-dependent effect. Journal of Immunology **172**: 6298-303.
- 53. Dykova, I., and J. Lom. 1979. Histopathological changes in *Trypanosoma danilewskyi* Laveran & Mesnil, 1904 and *Trypanoplasma borreli* Laveran & Mesnil, 1902 infections of goldfish, *Carassius auratus* L. Journal of Fish Disease 2: 381-390.
- 54. El-Sayed, N. M., and P. J. Myler, et al. 2005. The genome sequenceof *Trypanosoma cruzi*, etiologic agent of Chagas disease. Science. **309**: 409-415.
- 55. El-Sayed, N. M., and J. E. Donelson. 1997. African trypanosomes have differentially expressed genes encoding homologues of the *Leishmania* GP63 surface protease. Journal of Biological Chemistry 272: 26742-26748.
- Engstler, M., T. Pfohl, S. Herminghaus, M. Boshart, G. Wiegertjes, N. Heddergott, and P. Overath. 2007. Hydrodynamic flow-mediated protein sorting on the cell surface of trypanosomes. Cell 131: 505-515.
- 57. Engstler, M., G. Reuter, and R. Schauer. 1993. The developmentally regulated trans-sialidase from *Trypanosoma brucei* sialylates the procyclic acidic repetitive protein. Molecular and Biochemical parasitology 61: 1-13
- Fargeas, C., C. Y. Wu, T. Nakajima, D. Cox, T. Nutman, and G. Delespesse. 1992. Differential effect of transforming growth factor beta on the synthesis of Th1- and Th2-like lymphokines by human T lymphocytes. European Journal of Immunology 22: 2173-2176.
- Ferrante, A., and A. C. Allison. 1983. Natural agglutinins to African trypanosomes. Parasite Immunology 5: 539-546.
- 60. Fernandez-Presas, A. M., J. T. Zavala, I. B. Fauser, M. T. Merchant, L. R. Guerrero, and K. Willms. 2001. Ultrastructural damage of *Trypanosome*

cruzi epimastigotes exposed to decomplemented immune sera. Parasitology Research **87:** 619-625.

- Ferreira, V., C. Valck, G. Sanchez, A. Gingras, S. Tzima, M. C. Molina, R. Sim, W. Schwaeble, and A. Ferreira. 2004. The classical activation pathway of the human complement system is specifically inhibited by calreticulin from *Trypanosoma cruzi* Journal of Immunology **172**: 3042-3050.
- 62. Fischer, E., M. A. Ouaissi, P. Velge, J. Cornette, and M. D. Kazatchkine. 1988. gp 58/68, a parasite component that contributes to the escape of the trypomastigote form of *T. cruzi* from damage by the human alternative complement pathway. Immunology **65**: 299-303.
- Frasch, A.C. C. 2000. Functional diversity in members of the transialidase and mucin families in *Trypanosoma cruzi*. Parasitology Today 16: 282-286.
- 64. Garcia, E. S., F. A. Genta, P. Azambuja, and G. A. Schaub. 2010. Interactions between intestinal compounds of triatomines and *Trypanosoma cruzi*. Trends in Parasitology 26: 499-505.
- 65. Garcia, E. S., N. A. Ratcliffe, M. W. Whitten, M. S. Gonzalez, and P. Azambuja. 2007. Exploring the role of insect host factors in the dynamics of *Trypanosoma cruzi*-Rhodnius prolixus interactions. Journal of insect physiology **53**: 11-21.
- Garcia, E. S., and P. Azambuja. 1991. Development and interactions of *Trypanosoma cruzi* within the insect vector. Parasitology Today 7: 240-244.
- 67. Gasbarre, L. C., K. Hug, J. A. Louis. 1980. Murine T lymphocyte specificity for African trypanosomes. I. Induction of a T lymphocytedependent proliferative response to *Trypanosoma brucei*. Clinical and Experimental Immunology **41**: 97-106.
- 68. Gazzinelli, R. T., I. P. Oswald, S. Hieny, S. I. James, and A. Sher. 1992. The microbicidal activity of interferon-γ treated macrophages against *Trypanosoma cruzi* involves an L-arginine-dependent, nitrogen oxide-

mediated mechanism inhibitable by IL-10 and transforming growth factorβ. European Journal of Immunology **22**: 2501-2506.

- Grandgenett, P. M., K. Otsu, H. R. Wilson, M. E. Wilson, and J. E. Donelson. 2007. A function for a specific zinc metalloprotease of African trypanosomes. PLoS Pathogens 3: 1432-1445.
- Grosskinsky, C. M., and B. A. Askonas. 1981. Macrophages as primary target cells and mediators of immune dysfunction in African trypanosomiasis. Infection and Immunity 33: 149-155.
- Gruszynski, A. E., F. J. Van Deursen, M. C. Albareda, A. Best, K. Chaudhary, L. J. Cliffe, L. Del Rio, J. D. Dunn, L. Ellis, K.J. Evans et al. 2006. Regulation of surface coat exchange by differentiating African trypanosomes. Molecular and Biochemical Parasitology 147: 211-223.
- 72. Guirnalda, P., N. B. Murphy, D. Nolan, and S. J. Black. 2007. Anti-*Trypanosoma brucei* activity in Cape buffalo serum during the cryptic phase of parasitemia is mediated by antibodies. International Journal for Parasitology **37**: 1391-1399.
- 73. Gutierrez, F. R. S, T. W. P. Mineo, W. R. Pavanelli, P. M. Guedes, and J. S. Silva. 2009. The effects of nitric oxide on the immune system during *Trypanosoma cruzi* infection. Oswaldo Cruz Memorial Institute 104: 236-45.
- 74. Hamadien, M., N. Lycke, and M. Bakhiet. 1999. Induction of the trypanosome lymphocyte-triggering factor (TLTF) and neutralising antibodies to the TLTF in experimental African trypanosomiasis. Immunology 96: 606-611.
- 75. Hao, Z., I. Kasumba and S. Aksoy. 2003. Proventriculus (cardia) plays a crucial role in immunity in tsetse fly (Diptera: Glossinidiae). Insect Biochemistry and Molecular Biology 33: 1155-1164.
- 76. Hao, Z., I. Kasumba, M. J. Lehane, W. C. Gibson, J. Kwon, and S. Aksoy.2001. Tsetse immune responses and trypanosome transmission:implications for the development of tsetse-based strategies to reduce

trypanosomiasis. Proceedings of National Academy of Sciences U S A **98**: 12648-12653.

- 77. Hertz, C. J., H. Filutowicz, and J. M. Mansfield. 1998. Resistance to the African trypanosomes is IFN-gamma dependent. Journal of Immunology 161: 6775-6783
- 78. Hu, C., and S. Aksoy. 2006. Innate immune reponses regulate trypanosome parasite infection of the tsetse fly Glossina morsitans morsitans. Molecular Microbiology 60: 1194-11204.
- Hu, Y., and S. Aksoy. 2005. An antimicrobial peptide with trypanocidal activity characterized from Glossina morsitans morsitans. Insect Biochemistry and Molecular Biology 35: 105-115.
- Hunter, C. A., T. Slifer, and F. Araujo. 1996. Interleukin-12-mediated resistance to *Trypanosoma cruzi* infection in mice is dependent on tumor necrosis factor alpha and gamma interferon. Infection and Immunity 64: 2381-2386.
- 81. Hunter, C. A., F. W. Jennings, P. G. Kennedy, and M. Murray. 1992. Astrocyte activation correlates with cytokine production in central nervous system of *Trypanosoma brucei brucei*-infected mice. Laboratory Investigation 67: 635-642.
- Horn, D., and J. D. Barry. 2005. The central roles of telomeres and sub telomeres in antigenic variation in African trypanosomes. Chromosome Research 13: 525-533.
- 83. Ilemobade, A. A., D. S. Adegboye, O. Onoviran, J. C. Chima. 1982. Immunodepressive effects of trypanosomal infection in cattle immunized against contagious bovine pleuropneumonia. Parasite Immunology 4: 273-282.
- 84. Islam, A.K.M.N. and P.T.K. Woo. 1991. *Trypanosoma danilewskyi* in *Carassius auratus*: The nature of protective immunity in recovered goldfish. Journal of Parasitology 77: 258-262.
- B. Bouteille, M. Dumas, J. C. Breton, and N. Baumann. 1991. Galactocerebrosides are

antigens for immunoglobulins in sera of an experimental model of trypanosomiasis in sheep. Journal of the Neurological Sciences **101**: 82-86.

- 86. Joiner, K. A., W. D. DaSilva, M. T. Rimoldi, C. H. Hammer, A. Sher, and T. I. Kipnis. 1988. Biochemical characterization of a factor produced by trypomastigotes of *Trypanosoma cruzi* that accelerates the decay of complement C3 convertases. Journal of Biological Chemistry 263: 11327-11335.
- 87. Katzenback, B. A., D. A. Plouffe, G. Haddad, and M. Belosevic. 2008. Administration of recombinant parasite β-tubulin to goldfish (*Carassius auratus* L) confers partial protection against challenge with *Trypanosoma danilewskyi* Laveran and Mesnil, 1904. Veterinary Parasitology 151: 36-45.
- 88. Kemp, M., A. S. Hey, J. A. Kurtzhals, C. B. Christensen, A. Gaafar, M. D. Mustafa, A.A. Kordofani, A. Ismail, A. Kharazmi, and T. G. Theander. 1994. Dichotomy of the human T cell response to *Leishmania* antigens.I. Th1-like response to *Leishmania major* promastigote antigens in individuals recovered from cutaneous leishmaniasis. Clinical and Experimental Immunology **96:** 410–415
- 89. Klemba, M., and D. E. Goldberg. 2002. Biological roles of proteases in parasitic protozoa. Annual Review of Biochemistry **71**: 275-305.
- 90. Kovacs, H., I. D. Campbell, P. Strong, S. Johnson, F. J. Ward, K. B. Reid, P. Eggleton. 1998. Evidence that C1q binds specifically to CH2-like immunoglobulin gamma motifs present in the autoantigen calretiulin and interferes with complement activation. Biochemistry **37**: 17865–17874.
- 91. Kishore, U., R. D. Sontheimer, K. N. Sastry, K. S. Zaner, E. G. Zappi, G. R. Hughes, M. A. Khamashta, P. Strong, K. B. Reid, P. Eggleton. 1997. Release of calreticulin from neutrophils may alter C1q-mediated immune functions. Biochemistry Journal **322**: 543–550.
- Kitani, H., S. J. Black, Y. Nakamura, J. Naessens, N. B. Murphy, Y. Yokomizo, J. Gibson, and F. Iraqi. 2002. Recombinant tumor necrosis

factor alpha does not inhibit the growth of African trypanosomes in axenic cultures. Infection and Immunity **70:** 2210-2214.

- 93. Kobayakawa, T. J., J. Louis, S. Izui, P. H. Lambert. 1979. Autoimmune response to DNA, red blood cells, and thymocyte antigens in association with polyclonal antibody synthesis during experimental African trypanosomiasis. Journal of Immunology 122: 296-301.
- 94. Kollien, A. H., P. J. Wanick, A. J. Nisbet, P. F. Billingsley, and G. A. Schaub. 2004. Activity and sequence characterization of two cysteine proteases in the digestive tract of the reduviid bug *Triatoma infestans*. Insect Molecular Biology 13: 569-579.
- 95. Kollien, A. H., S. Fechner, P. J. Waniek, and G. A. Schaub. 2003. Isolation and characterization of a cDNA encoding for a lysozyme from the gut of the reduviid bug *Triatoma infestans*. Archives of Insect Biochemistry and Physiology 53: 134-145.
- 96. Kollien, A. H, and G. A. Schaub. 2000. The development of *Trypanosoma cruzi* in triatominae. Parasitology Today **16:** 381-387.
- 97. Kollien, A. H., and G. A. Schaub. 1998. The development of *Trypanosoma cruzi* (Trypanosomatidae) in the reduviid bug Triatoma infestans (Insecta): influence of starvation. The Journal of Eukaryotic Microbiology 45: 59-63.
- 98. Kollien, A.H., J. Schmidt, G. A. Schaub. 1998. Modes of association of *Trypanosoma cruzi* with the intestinal tract of the vector *Triatoma infestans*, Acta Tropica **30**:127–141.
- Krautz, G. M., J. C. Kissinger, and A. U. Krettli. 2000. The targets of the lytic antibody response against *Trypanosoma cruzi*. Parasitology Today 16: 31-34.
- 100. Krettli, A. U., and Z. Brener. 1982. Resistance against Trypanosoma cruzi associated to anti-living trypomastigote antibodies. Journal of Immunology 128: 2009-2012.
- 101. Krettli, A. U., P. Weisz-Carrington, and R. S. Nussenzweig. 1979.Membrane-bound antibodies to bloodstream Trypanosoma cruzi in mice:

strain differences in susceptibility to complement-mediated lysis. Clinical and Experimental Immunology 37: 416-423.

- 102. Kulkarni, M. M., C. L. Olson, D. M. Engman, and B. S. McGwire. 2009. *Trypanosoma cruzi* gp63 proteins undergo stage-specific differential posttranslational modification and are important for host cell infection. Infection and Immunity **77**: 2193–2200.
- 103. Kurtzhals, J. A., A. S. Hey, A. Jardim, M. Kemp, K. U. Schaefer, E. O. Odera, C. B. Christensen, J. I. Githure, R. W. Olafson, T. G. Theander et al. 1994. Dichotomy of the human T cell response to *Leishmania* antigens. II. Absent or Th2-like response to gp63 and Th1-like response to lipophosphoglycan-associated protein in cells from cured visceral leishmaniasis patients. Clinical and Experimental Immunology **96**: 416–421.
- 104. LaCount, D. J., A. E. Gruszynski, P. M. Grandgenett, J. D. Bangs, and J. E. Donelson. 2003. Expression and function of the *Trypanosoma brucei* major surface protease (GP63) genes. Journal of Biological Chemistry 278: 24658-24664.
- 105. Lange, U., W. Keilholz, G. A. Schaub, H. Landmann, F. Markwardt, and G. Nowak. 2000. Biochemical characterization of a thrombin inhibitor from the bloodsucking bug *Dipetalogaster maximus*. Haemostasis **29**: 204-211.
- 106. Li, S. Q., M. C. Fung, S. A. Reid, N. Inoue, and Z. R. Lun, 2007. Immunization with recombinant beta-tubulin from *Trypanosoma evansi* induced protection against *T. evansi*, *T. equiperdum* and *T. b. brucei* infection in mice. Parasite Immunology **29**: 191-199.
- 107. Liniger, M., A. Acosta-serrano, J. Van Den Abbeele, C. Kunz-renggli, R. Brun, P. T. Englund, and I. Roditi. 2003. Cleavage of trypanosome surface glycoproteins by alkaline trypsin-like enzyme(s) in the midgut of Glossina morsitans. International Journal of Parasitology. **33**: 1319-1328.
- 108. Lischke, A., C. Klein, Y. D. Stierhof, M. Hempel, A. Mehlert, I. C. Almeida, M. A. Ferguson, and P. Overath. 2000. Isolation and

characterization of glycosylphosphatidylinositol-anchored, mucin-like surface glycoproteins from bloodstream forms of the freshwater-fish parasite *Trypanosoma carassii*. Biochemistry Journal **345:** 693-700.

- 109. Liu, X., and K. P. Chang. 1992. Extrachromosomal genetic complementation of surface metalloproteinase (gp63)-deficient *Leishmania* increases their binding to macrophages Proceedings of National Academy of Science USA 89: 4991-4995.
- 110. Liu, Y., E. Ragaa, Z. Li, l. Nuortio, A. Mustafa, and M. Bakhiet. 1999.
 Interferon-γ and interleukin-12 genes are preferentially expressed during early experimental African trypanosomiasis and suppressed by denervation of the spleen. Scandinavian Journal of Immunology 50: 485-491.
- 111. Lom, J. and I. Dykova. 1992. *Protozoan Parasites of Fishes*.Developments in Aquaculture and Fisheries Science. Vol. 26, Amsterdam: Elsevier Science Publishers.
- 112. Lom, J. Biology of fish trypanosomes and trypanoplasms, in Biology of Kinetoplastida, W.H.R. Lumsden and D.A. Evans, Editors., Academic Press: New York. 1979. p. 269-337.
- 113. Lom, J. Experimental infections of goldfish with blood flagellates, in International Congress on Protozoology. 1973.
- 114. Lopez, N. C., G. Ramirez, M. Rodriguez, G. Ribeiro, J. Orellana, I. Maldonado, A. Albini, D. Anacona, D. Lemus, L. Aquilar, W. Schwaeble, and A. Ferreira. 2010. Antiangiogenic and antitumor effects of *Trypanosoma cruzi* calreticulin. PLoS Neglected Tropical Disease 4: e730.
- 115. Lopez, L., G. Morales, R. Ursic, M. Wolff, and C. Lowenberger. 2003. Isolation and characterization of a novel insect defensin from *Rhodnius prolixus*, a vector of Chagas disease. Insect Biochemistry and Molecular Biology. **33**: 439-47.

- 116. Lucas, R., S. Magez, R. De Leys, I. Fransen, J. P. Scheerlinck, M. Rampelberg, E. Sablon, and P. De Baestselier. 1994. Mapping the lectin-like activity of tumor necrosis factor. Science 263: 814-817.
- 117. Lubega, G. W., D. K. Byarugaba, and R. K. Prichard. 2002. Immunization with a tubulin-rich preparation from *Trypanosoma brucei* confers broad protection against African trypanosomiasis. Experimental Parasitology **102:** 9-22.
- 118. Ma, L., K. Chen, Q Meng, Q. Liu, P. Tang, S. Hu, and J. Yu. 2011. An evolutionary analysis of trypanosomatid Gp63 proteases. Parasitology Research. (In press). DOI:10.1007/s00436-011-2348x)
- 119. Macaskill, J. A., P. H. Holmes, D. D. Whitelaw, I. McConnell, F. W. Jennings, and G. M. Urquhart. 1980. Immunological clearance of 75Selabelled *Trypanosoma brucei* in mice. II. Mechanisms in immune animals. Immunology **40**: 629-635.
- 120. MacLean, L., M. Odiit, and J. M. Sternberg. 2001. Nitric oxide and cytokine synthesis in human African trypanosomiasis. Journal of Infectious Disease 184: 1086-1090.
- 121. Magez, S., M. Radwanska, M. Drennan, I. Fick, T. N. Baral, F. Brombacher, and P. De Baestselier. 2006. Interferon-gamma and nitric oxide in combination with antibodies are key protective host immune factors during *Trypanosoma congolense* Tc13 infections. Journal of Infectious Disease **193**: 1575-1583.
- 122. Magez, S., B. Stijlemans, T. Baral, and P. De Baestselier. 2002. VSG-GPI anchors of African trypanosomes: Their role in macrophage activation and induction of infection-associated immunopathology. Microbes and Infection 4: 999-1006.
- 123. Magez, S., M. Radwanska, B. Stijlemans, H. V. Xong, E. Pays, and P. De Baetselier. 2001. A conserved flagellar pocket exposed high mannose moiety is used by African trypanosomes as a host cytokine binding molecule. Journal of Biological Chemistry 276: 33458-33464.

- 124. Magez, S., B. Stijlemans, M. Radwanska, E. Pays, M. A. Ferguson, and P. De Baetselier. 1998. The glycosyl-inositol-phosphate and dimyristoylglycerol moieties of the glycosylphosphatidylinositol anchor of the trypanosome variant-specific surface glycoprotein are distinct macrophage-activating factors. Journal of Immunology **160**: 1949-1956.
- 125. Maina, N., J. M. Ngotho, T. Were, J. K. Thuita, D. M. Mwangangi, J. M. Kagira, M. Ndung'u, and J. Sternberg. 2004. Pro-inflammatory cytokine expression in the early phase of *Trypanosoma brucei rhodesiense* infection in Vervet monkeys (*Cercopithecus aethiops*). Infection and Immunity **72:** 3063-3065.
- 126. Marcelain, K., A. Colombo, M. C. Molina, I. Ferreira, M. Lorca, J. C. Aguillón, and A. Ferreira. 2000. Development of an immunoenzymatic assay for the detection of human antibodies against *Trypanosoma cruzi* calreticulin, an immunodominant antigen. Acta Tropica **75**: 291-300.
- 127. Martins, R. M., R. Amino, K. R. Daghastanli, I. M. Cuccovia, M. A. Juliano, S. Schenkman. 2008. A short proregion of trialysin, a pore-forming protein of *Triatoma infestans* salivary glands, controls activity by folding the N-terminal lytic motif. The FEBS Journal **275**: 994-1002
- Matthews, K. R., J. R. Ellis, and A. Paterou. 2004. Molecular regulation of the life cycle of African trypanosomes. Trends in Parasitology 20: 40-47.
- Matthews, K. R., and K. Gull. 1994. Cycles within cycles: The interplay between differentiation and cell division in *Trypanosoma brucei*.
 Parasitology Today 10: 473-476.
- Maudlin, I., S. C. Welburn. 1987. Lectin mediated establishment of midgut infections of *Trypanosoma congolense* and *Trypanosoma brucei* in Glossina morsitans. Tropical Medicine and Parasitology 38: 167-170.
- Maudlin, I. 1982. Inheritance of susceptibility to *Trypanosoma* congolense infection in *Glossina morsitans*. Annals of Tropical Medicine and Parasitology **76**: 225-227.

- 132. McKerrow, J. H., E. Sun, P. J. Rosenthal, and J. Bouvier. 1993. The proteases and pathogenicity of parasitic protozoa. Annual Review of Microbiology 47: 821-853
- Mello, C. B., P. Azambuja, E. S. Garcia, and N. A. Ratcliffe, 1996.
 Differential in vitro and in vivo behavior of three strains of *Trypanosoma cruzi* in the gut and hemolymph of *Rhodnius prolixus*. Experimental Parasitology 82: 112-121.
- 134. Moloo, S. K., J. M. Kabata, F. Waweru, and R. H. Gooding. 1998. Selection of susceptible and refractory lines of Glossina morsitans centralis for *Trypanosoma congolense* infection and their susceptibility to different pathogenic Trypanosoma species. Medical and Veterinary Entomology **12**: 391-398.
- Montagna, G., M. L. Cremona, G. Paris, M. F. Amaya, A. Buschiazzo.
 P. M. Alzari, and A. C. Frasch. 2002. The trans-sialidase from the african trypanosome *Trypanosoma brucei*. European Journal of Biochemistry 269: 2941-2950.
- 136. Mortazavi-milani, S. M., C. A. Facer, and E. J. Holborow. 1984. Induction of anti-intermediate filament antibody in rabbits experimentally infected with *Trypanosoma brucei brucei*. Immunology. **52:** 423-426.
- 137. Muranjan, M., V. Nussensweig, and S. Tomlinson. 1998.Characterization of the human serum trypanosome toxin, haptoglobinrelated protein. Journal Biological Chemistry 273: 3884-3887.
- 138. Nagamune, K., A. Acosta-serrano, H. Uemura, R. Brun, C. Kunz-Renggli, Y. Maeda, M. A. Ferguson, T. Kinoshita. 2004. Surface sialic acids taken from the host allow trypanosome survival in tsetse fly vectors. Journal of Experimental Medicine **199**: 1445-1450.
- 139. Nagamune, K., T. Nozaki, Y. Maeda, K. Ohishi, T. Fukuma, T. Hara, R. T. Schwarz, C. Sutterlin, R. Brun, H. Riezman, and T. Kinoshita. 2000. Critical roles of glycosylphosphatidylinositol for *Trypanosoma brucei*. Proceedings of the National Academy of Sciences US A. 97: 10336-10341.

- 140. Namangala, B., W. Noel, P. De Baetselier, I. Brys, and A. Beschin.
 2001. Relative contribution of interferon-γ and interleukin-10 to resistance to murine African trypanosomiasis. Journal of Infectious Disease 183: 1794-1800
- 141. Naresha, S., A. Suryawanshi, M. Agarwal, B. P. Singh, and P. Joshi.
 2009. Mapping the complement C1q binding site in *Haemonchus contortus* calreticulin. Molecular and Biochemical Parasitology 166: 42–46.
- 142. Norris, K. A. 1998. Stable transfection of *Trypanosoma cruzi* epimastigotes with the trypomastigote-specific complement regulatory protein cDNA confers complement resistance. Infection and Immunity 66: 2460-2465.
- 143. Norris, K. A., J. E. Schrimpf, and M. J. Szabo. 1997. Identification of the gene family encoding the 160-kilodalton *Trypanosoma cruzi* complement regulatory protein. Infection and Immunity 65: 349-357
- 144. Norris, K. A., B. Bradt, N. R. Cooper, and M. So. 1991. Characterization of a *Trypanosoma cruzi* C3 binding protein with functional and genetic similarities to the human complement regulatory protein, decayaccelerating factor. Journal of Immunology 147: 2240-2247.
- 145. Oladiran, A., D. Beauparlant, and M. Belosevic. 2011. The expression analysis of inflammatory and antimicrobial genes in the goldfish (*Carassius auratus* L.) infected with *Trypanosoma carassii*. Fish and Shellfish Immunology **31:** 606-613.
- Oladiran, A., and M. Belosevic. 2010. *Trypanosoma carassii* calreticulin binds host complement component C1q and inhibits classical complement pathway-mediated lysis. Developmental and Comparative Immunology 34: 396-405.
- 147. Oladiran, A., and M. Belosevic. 2009. *Trypanosoma carassii* hsp70 increases expression of inflammatory cytokines and chemokines in macrophages of the goldfish (*Carassius auratus* L.). Developmental and Comparative Immunology **33:** 1128–1136.

- 148. Olobo, J. O., C. O. Anjili, M. M. Gicheru, P. A. Mbati, T. M. Kariuki, J. I. Githure, D. K. Koech, and W. R. McMaster. 1995. Vaccination of vervet monkeys against cutaneous leishmaniasis using recombinant *Leishmania major* surface glycoprotein (gp63). Veterinary Parasitology 60: 199-212.
- 149. Olsson, T., M. Bakhiet, B. Hojeberg, A. Ljungdahl, C. Edlund, G. Andersson, H. P. Ekre, W. P. Fung-Leung, T. Mak, H. Wigzell, U. Fiszer, K. Kristensson et al. 1993. CD8 is critically involved in lymphocyte activation by a *T. brucei brucei*-released molecule. Cell **72**: 715-727.
- Olsson, T., M. Bakhiet, and K. Kristensson. 1992. Interactions between *Trypanosoma brucei* and CD8+ T cells. Parasitology Today 8: 237-239.
- 151. Olsson, T., M. Bakhiet, C. Edlund, B. Hojeberg, P. H. Van der Meide, and K. Kristensson. 1991. Bidirectional activating signals between *Trypanosoma brucei* and CD8+ T cells: a trypanosome-released factor triggers interferon-gamma production that stimulates parasite growth. European Journal of Immunology **21**: 2447-2454.
- 152. Overath, P., J. Haag, M. G. Mameza, and A. Lischke. 1999. Freshwater fish trypanosomes: Definition of two types, host control by antibodies and lack of antigenic variation. Parasitology **119**: 591-601.
- 153. Pan, W., O. Ogunremi, G. Wei, M. Shi, and H. Tabel. 2006. CR3 (CD11b/CD18) is a major macrophage receptor for IgM antibodymediated phagocytosis of African trypanosomes: Diverse effect on subsequent synthesis of tumor necrosis factor α and nitric oxide. Microbes and Infection **5**: 1209-1218
- Paulnock, D. M., and S. P. Coller. 2001. Analysis of macrophage activation in African trypanosomiasis. Journal of Leukocyte Biology 69: 685-690.
- 155. Pereira, M. E., A. F. Andrade, J. M. Ribeiro. 1981. Lectins of distinct specificity in *Rhodnius prolixus* interact selectively with *Trypanosoma cruzi*. Science. **211**: 597-600.

- 156. Plouffe, D. A., and M. Belosevic. 2004. Enzyme treatment of *Trypanosoma danilewskyi* (Laveran and Mesnil) increases its susceptibility to lysis by the alternative complement pathway of goldfish *Carassius auratus* (L). Journal of Fish Disease 27: 277-285.
- 157. Previato, J. O., R. Wait, C. Jones, G. A. DosReis, A. R. Todeschini, N. Heise, L. M. Previato. 2004. Glycoinositolphospholipid from *Trypanosoma cruzi*: structure, biosynthesis and immunobiology. Advances in Parasitology 56:1-41.
- Qadri, S. 1962. An experimental study of the life cycle of *Trypanosoma* danilewskyi in the leech, *Hemiclepsis marginata*. Journal of Protozoology 9: 254-258.
- Ribeiro, C. H., N. C. Lopez, G. A. Ramirez, C. E. Valck, M. C. Molina, L. Aguilar, M. Rodriguez, I. Maldonado, R. Martinez, C. Gonzalez et al. 2009. *Trypanosoma cruzi* calreticulin: A possible role in Chagas' disease autoimmunity. Molecular Immunology **46**: 1092-1099.
- 160. Ribeiro, J. M., M. Schneider, T. Isaias, J. Jurberg, C. Galvao, and J. A. Guimaraes. 1998. Role of salivary antihemostatic components in blood feeding by triatomine bugs (Heteroptera). Journal of Medical Entomology. 35: 599-610.
- Ribeiro Carla, M. S., J.S. I. Pontes Maria, S. Bird, M. Chadzinska, M. Scheer, B. M. Lidy Verburg-van Kemenade, F. J. Savelkoul Huub, and G. F. Wiegertjes. 2010. Trypanosomiasis-induced Th17-like immune responses in carp. PLos One 5: e13012.
- 162. Richardson, J. P., R. P. Beecroft, D. L. Tolson, M. K. Liu, and T. W. Pearson. 1988. Procyclin: an unusual immunodominant glycoprotein surface antigen from the procyclic stage of African trypanosomes. Molecular and Biochemical Parasitology **31**: 203-216
- 163. Ruszczyk, A., M. Forlenza, H. F. Savelkoul, and G. F. Wiegertjes. 2008. Molecular cloning and functional charactrization of a cathepsin L-like proteinase from the fish kinetoplastid parasite *Trypanosoma carassii*. Fish Shellfish Immunology **24**: 205-214.

- Robertson, M. 1912. Transmission of flagellates living in the blood of certain fresh-water fishes. Philosophical Transactions of The Royal Society 202: 29-50
- 165. Roditi, I., and M. J. Lehane. 2008. Interactions between trypanosomes and tsetse flies. Current Opinion in Microbiology 11: 245-351
- 166. Roditi, I., and C. Clayton. 1999. An unambiguous nomenclature for the major surface glycoproteins of the procyclic form of *Trypanosoma brucei*. Molecular and Biochemical Parasitology **103**: 99-100.
- 167. Roditi, I., A. Furger, S. Ruepp, N. Schurch, P. Butikofer. 1998. Unravelling the procyclin coat of *Trypanosoma brucei*. Molecular and Biochemical Parasitology **91**: 117-130.
- 168. Ruepp, S., A. Furger, U. Kurath, C. K. Renggli, A. Hemphill, R. Brun and I. Roditi. 1997. Survival of *Trypanosoma brucei* in the tsetse fly is enhanced by the expression of specific forms of procyclin. Journal of Cell Biology **137**:1369-1379.
- 169. Ruszczyk, A., M. Forlenza, H. F. Savelkoul. 2008. Molecular cloning and functional charactrisation of a cathepsin L-like proteinase from the fish kinetoplastid parasite *Trypanosoma carassii*. Fish Shellfish Immunology 24: 205-214.
- Salzet, M. 2002. Leech thrombin inhibitors. Current Pharmaceutical Design 8: 493-503.
- 171. Samudio, M., S. Montenegro-James, M. Cabral, J. Martinez, A. Rojas de Arias, and M. A. James. 1998. Cytokine responses in *Trypanosoma cruzi*infected children in Paraguay. American Journal Tropical Medicine 58: 119-121.
- 172. Sbicego, S., E. Vassella, U. Kurath, B. Blum, and I. Roditi. 1999. The use of transgenic *Trypanosoma brucei* to identify compounds inducing the differentiation of bloodstream forms to procyclic forms. Molecular and Biochemical Parasitology **104**: 311-322.

- 173. Schaub, G. A. 1989. *Trypanosoma cruzi*: quantitative studies of development of two strains in small intestine and rectum of the vector *Triatoma infestans*. Experimental Parasitology **68**: 260-73.
- 174. Schaub, G. A., and P. Losch. 1989. Parasite/host-interrelationships of the trypanosomatids *Trypanosoma cruzi* and Blastocrithidia triatomae and the reduviid bug Triatoma infestans: influence of starvation of the bug. Annals of Tropical Medicine and Parasitology 83: 215-223.
- 175. Schaub, G. A., C. G. Grunfelder, D. Zimmermann, and W. Peters. 1989.
 Binding of lectin-gold conjugates by two Trypanosoma cruzi strains in ampullae and rectum of *Triatoma infestans*. Acta Tropica 46: 291-301.
- 176. Schleifer, K. W, and J. M. Mansfield. 1993. Suppressor macrophages in African trypanosomiasis inhibit T cell proliferative responses by nitric oxide and prostaglandins. Journal of Immunology 151: 5492-5503.
- 177. Sepulveda, P., M. Hontebeyrie, P. Liegeard, A. Mascilli, and K. A. Norris. 2000. DNA-Based immunization with *Trypanosoma cruzi* complement regulatory protein elicits complement lytic antibodies and confers protection against *Trypanosoma cruzi* infection. Infection and Immunity **68:** 4986-4991.
- 178. Schofield, C. J., and C. Galvao, 2009. Classification, evolution, and species groups within the Triatominae. Acta Tropica **110**: 88-100.
- 179. Schofield C.J., J. Jannin, and R. Salvatella. 2006. The future of Chagas disease control, Trends in Parasitology **22**: 583–588.
- Schuster, J. P., and G. A. Schaub. 2000. *Trypanosoma cruzi*: skinpenetration kinetics of vector-derived metacyclic trypomastigotes. International journal for parasitology. **30**: 1475-1479.
- 181. Sharafeldin, A., M. Hamadien, A. Diab, H-I. Li, F-d. Shi, and M. Bakhiet. 1999. Cytokine profiles in the central nervous system and the spleen during the early course of experimental African trypanosomiasis. Scandinavian Journal of Immunology **50**: 256-261.
- Sheader, K., S. Vaughan, J. Minchin, K. Hughes, K. Gull, and G.
 Rudenko. 2005. Variant surface glycoprotein RNA interference triggers a

precytokinesis cell cycle arrest in African trypanosomes. Proceedings of the National Academy of Sciences U S A. **102**: 8716-8721.

- 183. Shi, M., G. Wei, W. Pan, and H. Tabel. 2004. *Trypanosoma congolense* infections: Antibody-mediated phagocytosis by Kupffer cells. Journal of Leukocyte Biology **76**: 399-405.
- 184. Sileghem, M., A. Darji, R. Hamers, M. Van de Winkel, and P. De Baetselier. 1989. Dual role of macrophages in the suppression of interleukin 2 production and interleukin 2 receptor expression in trypanosome-infected mice. European Journal of Immunology 19: 829-835.
- 185. Silva, J. S., G. N. R. Vespa, M. A. G. Cardoso, J. C. S. Aliberti, and F. Q. Cunha. 1995. Tumor necrosis factor alpha mediates resistance to *Trypanosoma cruzi* infection in mice by inducing nitric oxide production in infected gamma interferon-activated macrophages. Infection and Immunity **63:** 4862-4867.
- 186. Silva, J. S., P. J. Morrissey, K. H. Grabstein, K. M. Moller, D. Anderson, and S. G. Reed. 1992. Interleukin 10 and interferon-γ regulation of experimental *Trypanosoma cruzi* infection. Journal of Experimental Medicine **175**: 169-174
- 187. Silva, J. S., D. R. Twardzik, and S. G. Reed. 1991. Regulation of *T. cruzi* infections in vitro and in vivo by transforming growth factor-β (TGF-β). Journal of Experimental Medicine 174: 539-545.
- 188. Smith, A. B., and S. I. Hajduk. 1995. Identification of haptoglobin as a natural inhibitor of trypanocidal activity in human serum. Proceedings of National Academy of Science USA 92: 10262-10266
- 189. Smith, A. B., J. D. Esko, and S. I. Hajduk. 1995. Killing of trypanosomes by the human haptoglobin-related protein. Science 268: 284-286.
- 190. Soteriadou, K. P., M. S. Remoundos, M. C. Katsikas, A. K. Tzinia, V. Tsikaris, C. Sakarellos, and S. J. Tzartos. 1992. The Ser-Arg-Tyr-Asp region of the major surface glycoprotein of *Leishmania* mimics the Arg-

Gly-Asp-Ser cell attachment region of fibronectin. Journal of Biological Chemistry **267:** 13980-13985

- Sternberg, J. M. 1998. Immunobiology of African trypanosomiasis. Chemical Immunology 70: 186-199.
- Stijlemans, B., M. Guilliams, G. Raes, A. Beschin, S. Magez, and P. De Baestselier. 2007. African trypanosomosis: From immune escape and immunopathology to immune intervention. Veterinary Parasitology 148: 3-13.
- 193. Stuart, K., R. Brun, S. Croft, A. Fairlamb, R. E. Gurtler, J. Mckerrow, S. Reed, and R. Tarleton. 2008. Kinetoplastids: Related protozoan pathogens, different diseases. Journal of Clinical Investigation 118: 1301-1310.
- 194. Szoor, B., J. Wilson, H. Mcelhinney, L. Tabernero, and K. R. Matthews. 2006. Protein tyrosine phosphatase TbPTP1: A molecular switch controlling life cycle differentiation in trypanosomes. Journal of cell biology. **175**: 293-303.
- 195. Tachado, D. S., and I. Schofield. 1994. Glycosylphosphatidylinositol toxin of *Trypanosoma brucei* regulates IL-1a and TNF α expression in macrophages by protein tryrosine kinase mediated signal transduction. Biochemical and Biophysical Research Communication **205**: 984-991
- 196. Takahashi, Y., H. Miyamoto, T. Fukuma, T. Nishiyama, T. Araki, and S. Shinka. 1987. In vivo interaction between *Trypanosoma gambiense* and leucocytes in mice. Zentralblat Bakteriologie Mikrobiologie Hygiene A 264: 399-406.
- 197. Taylor, J. E., and G. Rudenko. 2006. Switching trypanosome coats: What's in the wardrobe? Trends in Genetics 22: 614-620
- 198. Templeton, T.J., I. M. Iyer, V. Anantharaman, S. Enomoto, J. E. Abrahante, G. M. Subramanian, S. I. Hoffman, M. S. Abrahamsen, and I. Aravind. 2004. Comparative analysis of Apicomplexa and genomic diversity in eukaryotes. Genome Research 14: 1686-1695.

- 199. Todeschini, A. R., M. P. Nunez, R. S. Pires, M. F. Lopes, J. O. Previato, L. Mendoca-previato, and G. A. Dosreis. 2002. Costimulation of host T lymphocytes by a trypanosomal *trans*-sialidase: involvement of CD43 signalling. Journal of Immunology **168:** 5192-5198.
- 200. Tribulatti, M. V., J. Mucci, N. Van Rooljen, M. S. Leguizamon, O. Campetella. 2005. The *trans*-sialidase from *Trypanosomal cruzi* induces thrombocytopenia during acute Chagas' disease by reducing the platelet sialic acid contents. Infection and Immunity **73**: 201-207.
- Turner, C. M., and J. D. Barry. 1989. High frequency of antigenic variation in *Trypanosoma brucei rhodesiense* infections. Parasitology **99**: 67-75.
- 202. Umezawa, E. S., A. M. Stolf, B. Zingales. 1993. *Trypanosome cruzi*: different surface antigens of trypomastigotes are targets of lytic antibodies. Acta Tropical 54: 41-53
- 203. Ursic-Bedoya, R. J., H. Nazzari, D. Cooper, O. Triana, M. Wolff, and C. Lowenberger. 2008. Identification and characterization of two novel lysozymes from Rhodnius prolixus, a vector of Chagas disease. Journal of Insect Physiology. 54: 593-603.
- 204. Uzonna, J. E., R. S. Kaushik, J. R. Gordon, and H. Tabel. 1998a. Immunoregulation in experimental murine *Trypanosoma congolense* infection: anti-IL-10 antibodies reverse trypanosome-mediated suppression of lymphocyte proliferation in vitro and moderately prolong the lifespan of genetically susceptible BALB/c mice. Parasite Immunology **20**: 293-302.
- 205. Uzonna, J. E., R. S. Kaushik, J. R. Gordon, and H. Tabel. 1998b.
 Experimental murine *Trypanosoma congolense* infections. 1.
 Administration of anti-IFN-gamma antibodies alters trypanosome-susceptible mice to a resistant-like phenotype. Journal of Immunology 161: 5507-5515.
- 206. Uzonna, J. E., R. S. Kaushik, Y. Zhang, J. R. Gordon, and H. Tabel.1998c. Experimental murine *Trypanosoma congolense* infections. II. Role

of splenic adherent CD3+Thy1.2+ TCR-alpha beta- gamma delta- CD4+8and CD3+Thy1.2+ TCR-alpha beta- gamma delta- CD4-8- cells in the production of IL-4, IL-10, and IFN-gamma and in trypanosome-elicited immunosuppression. Journal of Immunology **161:** 6189-6197.

- 207. Vaidya, T., M. Bakhiet, K. I. Hill, T. Olsson, K. Kristensson, and J. E. Donelson. 1997. The gene for T lymphocyte triggering factor from African trypanosomes. Journal of Experimental Medicine 186: 433-438.
- 208. Valck, C., G. Ramirez, N. Lopez, C. H. Ribeiro, I. Maldonado, G. Sanchez, V. P. Ferreira, W. Schwaeble, and A. Ferreira. 2010. Molecular mechanisms involved in the inactivation of the first component of human complement by *Trypanosoma cruzi* calreticulin. Molecular Immunology **47:** 1516-1521.
- 209. Van der Ploeg, L. H., D. Valerio, T. De Lange, A. Bernads, P. Borst, and F. G. Grosveld. 1982. An analysis of cosmid clones of nuclear DNA from *Trypanosoma brucei* shows that the genes for variant surface glycoproteins are clustered in the genome. Nucleic Acids Research 10: 5905-5923.
- 210. Van Voorhis, W. C., I. Barrett, R. Koelling, and G. Farr. 1993. FL-160 proteins of *Trypanosoma cruzi* are expressed from a multigene family and contain two distinct epitopes that mimic nervous tissues. Journal of Experimental Medicine **178**: 681-694.
- 211. Vassella, E., A. Acosta-serrano, E. Studer, S. H. Lee, P. T. Englund, I. Roditi. 2001. Multiple procyclin isoforms are expressed differentially during the development of insect forms of *Trypanosoma brucei*. Journal of Molecular Biology. **312:** 597-607.
- 212. Vassella, E., M. Oberle, S. Urwyler, C. K., Renggli, E. Studer, A. Hemphill, C. Fragoso, P. Butikofer, R. Brun, I. Roditi. 2009. Major surface glycoproteins of insect forms of *Trypanosoma brucei* are not essential for cyclical transmission by tsetse. PLoS One. 4: e4493.
- 213. Vickerman K. 2004. The evolutionary expansion of the trypanosomatid flagellates. International Journal of Parasitology **24:** 1317-1331.

- 214. Vincendeau, P., and B. Bouteille. 2006. Immunology and Immunopathology of African trypanosomiasis. Annals of Brazilian Academy of Science 78: 645-665.
- 215. Waniek, P. J., H. C. Castro, P. C. Sathler, L. Miceli, A. M. Jansen, and C. A. Araujo. 2009. Two novel defensin-encoding genes of the Chagas disease vector *Triatoma brasiliensis* (Reduviidae, Triatominae): gene expression and peptide-structure modeling. Journal of Insect Physiology. 55: 840-848.
- 216. Webb, H., N. Carnall, I. Vanhamme, S. Rolin, J. Van Den Abbeele, S. Welburn, E. Pays, and M. Carrington. 1997. The GPI-phospholipase C of *Trypanosoma brucei* is nonessential but influences parasitemia in mice. Journal of Cell Biology 139: 103-114.
- 217. Welburn, S. C., I. Maudlin. 1999. Tsetse-trypanosome interactions: rites of passage. Parasitology Today. **15:** 399-403.
- 218. Whitten, M., F. Sun, I. Tew, G. Schaub, C. Soukou, A. Nappi, N. Ratcliffe. 2007. Differential modulation of *Rhodnius prolixus* nitric oxide activities following challenge with *Trypanosoma rangeli*, *T. cruzi* and bacterial cell wall components. Insect Biochemistry and Molecular Biology **37:** 440-452.
- 219. WHO, 2002. Control of Chagas' disease, Geneva, World Health Organization, WHO Technical Report Series No. 905, Geneva, Switzerland, 109p.
- 220. WHO, 2001. African trypanosomiasis (sleeping sickness). Report of the scientific working group meeting on African trypanosomiasis. World Health Organisation, Geneva, Switzerland, 169p.
- 221. WHO, 1998. Control and surveillance of African trypanosomiasis.Report of a WHO Expert Committee. World Health Organization, WHO Technical Report Series No. 881, Geneva, Switzerland, 113p.
- 222. Woo, P. T. K., and G. A. Black. 1984. Trypanosoma danilewskyi: Host specificity and host's effect on morphometrics. Journal of Parasitology 70: 788-793.

- 223. Woo, P. T. K. 1981. Acquired immunity against *Trypanosoma danilewskyi* in goldfish, *Carassius auratus*. Parasitology **83:** 343-346.
- 224. Xong, H., I. Vanhamme, M. Chamekh, C. E. Chimfwembe, J. Van Den Abbeele, A. Pays, N. Van Meirvenne, R. Hamers, P. De Baetselier, and E. Pays. 1998. A VSG expression site-associated gene confers resistance to human serum in *Trypanosoma rhodesiense*. Cell **95**: 839-846.
- 225. Xu, D., S. J. McSorley, S. N. Chatfield, G. Dougan, and F. Y. Liew. 1995. Protection against *Leishmania major* infection in genetically susceptible BALB/c mice by gp63 delivered orally in attenuated *Salmonella typhimurium* (AroA- AroD-). Immunology **85:** 1-7.
- 226. Yoshida, N. 2008. *Trypanosoma cruzi* infection by oral route: how the interplay between parasite and host components modulates infectivity. Parasitology International 57: 105-109.
- 227. Zhang, L., and R. L. Tarleton. 1996. Characterization of cytokine production in murine *T. cruzi* infection by in situ immunochemistry: Lack of association between susceptibility and T helper type 2 cytokine production. European Journal of Immunology **26:** 102-109.
- 228. Ziegelbauer, K., M. Quinten, H. Schwarz, T. W. Pearson, P. Overath.
 1990. Synchronous differentiation of *Trypanosoma brucei* from bloodstream to procyclic forms in vitro. European Journal of Biochemistry. 192: 373-378.

CHAPTER 3

MATERIALS AND METHODS

3.0 General procedures

3.1 Fish and parasite culture conditions

3.1.1 Fish

Goldfish (*Carassius auratus* L.) were purchased from Mt. Parnell Fisheries Inc. (Mercersburg, PA) or Aquatic Imports (Calgary, Alberta, Canada) and maintained at the Aquatic Facility of the Department of Biological Sciences, University of Alberta, according Canadian Council for Animal Care (CCAC) guidelines. The fish were kept at 20°C in a flow-through water system on a simulated natural photoperiod, and fed to satiation daily with trout pellets. The fish were acclimated to this environment for at least three weeks prior to use in experiments. All of the fish ranged from 10 to 15 cm in length and whenever possible an equal number of both sexes were used in experiments. Fish were anaesthetized by immersion in tricaine methane sulfonate solution before bleeding, clipping and infection.

3.1.2 Fish blood and serum preparation

To obtain goldfish serum, blood was collected through caudal vein of fish anaesthetized with tricane methanosulfonate (TMS, Syndel Laboratories) using a 23-guage needle fitted to a syringe. Blood was allowed to clot for overnight at 4°C and serum was isolated by centrifugation at 3000 x g for 30 minutes. Serum used for culturing trypanosomes was heat-inactivated at 56°C for 30 minutes, filter sterilized (Millipore 0.22µm), and stored at -20°C until used. Immune serum was obtained from individual goldfish infected with 6.25×10^6 *in vitro* grown parasites. Sera were collected 7, 14, 21, 28, and 56 days post-infection, pooled and stored at -20° C until used.

Serum for cultivation of goldfish monocytes and macrophages was obtained from common carp (*Cyprinus carpio*). Fish used for obtaining blood and serum preparations were maintained strictly for this purpose and were bled every 6-8 weeks. Carp serum was prepared as described for goldfish serum.

3.1.3 Parasites

The parasite used in this study was bloodstream form *T. carassii* strain TrCa (=*T danilewskyi* Laveran and Mesnil 1904). The parasite was isolated by Dr. J. Lom in 1977 from a crucian carp. Isolation and cultivation of this trypanosome strain has been previously described [4, 16]. Parasites cultures were maintained *in vitro* at 20°C in TDL-15 medium [2], supplemented with 10% heat inactivated goldfish serum. Components of TDL-15 medium are shown in Table 3.1. Parasites were harvested from 5-7day old cultures for use in the experiments.

3.1.4 Purification of parasites from blood

Blood from infected fish was mixed with tri-sodium citrate (100 mM trisodium citrate, 40 mM glucose, pH 7.3), to prevent clotting. Samples were centrifuged at 750 x g, and the supernatant was removed. The trypanosomes located in the layer on top of the red blood cell pellet were removed and gently resuspended in separation buffer (57 mM Na₂HPO₄, 3 mM NaH₂PO₄, 44 mM NaCl, 4 mM KCl, 5 mM glucose, pH 8). Parasites were re-suspended twice to ensure most of the blood cells were removed. Following the last centrifugation step, the vial was tilted for 5 minutes at a 20-degree angle, to allow parasites to separate from the pelleted cells and settle on the side- wall of the vial. The separated parasites were removed and used immediately in the experiments.

3.1.5 Infection of fish

Prior to use in the experiments all fish were examined for the presence of natural hemoflagellate infections. Blood samples were collected from individual fish and examined for the presence of hemoflagellates using the haematocrit centrifugation technique, as described previously [2, 10, 18]. All fish used in experiments were free of natural hemoflagellates.

Fish were infected intraperitoneally (ip) with 6.25×10^6 *in vitro* cultured parasites suspended in 100 µL of TDL-15 medium (Table 3.1) using a 25-guage needle and a tuberculin syringe. This dose was selected based on previously published results from our laboratory, which showed that all fish inoculated with this number of parasites became infected [16].

3.1.6 Assessment of infection

In order to determine the number of circulating parasites during the course of infection, 100 μ L of blood samples were collected from the caudal vein of the fish. Four microliters of blood was collected and mixed with 96 μ L of tri-sodium citrate anticoagulant solution, and the number of trypanosomes was determined by counting the parasites using a microscope (400x), and an improved Neubauer hemocytometer fitted with a glass cover slip (22 x 22 mm).

Rarely, fish had very low number of circulating parasites that could not be detected using a hemocytometer. In this case, blood samples were collected in heparinized capillary tube and examined for parasites using hematocrit centrifugation technique [18]. When the number of parasites in the heparinized capillary tube was difficult to count accurately, the blood was placed in a separate capillary tube, centrifuged, and re-examined. The detection limit for assessment of parasites in the blood of fish was 10³ trypanosomes per mL.

3.1.7 Preparation of parasite antigens

3.1.7.1 Excretory/secretory (ES) products

In vitro cultured trypanosomes were washed twice at 400 x g for 10 minutes in serum free culture medium and re-suspended to a density of 4.28 x 10^8 /mL. Parasites were incubated at 20°C for 1 hour and the cultures were then centrifuged at 12,000 x g for 10 minutes. Supernatants containing ES products were removed. To ensure removal of any remaining parasites the supernatant was re-centrifuged again at 12,000 x g for 10 minutes, filtered (0.22 µm), concentrated 10 fold and stored at -20°C until used.

3.1.7.2 Whole cell lysates

Trypanosome cell lysates were prepared by washing *in vitro* cultured parasites twice in serum free culture medium at 400 x g for 10 minutes. Parasites were re-suspended in culture medium containing protease inhibitors (1 mM phenylmethylsulfonyl fluoride-PMSF, 1 μ g/mL leupeptin, 1 μ g/mL antipain, 5 μ g/mL aprotinin) and subjected to 3 cycles of freeze-thaw.

3.1.8 Detergent solubilization of membrane proteins

Membrane proteins were prepared from 10^{10} bloodstream parasites using Triton X-114 (Sigma) [5]. Parasite suspensions were centrifuged at 400 x *g* for 10 minutes and washed 2 twice in serum free medium at 400 x *g* for 10 minutes. Parasites were re-suspended in 1 mL of PBS containing protease inhibitors (1 mM phenylmethylsulfonyl fluoride, 1 µg/mL pepstatin, 1 µg/mL leupeptin, 1 µg/mL antipain, 5 µg/mL aprotinin) on ice. One mL of Triton X-114 containing protease inhibitors was added and the suspension incubated on ice for 1 hour with intermittent mixing. Samples were centrifuged at 100,000 x *g* for 1 hour at 4°C using ultracentrifuge (Beckman Instruments). The supernatants containing solubilized membrane proteins were collected and used immediately for twodimensional electrophoresis (2-DE). Protein concentrations were estimated using Micro BCA protein assay (Pierce).

3.1.9 Anti-carp IgM

Hybridoma cell supernatants containing monoclonal antibodies directed towards an epitope of the heavy chain of carp Ig (designated WCI 12) were obtained from previously grown cultures [3]. The specificity of monoclonal antibody for goldfish immunoglobulin was assessed using SDS-PAGE and immunoblotting. Immunoblotting revealed a single band of approximately 80 kDa further supporting the previous findings that indicate the size of the heavy chain to be approximately 79 kDa [17]

3.2 Goldfish kidney-derived monocyte/macrophage cultures

3.2.1 Culture medium

Goldfish leukocytes were cultured in MGFL-15 medium [15]. The components of the medium are shown in Table 3.2. Complete medium (C-MGFL-15) comprised of MGFL containing 100 U/mL penicillin, 100 µg/mL streptomycin (Gibco), 100 µg/mL gentamicin (Gibco), 10% newborn calf serum (Hyclone) and 5% carp serum. Cultures were maintained at 20°C.

3.2.2 Isolation of goldfish kidney leukocytes

Leukocytes were isolated from goldfish kidneys as described previously [12, 15]. Briefly, fish were anaesthetized using TMS (approximately 40 mg/mL) and the kidneys were removed aseptically. The kidneys were placed in homogenizing solution (MGFL-15, Pen/Strep and heparin) and gently passed through sterile stainless steel screens using homogenizing solution. The resulting cell suspension was layered over 51% Percoll (Sigma) and centrifuged at 400 x *g* for 25 minutes at 4°C. Cells at the medium-Percoll interface were transferred into clean tubes and washed twice with MGFL-15 by centrifugation at 200 x *g* for 10 minutes at 4°C. The resulting cell pellets were re-suspended in complete medium (CMGFL-15) and cultured at 20°C. Goldfish macrophage cultures were established by seeding freshly isolated kidney leukocytes (1.8-2 x 10^6 cells per flask from individual fish) into 75 cm² tissue culture flasks containing 15 mL of CMGFL-15 and 5 mL of cell-conditioned medium (CCM) from previous cultures. These primary kidney macrophage (PKM) cultures contain heterogenous

populations of cells including progenitors, monocytes and mature macrophages as determined by flow cytometry, morphology, cytochemistry and function [12]. Leukocytes that were cultured for 2-4 for days without CCM were predominantly monocytic cells, while macrophages were obtained from 6-8 days cultures, before use in assays.

3.2.3 Preparation of cell-conditioned medium (CCM)

Cultures of primary kidney macrophages maintained at 20°C for 6-8 days were centrifuged at 200 x g for 10 minutes at 4° and the resulting CCM (supernatants) from individual cultures were pooled, filter-sterilized (0.22 μ m) and stored at 4°C until use.

3.3 Molecular procedures

3.3.1 Isolation of total RNA

Total RNA was extracted from parasite grown for 5-7 days *in vitro* using TRIZOL reagent (Invitrogen) according to manufacturer's instructions and RNA samples were stored at -80°C until used. The purity of RNA was confirmed by calculating the ratio of the absorbance at 260 nm and 280 nm. In all cases, this ratio was approximately 2.0, which is the generally acceptable value for pure RNA. The extracted RNA was reverse transcribed into cDNA using the Superscript II cDNA synthesis kit (Invitrogen) according to manufacturer's recommendation. The cDNA synthesized was used in polymerase chain reactions (PCR). 3.3.2 Cloning of *T. carassii* genes

3.3.2.1 T. carassii hsp70 (Tcahsp70)

Hsp70 fragment was amplified using primers designed to *T. carassii* spliced leader sequence and hsp70 peptides (IFDLGGGTFDVT LL and VVLVGGSTRIPKVMQ) identified by mass spectrometry. Table 3.3 shows the list of primers used. The rest of the sequence was amplified using 3' RACE, according to manufacturer's protocols (Clontech) and the full ORF was obtained. The sequence for Tcahsp70 has been submitted to GenBank (FJ970030).

3.3.2.2 *T. carassii* calreticulin (TcaCRT)

TcaCRT was originally identified in membrane protein fraction separated by 2-dimensional gel electrophoresis (2-DE) and identified by mass spectrometry. TcaCRT peptides identified in mass spectrometry analysis are (FYADAEK , CGGGYIK, YWLMFGPDR, EAPMIPNPK, QIPNPAYK). These peptides matched *T. cruzi* calreticulin (TcCRT) with accession number (XM_799098). TcaCRT fragment was amplified using primers (Table 3.3) designed to *T. carassii* spliced leader sequence and peptides identified by mass spectrometry. The rest of the sequence was amplified using 3' RACE, according to manufacturer's protocols (Clontech) and full ORF was obtained. The full sequence for TcaCRT has been submitted to GenBank (GQ406204).

3.3.2.3 T. carassii glycoprotein 63 (Tcagp63)

Tcagp63 fragment was amplified using primers (Table 3.3) designed to T.

carassii spliced leader sequence and a highly homologous region of trypanosome gp63 sequences. The rest of the sequence was amplified using 3' RACE, according to manufacturer's protocols (Clontech) and the full ORF was obtained. The full sequence for Tcagp63 has been submitted to GenBank (JN660149).

3.3.2.4 In silico analysis of T. carasssii genes

Protein sequence alignments were performed with Clustal W software. Phylogenetic analysis of Tcagp63 was conducted using Clustal X software neighbor joining method for relationship calculation and bootstrapped 10,000 times, with values expressed as percentages. Phylogenetic analysis was visualized using the njplot program. As indicated previously, all nucleotide and protein sequences used in this thesis have been submitted to GenBank. Protein sequences used for phylogenetic analysis of Tcagp63 and their accession numbers are: *T. cruzi* (XP_817187), *T. brucei* (XP_846998), *T. congolense* (CCC95446), *T. vivax* (CCC53361), *L. major* (XP_001684335), *L. mexicana* (CAA45733), *L. donovani donovani* (ACT31401), *L. infantum* (XP_003392667), *L. braziliensis* (XP_001562828), *T. carsssii* Cathepsin L (ABQ23397), *T. cruzi* cruzipain (AAF75547), *T. brucei* cysteine peptidase (XP_845218). *T. carsssii* Cathepsin L, *T. cruzi* cysteine protease and *T. brucei* cysteine peptidase were used as outgroups.

3.4 Using prokaryotic expression system for production of *T. carassii* recombinant proteins

3.4.1 Cloning of *T. carassii* genes into pET SUMO expression vectors

The production of recombinant parasite proteins characterized in this thesis has been described previously [13, 14]. *T. carassii* hsp70, calreticulin and gp63 were amplified from parasite cDNA using primers that allow cloning into the pET SUMO expression vector (Invitrogen). The PCR products were gel purified (QIAquick gel extraction kit) and ligated into the pET SUMO vector (Invitrogen). The recombinant molecules were used to transform competent *Escherichia coli* (One Shot chemically competent cell), plated onto LB plates containing kanamycin (50 µg/mL) and incubated overnight at 37°C. Recombinant clones were identified by colony PCR and cultured in LB broth containing kanamycin (50 µg/mL). Plasmid DNA was purified from recombinant clones using a QIAquickSpin miniprepkit (Qiagen) and sequenced using the vector specific primers (Table 3.3) to verify that insert was in correct orientation and in frame. The pET SUMO vector encodes an N-terminal 6XHis tags for purification and detection of the recombinant molecule.

3.4.2 Pilot expression studies of recombinant Tcahsp70, TcaCRT, Tcagp63

Recombinant plasmids containing Tcahsp70, TcaCRT or Tcagp63 insert were transformed into BL21 Star One Shot *E. coli* cells (Invitrogen) and the whole transformation mixture cultured overnight in 5 mL LB containing kanamycin (50 μ g/mL). In this pilot run, 500 μ L of the overnight culture was used to inoculate 10 mL LB containing kanamycin (50 μ g/mL), grown to mid-log phase (OD₆₀₀ of 0.5-0.8) and split into two 5 mL cultures. IPTG (1 mM) was added to one of the cultures to induce expression of the recombinant protein. Samples were taken every 2 hours to determine the optimal induction times for each respective protein. Following sampling, bacteria lysate supernatant and pelleted bacteria fractions were resolved by SDS-PAGE and visualized by western blotting using mouse monoclonal anti-polyhistidine antibody (1:5000) (invitrogen) against the N-terminal 6X His tag on the recombinant proteins. From these pilot studies, I determined that the recombinant hsp70 and CRT were present in both supernatant and pellet fraction, gp63 was present in the pellet fraction, and the optimal induction time for all proteins to be 2 hours.

3.4.3 Scaled up expression of recombinant Tcahsp70 and TcaCRT in E. coli

Bacterial culture expressing the desired recombinant proteins were scaled up and incubated for the 2 hours optimal induction time. The pilot expression showed these proteins were present in both supernatant and pellet fraction although the majority of the proteins were present in the pellet fraction. The induced cells in the scaled up cultures were centrifuged at 10,000 x g for 10 minutes, the supernatants were removed and the cell pellets frozen at -20° C until needed. Cells were suspended in lysis buffer (5 mL of 10X FastBreak cell lysis reagent (Promega) in 45 mL of denaturing wash buffer (100 mM Hepes, 10 mM imidazole, 7 M urea, 0.1 M PMSF; pH 7.5) and incubated with Magne-His Niparticles (Promega) as recommended by the manufacturer. Ni-particles bound to

85
the recombinant proteins were retained by a PolyATtrack system 1000 magnet (Promega), were extensively washed with the wash buffer described above. The supernatants were discarded and the beads washed 3 times under denaturing condition using the denaturing wash buffer described above. The recombinant protein was eluted from the beads using 500 mM imidazole. Western blot using monoclonal anti-His antibody was performed to confirm the presence of the recombinant protein. The protein was renatured in 10 volumes of renaturation buffer (4 mM reduced glutathione, 2 mM oxidized glutathione, 50 mM sodium borate, 5 mM EDTA, pH 8.5) overnight and dialyzed against 4 L of 1X PBS overnight using 3 kDa MWCO Snakeskin dialysis tubing (Pierce), Protein concentration was done in dialysis tubing in polyethylene glycol flakes for 8 hours and protein was further dialyzed overnight against 1X PBS to remove traces of imidazole and urea. The recombinant protein was passed through an EndoTrap Red endotoxin removal column (Cambrex) as recommended by the manufacturer to remove bacteria LPS. Recombinant protein was analyzed using the *Limulus* amebocyte lysate kit (Cambrex) and found to be endotoxin-free (< 0.1 EU/mL). Protein concentration was determined using a Micro BCA Protein Assay Kit (Pierce) and the presence of the recombinant protein verified by Western blot. The identity of the protein was confirmed by mass spectrometry. A vector control prepared by growing *E. coli* cells expressing an empty pET SUMO vector and following exact procedure described above was used as control in functional assays. The vector control had no protein contents as measured by Micro BCA Protein Assay (Pierce). The concentrations of rTcahsp70 used in functional assays

were based on previous studies [1]. Also, the concentrations of rTcaCRT used in hemolysis assay were based on previous studies [6].

3.4.4 Scaled up expression of recombinant Tcagp63 in E. coli

The pilot studies showed substantial degradation of the recombinant gp63 produced by vector expressing full ORF and ORF lacking the signal peptide. Stable gp63 protein was obtained from vector expressing fragment lacking Nterminal portion of the protein containing both the signal peptide and the catalytic domain. Bacteria culture expressing the desired recombinant protein was scaled up and incubated for the 2 hours optimal induction time. In the pilot expression study, recombinant gp63 was present in the pellet fraction. Therefore, the induced cells in the scaled up cultures were centrifuged at $10,000 \ge g$ for 10 minutes, the supernatants were removed and the cell pellets was frozen at -20°C until needed. Cells were suspended in lysis buffer [5 mL of 10X FastBreak cell lysis reagent (Promega) in 45 mL of denaturing wash buffer (100 mM Hepes, 10 mM imidazole, 7 M urea, pH 7.5] and incubated with Magne His Ni-particles (Promega) as recommended by the manufacturer. The supernatants were discarded and the beads washed 3 times under denaturing condition using the denaturing wash buffer described above. The recombinant protein was eluted from the beads using 500 mM imidazole. Western blot using monoclonal anti-His antibody was performed to confirm the presence of the recombinant protein. The protein was re-natured in 10 volumes of renaturation buffer (40 mM Tris, pH 8) overnight and dialyzed against 4 L of 1X PBS using 3 kDa MWCO Snakeskin

dialysis tubing (Pierce). Protease inhibitor cocktail (AEBSF 1 mM, aprotinin 800 nM, bestatin 50 μ M, E64 15 μ M, leupeptin 20 μ M, pepstatin A 10 μ M) were added to all buffers to prevent protein degradation. Protein concentration was done in dialysis tubing in polyethylene glycol flakes for 8 hours and protein was further dialyzed overnight against 1X PBS to remove traces of imidazole and urea. The recombinant protein was passed through an EndoTrap Red endotoxin removal column (Cambrex) as recommended by the manufacturer to remove bacteria LPS. Recombinant protein was analyzed using the *Limulus* amebocyte lysate kit (Cambrex) and found to be endotoxin-free free (< 0.1 EU/mL). Protein concentration was determined using a Micro BCA Protein Assay Kit (Pierce) and the presence of the recombinant protein verified by Western blot. The identity of the protein was confirmed by mass spectrometry. A vector control prepared by growing E. coli cells expressing an empty pET SUMO vector and following exact procedure described above was used as control in functional assays. The vector control had no protein contents as measured by Micro BCA Protein Assay (Pierce) however, volumes corresponding to recombinant proteins were used in functional assays. The concentrations of rTcagp63 used in functional assays were based on previous studies [9].

3.5 Quantitative expression analysis (Q-PCR) of goldfish immune genes

3.5.1 Q-PCR of monocytes and macrophages genes

The Q-PCR primer sequences used in this thesis are shown in Table 3.4. Expression primers design, validation Q-PCR conditions have been described previously [7]. Briefly, primers were designed using Primer Express software (Applied Biosystems). Primers were validated by measuring relative quantification efficiency using a serially diluted goldfish cDNA. Primers were deemed acceptable for use if the experimentally derived R² values from the cDNA dilution curves were greater than 0.980. In most instances, subsequent to primer validation experiments, Q-PCR products were resolved by agarose gel electophoresis and the resulting products sequenced to confirm primer specificities.

Total RNA was extracted from *in vitro* cultured goldfish monocytes (2-4 days) and macrophages (6-8 days) using TRIzol reagent (Invitrogen) according to manufacturer's instructions and RNA samples were stored at -80°C until used. The extracted RNA was reverse transcribed into cDNA using the Superscript II cDNA synthesis kit (Invitrogen) according to manufacturer's recommendation. The cDNA synthesized was used in quantitative real-time PCR reactions. In all cases, equal amounts of RNA and cDNAs were used in the Q-PCR reaction. Thermocycling parameters were as follows: 95°C for 2 minutes, followed by 40 cycles of 95°C for 15 seconds and 60°C for 1 minute. Analyses of gene expression data were performed using 7500 Fast software (Applied Biosystems).

3.5.1.1 Pro-inflammatory cytokine expression in goldfish macrophages treated with recombinant *T. carassii* hsp70

Macrophage cultures obtained from four individual fish (n = 4) were cultured for 8 days and seeded into individual wells of 96-well plates at a density of 1×10^{6} cells per well. Cells were treated with medium alone or 5,000 ng/mL of *T. carasii* hsp70 (rhsp70) for 3 and 6 hours. Each treatment group consisted of 1×10^{6} cells in a final volume of 400 µL of complete medium. Total RNA was isolated from the cells at indicated times and RNA was reverse transcribed into cDNA as indicated previously. Changes in expression pattern of cytokines IFN γ , TNF α -1, TNF α -2, IL-1 β -1, IL-1 β -2; IL-12-p35; chemokines CXCL-8, CCL-1; and anti-microbial genes iNOS isoforms A and B were examined. Relative expression was determined in relation to elongation factor 1 alpha (EF-1 α). All expression data were normalized using the baseline expression at 0 hour time point.

To confirm that hsp70 induced cytokine expression was due to cell surface signaling, macrophages were pre-treated with either culture medium only, or 10mM pronase for 30 minutes [1]. Cells were washed twice with incomplete medium and re-suspended in complete medium containing 5,000 ng/mL of rhsp70. Total RNA was isolated at 0, 3 and 6 hours post treatment and expression data normalized to baseline expression at 0 hour time point.

3.5.1.2 Immune gene expression in macrophages exposed to rTcagp63 and *A*. *salmoncida*

Goldfish macrophage cultures obtained from four individual fish (n = 4) were cultured for 8 days and seeded into individual wells of 96-well plates at a density of 1×10^6 cells per well. Goldfish monocytes cultures obtained from four-day-old cultures and seeded into individual wells of 96-well plates at a density of

 1×10^{6} cells per well. Cells were treated with medium alone or 10 µg/mL of recombinant *T*. carassii gp63 (rTcagp63) at 20°C for 2 hours,, washed twice and then stimulated with heat-killed *A. salmonicida* at 20°C for 6 hours. Each treatment group consisted of 1×10^{6} cells in a final volume of 400 µL of complete medium. Total RNA was isolated from the cells and RNA was reverse transcribed into cDNA as indicated previously. Changes in expression pattern of select immune genes (iNOS, TNFa1 and 2, IL10 and NADPH oxidase component genes) were examined. Relative expression was determined in relation to elongation factor 1 alpha (EF-1a). All expression data were normalized using the baseline expression of cells incubated in medium alone.

3.5.1.3 Immune gene expression in macrophages treated with rTcagp63

Goldfish macrophage cultures obtained from four individual fish (n = 4) were cultured for 8 days and seeded into individual wells of 96-well plates at a density of 1×10^6 cells per well. Cells were treated with 10 µg/mL of rTcagp63 for 0, 3 and 6 hours at 20°C. Each treatment group consisted of 1×10^6 cells in a final volume of 400 µL of complete medium. Total RNA was isolated from the cells at the indicated times and RNA was reverse transcribed into cDNA as described previously. Changes in expression pattern of select immune genes (iNOS A and B, TNF α 1 and 2, IL10) were examined. Relative expression was determined in relation to elongation factor 1 alpha (EF-1 α). All expression data were normalized using the baseline expression of cells incubated in medium alone.

3.5.2 Quantitative expression analysis of immune genes in goldfish tissues

Spleen, liver and kidney tissues were removed from infected (n = 4) and non-infected (n=4) fish on days 7, 14, 21, 28 and 56 pi, and were immediately flash frozen in liquid nitrogen and stored at -80°C until RNA isolation. Total RNA was extracted from goldfish tissues and RNA samples from different tissues were reverse transcribed into cDNA as described previously. In all cases, the purity of RNA was confirmed. Changes in the expression patterns of a panel of pro-inflammatory genes encoding IFN- γ , TNF α 1 and TNF α 2, IL-1 β -1 and IL-1 β -2, IL-12-p35 and IL-12-p40, CCL1, CXCL8, anti-inflammatory genes encoding IL-10 and TGF β , and antimicrobial genes encoding iNOS A and iNOS B were examined. Relative expression was determined in relation to elongation factor-1 alpha (EF-1 α). All expression data were normalized using the baseline expression in non-infected control fish for each gene. Results were provided as fold change in expression relative to non-infected control for each gene.

3.6 Measurement of the production of reactive oxygen and nitrogen intermediates: nitric oxide and respiratory burst assays

Individual cultures of primary macrophages were obtained from kidney of goldfish and the cells were cultured for 6 to 8 days (n = 7). Cells were harvested from cultures and seeded into individual wells of 96-well plates at a density of 3 x 10^5 cells per well. Cells were treated with culture medium alone (control) or with either heat-killed *A. salmonicida* or different concentrations of *T. carassii* and incubated at 20°C for 48 hours. Parasites were washed twice in serum-free culture

medium before use. Nitrite production was determined using the Griess reaction [8]. Briefly 1% sulphanilamide and 0.1% N-naphthyl-ethylenediamine were added to cultures and absorbance was measured at 540 nm. The amount of nitrite in culture supernatants was determined using nitrite standard curve.

For respiratory burst assay, individual cultures of goldfish monocytes were obtained from kidney of goldfish and cells were cultured for 3 to 4 days. Cells from 6 individual fish (n = 6) were seeded into 96-well plates at a density of 3×10^5 cells per well and incubated with medium alone (control), or with either macrophage activating factor (MAF) or different concentrations of *T. carassii*. ROI assays were incubated at 20°C for 48 hours. NBT (2 mg/mL, Sigma) and PMA (final conc. 100 ng/mL, Sigma) in PBS were added to the cultures and incubate at room temperature for an additional 30 minutes. The plates were than centrifuged at $400 \times g$ for 10 minutes, the supernatants aspirated and cells in the pellet fixed with absolute methanol. Non-reduced NBT was removed by washing with methanol and reduced NBT was dissolved with 2 M KOH. DMSO was added to induce the colorimetric response and the plates were read at 630 nm. Absorbance from cells alone (no PMA) was subtracted from treatment values to factor in background NBT reduction.

3.7 Functional analysis of the effects of *T. carassii* proteins on antimicrobial responses of goldfish monocytes and macrophages

3.7.1 Assessment of the ability of rTcahsp70 to induce nitric oxide response of goldfish macrophages

Individual cultures of primary macrophages were obtained from kidney of goldfish and the cells were cultured for 6 to 8 days (n = 7). Cells were harvested from cultures and seeded into individual wells of 96-well plates at a density of 3 x 10^5 cells per well. Cells were treated with culture medium alone (control) or with either heat-killed A. salmonicida or rhsp70 (500 ng/mL, 5,000 ng/mL or 10,000 ng/mL) and incubated at 20°C for 72 hours. Nitrite production was determined using the Griess reaction as described previously [8]. To ascertain whether the results obtained were not due to other contaminating proteins, E. coli bacterial lysates were prepared and subjected to the same purification protocol as rhsp70containing lysates. Following purification regimen, these lysates did not contain protein as indicated by micro BCA protein assay. Furthermore, separate goldfish macrophage cultures obtained from 4 fish (n = 4) were treated with either medium only (negative control), heat-killed A. salmonicida (positive control), rhsp70 (5,000 ng/mL, 10,000 ng/mL) or bacterial lysates (17 µL, 33 µL, equivalent volumes of rhsp70 lysates that corresponded to 5,000 and 10,000 ng/mL of rhsp70) and assayed for nitric oxide response using Griess reaction, as described previously [8].

3.7.2 Assessment of the effects of rTcagp63 on *A. salmonicida*-induced nitric oxide response of goldfish macrophages.

Individual cultures of primary macrophages were obtained from kidney of goldfish and the cells were cultured for 7 to 8 days. Cells from 6 individual fish (n = 6) were harvested from cultures and seeded into individual wells of 96-well plates at a density of 3×10^5 cells per well. Cells were incubated with culture medium alone (negative control), heat-killed A. salmonicida (positive control), rTcagp63 (1 μ g/mL, 5 μ g/mL or 10 μ g/mL) plus heat-killed A. salmonicida at 20°C for 48 hours. Nitrite production was determined using the Griess reaction as described previously [8]. To ascertain that the results obtained were not due to other contaminating proteins, E. coli bacterial lysates were prepared and subjected to the same purification protocol as rTcagp63-containing lysates. Following purification regimen, these lysates did not contain protein as indicated by micro BCA protein assay. Furthermore, separate goldfish macrophage cultures obtained from 6 fish (n = 6) were treated with medium alone (negative control), heat-killed A. salmonicida (positive control), rTcagp63 (10 µg/mL) or bacterial lysates (4.5 μ L, equivalent volumes of rTcagp63 lysates that corresponded to 10 μ g/mL of rTcagp63) plus heat-killed A. salmonicida and assayed for nitric oxide response using Griess reaction. Also, recombinant goldfish TGF β , purified using the same procedure as described for Tcagp63 were used in the NO assay at $10 \mu g/mL$.

3.7.3 Assessment of the effects of rTcagp63 on *A. salmonicida*-induced respiratory burst response of goldfish monocytes.

Individual cultures of goldfish monocytes were obtained from kidney of goldfish and cells were cultured for 3 to 4 days. Cells from 6 individual fish (n = 6) were incubated with medium alone (control), *A. salmonicida* alone or *A. salmonicida* plus different concentration of rTcagp63. In another experiment cells were treated with culture medium alone (control), rTcagp63 (5 µg/mL or 10 µg/mL) for 2 hours and washed 3 times with incomplete medium before used to set up ROI assay. Cells were seeded into 96-well plates at a density of 3×10^5 cells per well. ROI assay was performed with cells incubated in medium alone or with rTcagp63 using heat-killed *A. salmonicida* and recombinant goldfish TNFa2 (rgTNFa2 100 ng/mL) to stimulate production of reactive oxygen species. ROI was determined as described in Section 3.6.

3.8 SDS-PAGE and western blot

3.8.1 SDS-PAGE

Proteins were separated and visualized by reducing SDS-PAGE according to the method described originally by Laemmli [11]. Briefly, samples were dissolved in an equal volume of Laemmli sample buffer (BioRad) and heated at 95 °C for 5 minutes. Samples were electrophoresed through polyacrylamide gels (4% stacking, 12% separating) at 100 V for 15 minutes followed by 185 V for 45 minutes. The polyacrylamide gels were stained with Coomassie brilliant blue (BioRad) to visualize protein bands in some experiments. Coomassie staining was performed by soaking the gels in a solution of Coomassie brilliant blue (0.1% Coomassie brilliant blue, 10% acetic acid, 40% methanol) for 30 minutes at room temperature). Gels were destained using 10% acetic acid and 40% methanol.

3.8.2 Western blot

Following gel electrophoresis, proteins were transferred to 0.2µm nitrocellulose membranes (BioRad) at 100V for 1 hour in blotting buffer (25 mM Tris, 192 mM glycine, 20% methanol). Membranes containing transferred proteins were immunoblotted in the following manner. The nitrocellulose were blocked with 0.5% BSA in Tris-buffered saline/Tween 20 (TTBS; 0.1% Tween 20 in 100 mM Tris-HCl, 0.9% NaCl, pH 7.5; TBS) for 30 minutes at room temperature. Membranes were then incubated in a solution containing primary antibody in blocking buffer and incubated overnight at 4°C. Membranes were washed 3 times in TTBS and 3 times in TBS for 5 minutes each before incubation with secondary antibody diluted in blocking solution for 1 hour at room temperature. Blots were washed 3 times in TTBS and 3 times in TTBS and 3 times in TBS for 5 minutes each. Protein bands were visualized using chromogenic BCIP/NBT development solution according manufacturer's instructions (BioRad).

3.8.2.1 Analysis of excretory/secretory proteins using SDS-PAGE and western blot

Excretory/secretory proteins were separated by SDS-PAGE under reducing conditions. Briefly, 20 µg of proteins were mixed with equal volume of Laemmli sample buffer (BioRad), heated at 95°C for 5 minutes and loaded onto the gel. Electrophoresis was performed using 1 mm gel containing 5% stacked and 12% separating SDS polyacrylamide at 100 V for 15minutes then 180 V for 45 minutes. Proteins were electrophoretically transferred to nitrocellulose membrane (BioRad) at 100V for 1 hour in blotting buffer (25 mM Tris, 192 mM glycine, 20% methanol) for immunoblotting. Membrane strips containing ES proteins were blocked with 0.5% BSA in Tris-buffered saline/Tween 20 (TTBS; 0.1% Tween 20 in 100 mM Tris-HCl, 0.9% NaCl, pH 7.5; TBS) for 30 minutes at room temperature. Membranes were then incubated with either serum from recovered fish 56 days post infection or serum from non-infected fish (diluted 1: 25 in blocking buffer) overnight at 4°C. Membranes were washed 3 times in TTBS and 3 times in TBS for 5 minutes each before incubation with secondary antibody (hybridoma supernatant containing anti-carp IgM antibody) diluted 1:5 in blocking solution for 1 hour at room temperature. Blots were washed 3 times in TTBS and 3 times in TBS for 5 minutes and then incubated with alkaline phosphatase-labelled goat-anti-mouse IgG diluted 1:1500 in blocking buffer. Protein bands were visualized using chromogenic BCIP/NBT development solution according manufacturer's instructions (BioRad). Protein band corresponding to 68-70 KDa antigen detected on immunoblot was excised from a Coomassie blue stained SDS gel, analyzed by mass spectrometry and identified as T. carassii hsp70. Complete list of ES proteins identified using this method is shown in Table 3.5.

3.8.2.2 SDS-PAGE and western blot analysis of rTcahsp70, rTcaCRT and rTcagp63

Proteins were separated and visualized by reducing SDS-PAGE. Briefly, samples were dissolved in an equal volume of Laemmli sample buffer (BioRad), heated at 95 °C for 5 minutes and electrophoresed through polyacrylamide gels (4% stacking, 12% separating) at 100 V for 15 minutes followed by 185 V for 45 minutes. Proteins were transferred to 0.2 μm nitrocellulose membranes (BioRad) at 100 V for 1 hour in blotting buffer (25 mM Tris, 192 mM glycine, 20% methanol). Membranes containing the transferred proteins were treated in the following manner. The nitrocellulose was first incubated in blocking solution consisting of 5% skimmed milk in PBS for 30 minutes at room temperature. The membranes were then incubated with primary antibody (mouse monoclonal antipolyhistidine antibody 1: 5000 diluted in PBS and incubated overnight at 4 °C. Membranes were washed 3 times in TTBS and 3 times in TBS for 5 minutes each, prior to incubation with secondary antibody (alkaline phosphatase-labelled goat anti-mouse IgG 1:1500 or goat anti-rabbit IgG 1:3000) diluted in PBS for 1 hour at room temperature. Following incubation with the secondary antibody, the blots were washed 3 times in TTBS and 3 times in TBS for 5 minutes each. Protein bands were visualized using the chromogenic BCIP/NBT development kit according to the manufacturer's instructions (BioRad).

Polyclonal IgG antibodies raised against the recombinant proteins (1:2000) diluted in PBS were also used as primary antibody while alkaline phosphataselabelled goat anti-rabbit IgG (1:3000) diluted in PBS were also used to visualize

99

the recombinant proteins. Only rabbit anti-rTcahsp70 and rabbit anti-rTcaCRT polyclonal IgG raised against the recombinant proteins react with the native proteins in whole cell lysate. Anti-rTcagp63 failed to react with native gp63 in *T. carassii* lysate in Western blot analysis.

Preparation of ES proteins and parasite cell lysate has been described previously [14]. For detection of native Tcahsp70 and TcaCRT in ES proteins and in whole cell lysate, ten μg of ES products or cell lysate were separated by SDS PAGE. Following gel electrophoresis, proteins were transferred to nitrocellulose membranes and immunoblotted. Membrane strips containing ES products or whole cell lysate proteins were blocked with 5% skimmed milk in PBS for 30 minutes at room temperature. Membranes were then incubated with either polyclonal rabbit anti-rTcahsp70 or rTcaCRT affinity purified IgG antibodies diluted in PBS (1:1500) overnight at 4°C. Membranes were washed 3 times in TTBS and 3 times in TBS for 5 minutes each before incubation with alkaline phosphatase-labelled goat anti-rabbit IgG antibody diluted in PBS (1:3000) for 1 hour at room temperature. Blots were washed 3 times in TTBS and 3 times in TBS for 5 minutes at room temperature. Protein bands were visualized using chromogenic BCIP/NBT development solution according manufacturer's instructions (BioRad).

3.8.2.3 SDS-PAGE and western blot analysis of carp and goldfish first complement component C1q

C1q-enriched fraction was isolated from fresh goldfish serum as

100

described previously [13]. Briefly, 2.5 mL of 30% (w/v) PEG4000 in 20 mM Tris-buffered saline (pH 7.5) was added to 10 mL of freshly collected carp or goldfish serum drop at a time with continuous mixing. The mixture was kept on ice for 15 minutes and centrifuged at 400 x *g* for 10 minutes. The supernatant was removed and pellet was re-suspended in 2 mL of TBS and protein concentration determined using Micro BCA Protein Assay Kit (Pierce). The fraction was separated on SDS-gel and transferred to nitrocellulose membrane for western blotting. C1q were detected with anti-carp C1q-A antiserum (a kind gift from Dr Miki Nakao, Kyushu University, Japan) diluted 1:1500 in PBS.

For detection of carp and goldfish C1q, the membranes were incubated with primary antibody rabbit anti-carp C1q-A antiserum 1:1500 diluted in PBS and incubated overnight at 4 °C. Membranes were washed 3 times in TTBS and 3 times in TBS for 5 minutes each, prior to incubation with secondary antibody (goat anti-rabbit IgG 1:3000) diluted in PBS for 1 hour at room temperature. Following incubation with the secondary antibody, the blots were washed 3 times in TTBS and 3 times in TBS for 5 minutes each. Protein bands were visualized using the chromogenic BCIP/NBT development kit according to the manufacturer's instructions (BioRad).

3.8.2.4 Far western blot

Far western blotting was performed using a native gel. Native gels were prepared in the absence of SDS by lowering acrylamide concentration to 8% in separating and 3% in stacking gels. Protein samples were dissolved in native sample buffer (0.125 M Tris pH 6.5, 10% glycerol, 0.005% bromophenol blue) and proteins were electrophoresed using native running buffer (25 mM Tris-HCl pH 8.8, 250 mM glycine). Following electrophoresis, proteins were were transferred to nitrocellulose membrane as described above. Nitrocellulose strips were incubated with PBS alone or with 200 µg of recombinant TcaCRT (rTcaCRT) diluted in PBS for 3 hours at room temperature. Blots were washed and then incubated with rabbit anti-rTcaCRT antibody (1:1500) for 3 hours at room temperature. Following washes with TTBS and TBS, blots were incubated with alkaline phosphatase-labelled goat anti-rabbit IgG (1:3000) for 1 h at room temperature. Protein bands were visualized with BCIP/NBT development kit.

3.9 Investigation of the effects of rTcagp63 on goldfish macrophages

3.9.1 The interaction between rTcagp63 and goldfish macrophages

Goldfish macrophage cultures obtained from three individual fish (n = 3), were cultured for 8 days. 10^7 cells were washed twice and incubated with rTcagp63 (10 µg/mL) for 0, 1, 2, 3 hours at 20°C. Cells were washed 3 times with PBS and lysed with 30 µL of RIPA buffer. 30 µL of Laemmli sample buffer was added and the mixture loaded into SDS PAGE for western blotting. Detection of rTcagp63 was done with rabbit anti-rTcagp63 antibody (1:2000) followed by HRP-conjugated goat anti-rabbit IgG (1:3000). Stripping of western blot membrane for re-probing was done with mild stripping buffer (15 g glycine, 1 g SDS, 10 mL Tween 20, pH 2.2 in 11 L of ultrapure water). PVDF membranes were incubated twice for 10 minutes at room temperature with stripping buffer followed by incubation with PBS twice for 10 minutes. Blots were then incubated in TTBS twice for 5 minutes and checked to ensure signal was gone before blocking. Rabbit anti-actin-beta (NT) zebrafish polyclonal IgG (AnaSpec Inc.) (1:1000) was used for re-probing to detect goldfish actin which served as protein loading control.

3.9.2 Assessment of the effects of rTcagp63 on phospho-tyrosine protein pattern of goldfish macrophages

Goldfish macrophage cultures obtained from three individual fish (n = 3), were cultured for 8 days. 10^7 cells were washed twice and incubated with medium alone (control) or with rTcagp63 (10 µg/mL) for 6 hours. Cells were stimulated with *A. salmonicida* at 20°C for 10, 15 and 20 minutes before lysed with 30 µL of radio-immunoprecipitation assay (RIPA) buffer [150 mM NaCl, 5 mM EDTA, 50 mM HEPES, 0.5% sodium deoxycholate, 1% Nonidet P-40 (NP-40), 10 mM 2mercaptoethanol (pH 7.5)] containing 1 mM sodium vanadate, 1 mM pnitrophenyl-phosphate, 2 mM phenylmethylsulfonyl fluoride (PMSF), 50 µg/mL of aprotinin A, 25 µg/mL of leupeptin, and 25 µg/mL of pepstatin). 30 µL of Laemmli sample buffer was added and loaded in SDS PAGE for western blotting. Phospho-tyrosine protein patterns were detected with anti-phospho-tyrosine antibody (1: 4000) and HRP-conjugated goat anti-mouse monoclonal antibody (1: 3000). Membranes were developed using ECL (Pierce) on X-ray film (Eastman Kodak Co.)

3.10 Production and purification of polyclonal rabbit anti-recombinant Tcahsp70, TcaCRT and Tcagp63 antibodies

3.10.1 Production of polyclonal rabbit anti-recombinant Tcahsp70, TcaCRT and Tcagp63 antibodies

Purified recombinant proteins were used to immunize rabbits for production of polyclonal antibodies. Rabbits were injected with 250 μ g of purified recombinant protein in 750 μ L mixed with 750 μ L of Freund's complete adjuvant (FCA). Booster injections were performed every 4 weeks for 12 weeks using the same quantity of antigen in conjunction with Freund's incomplete adjuvant (FIA). The specificity of the antibody for the immunizing antigens and native Tcahsp70, TcaCRT and Tcagp63 present in *T. carassii* whole cell lysates was determined using Western blot as described above.

3.10.2 Affinity purification of anti-recombinant Tcahsp70, TcaCRT and Tcagp63 IgG from rabbit serum

The polyclonal antibodies were purified from rabbit serum using a HiTrap Protein A HP affinity column (GE Healthcare) according to the manufacturer's protocol. Briefly, 5 mL of polyclonal rabbit anti-recombinant Tcahsp70, TcaCRT or Tcagp63 was diluted in binding buffer 20 mM sodium phosphate (pH 7) and applied to a 1 mL HiTrap Protein A HP column that was pre-equilibrated with the binding buffer. The column was washed with binding buffer. The bound proteins were eluted from the column with a decreasing pH gradient (0.1 M citric acid pH 3–6). The pH of the eluted fractions was neutralized by the addition of 1 M Tris– HCl, pH 9. Each fraction was tested for the presence and purity of IgG using SDS-PAGE and immunoblotting. Fractions containing IgG were pooled and the protein concentration was determined using a Micro BCA Protein Assay Kit (Pierce). The purified IgG was dialyzed against PBS in 3 kDa MWCO Snakeskin dialysis tubing (Pierce), filter-sterilized (0.22 μ m filter; Millipore) and stored at 4°C until used.

3.11 Functional assessment of TcaCRT and investigation of its interaction with goldfish complement component C1q

3.11.1 Assessment of interaction of goldfish complement component C1q with immobilized rTcaCRT using affinity chromatography

To study interaction between goldfish C1q and parasite CRT, 5 mg of purified rTcaCRT was coupled to CNBr-activated Sepharose (GE Healthcare) according to instructions provided by the manufacturer. Three mg of C1q containing serum fraction was loaded onto CRT-Sepharose column equilibrated with 20 mM Tris–HCl (pH 7.5) containing 10 mM CaCl₂. The mixture was incubated at room temperature for 1 hour with gentle rocking to allow proteinprotein interaction. Following incubation, the column was washed 4 times with 2 mL PBS. The bound proteins were subsequently eluted by increasing NaCl concentration (150 mM, 250 mM, 500 mM) and analyzed by SDS-gel electrophoresis. Proteins were transferred to nitrocellulose membrane and C1q was detected with anti-carp C1q-A antiserum (1:1500) using Western blot. 3.11.2 Assessment of interaction of goldfish complement component C1q and rTcaCRT using co-immunoprecipitation

To further show that TcaCRT interacts with goldfish C1q, I used coimmunoprecipitation assay. Fifty µg of anti-rTcaCRT or rTcahsp70 was coupled to agarose resin according to manufacturer's instruction (Pierce). Fifty µg of rTcaCRT was incubated with 50 µg of C1q-containing fraction eluted from CNBr-activated sepharose column with 500 mM NaCl in the presence of 20 mM Tris–HCl (pH 7.5) containing 10 mM CaCl₂ for 6 hours at 4°C. The proteins were then incubated with the prepared agarose resin for 1 hour at room temperature. The columns were washed and bound proteins were eluted and separated by SDS-PAGE. Proteins were transferred to nitrocellulose membranes and eluted proteins were detected by Western blot.

3.11.3 Assessment of binding of human complement component C1q to immobilized rTcaCRT

Varying concentrations of rTcaCRT diluted in coating buffer (15 mM of Na₂CO₃, 35 mM of NaHCO₃, pH 9.6) were coated onto the wells of a microtitre plate (Corning) 100 μ L per well for 6 hours at room temperature (22°C to 25°C). Control wells received buffer alone or buffer with 3% w/v BSA. After each step, wells were washed 3 times with 0.05% Tween in PBS. Nonspecific binding sites were blocked with 3% w/v BSA in PBS for 1 hour at room temperature. Following blocking, 500 ng of pure human C1q (Sigma-Aldrich) in 100 μ L in PBS/1% w/v BSA/0.05% Tween 20 was added to each well and the plate was

incubated at 4°C overnight. To detect bound C1q, affinity-purified rabbit antihuman C1q antiserum (DAKO) 1:1500 dilution was added in 100 μ L final volume and the plates incubated at room temperature for 2 hours. Wells were washed with PBS-Tween followed by the addition of 100 μ L affinity-purified goat anti-rabbit IgG-peroxidase conjugate (DAKO) 1:3000 dilution. Plate was kept at room temperature for 2 hours followed by washings with PBS-Tween. The bound peroxidase activity was measured by adding OPD (Sigma-Aldrich) as a substrate. The color produced was read at 492 nm in a microplate reader according to manufacturer's instruction.

3.11.4 Assessment of the effects of rTcaCRT on classical complement pathwaymediated haemolysis

To assess the effect of parasite CRT on activation of classical pathway of complement of goldfish, I used sheep red blood cell lysis assay. Briefly, washed sheep erythrocytes (SRBC) re-suspended in 0.9% NaCl were sensitized by adding an equal volume of 1:250 diluted de-complemented rabbit anti-sheep erythrocyte antiserum (Cappel) in PBS containing 1 mM CaCl₂. Cells were washed thrice with 0.9% NaCl by centrifugation at 400 x *g* for 10 minutes. A total 10^7 sensitized erythrocytes in 0.9% NaCl containing 1 mM CaCl₂ was added to wells of a microtitre plate containing varying concentrations of TcaCRT or BSA mixed with fresh normal rabbit serum (1/100 dilution) in a total assay volume of 200 µL. The plate was incubated at 37°C for 45 minutes and then kept at 4°C for 3 hours.

new wells containing 50 μ L 0.9% NaCl and absorbance was measured at 415 nm. Total haemolysis was measured by lysing 10⁷ sensitized erythrocytes with water and haemolytic activity was expressed as per cent of the total haemolysis.

3.11.5 Assessment of surface localization of TcaCRT using

immunofluorescence and confocal microscopy

To determine whether TcaCRT is expressed on parasite surface, in vitro grown parasites were washed twice in serum-free medium and re-suspended to a concentration of 1×10^6 /mL. 100 µL of parasites were incubated with 160 µg of rabbit anti-rTcaCRT IgG or rabbit anti-goldfish CSF-1 antibody (isotype control) for 24 hours at 20°C. Parasites were then spun onto poly-L-lysine coated glass slides using a Cytospin 2 at 55 x g, for 7 minutes (Shandon). The parasites were fixed to the slides in methanol at -20° C for 30 minutes and permeabilized using 1% Triton X-100 for 30 minutes at room temperature. Following permeabilization, the slides were washed 3 times for 20 minutes in Milli-Q water. The slides were then blocked with 0.5% BSA in PBS for 30 minutes at room temperature. The slides were then washed 3 times for 20 minutes each in Milli-Q water prior to incubation with FITC-conjugated goat anti-rabbit IgG 1:100 dilution (Sigma-Aldrich) for 2 hours at room temperature. To assess the level of TcaCRT expression on parasite surface, cells were treated the same way except permeabilization step was omitted. The cells were subsequently washed 3 times for 20 minutes each in Milli-Q water and observed using DIC imaging with a Leica confocal microscope. Parasite staining was observed using differential

interference contrast (DIC) imaging with a Leica confocal microspcope.

3.12 Proteomic analysis of *T. carassii* surface proteins

3.12.1 Two-dimensional gel electrophoresis (2-DE) of *T. carassii* surface protein fraction.

Triton X-114 was used to solubilize parasite membrane proteins in the presence of protease inhibitors and the solubilized proteins were analyzed by 2-DE. Protein concentration was determined by Micro BCA protein assay (Pierce) and samples were precipitated with acetone. Five hundred μg of proteins from *in vitro* cultured parasites were rehydrated overnight on 24 cm immobiline dry strips at pH 4-10 and separated by two-dimensional gel electrophoresis. In first dimension, samples were run on 24 cm, pH 4-10, Immobiline Drystrips (Amersham) on an IPGphor II IEF system (Amersham). Equilibration of strips were done in equilibration buffer (75 mM Tris-HCl pH 8.8, 6 M urea, 29.3% glycerol, 2% SDS, 0.002% bromophenol blue) containing 100 mg/mL DTT for 15minutes and then in equilibration buffer containing 250 mg/mL iodoacetamide for 15 minutes. Strips were sealed to 12% acrylamide gels using 0.5% agarose in standard Tris-glycine electrophoresis buffer. Second dimension was run using Ettan DALT electrophoresis unit (Amersham) at 40 mA/gel and 16°C until tracking dye reached the end of the gel. Proteins were visualised by Sypro® Ruby fluorescence (Invitrogen). Gels were fixed overnight in 50% methanol, 7% acetic acid and stained overnight. Gels were destained in 10% methanol, 7% acetic acid for 30 minutes and stored in 7.5% acetic acid prior to imaging. Imaging was done with FLA-5000 laser scanner (Fujifilm) using 473 nm excitation and 450 nm

emission filters. Protein spots were detected using ImageMaster 2D platinum software. Spots were detected based on the following parameters: minimum area of 6, smoothness of 4 and saliency of 2500. Gel plugs containing the proteins were excised using Ettan spot picker (Amersham) and identified by mass spectrometry.

3.12.2 Mass spectrometry analysis

Excreted/secreted antigens and membrane proteins were separated by SDS-PAGE and 2-DE, respectively. Protein band excised from SDS gel were analyzed at the Institute for Biomolecular Design, University of Alberta. Protein spots from 2-D gels were analyzed at the Department of Chemistry, University of Alberta. The visualized bands or spots excised from gels were destained, reduced with DTT (Roche), alkylated with iodoacetamide (Sigma) and digested with trypsin (Promega Sequencing Grade Modified) overnight at 37°C. The extracted peptides were analyzed via LC-MS/MS on a nanoAcquity UPLC (Waters, MA) coupled with Q-ToF-Premier mass spectrometer (Micromass, UK/Waters, MA). Trypsin digested peptide fragments were separated using a linear water/acetonitrile gradient (0.1% Formic acid) on a nanoAcquity column (3 micron Atlantis dC18, 100 A pore size, 75 µm ID x 15 cm) (Waters, MA) as a loading /desalting column. Protein identification using the generated MS/MS spectra was performed by searching the NCBI non-redundant database using MASCOT or PEAKS studio 4.5 with consideration for carbamidomethylation of cysteine and the oxidation of methionine.

Analysis of LC-MS/MS was done with Mascot (Matrix science Ltd, London, UK) or PEAKS studio 4.5 software by comparing with the data in the National Centre for Biotechnology (NCBI) non-redundant database. Parameters used for analysis of MS results for both ES and membrane proteins included probabilistic score-based confidence interval, appearance of kinetoplastid proteins as top candidate, number of peptides matched and extent of sequence coverage. List of proteins identified in this study are shown in Table 3.5 for *T. carassii* ES fraction and Table 3.6 for surface proteins

3.13 Identification of *T. carassii* genes up-regulated *in vivo* using suppression subtraction hybridisation (SSH).

To identify differentially expressed transcripts between mRNAs isolated from *in vivo* and *in vitro* grown parasites, suppression subtraction hybridisation strategy was employed. This allows comparison of two mRNA populations to identify transcripts that are differentially expressed between two RNA samples. SSH was performed using cDNAs synthesised from *in vivo* cultured parasite as tester and *in vitro* cultured parasite as driver. For *in vivo* RNA, goldfish were infected with approximately 6.25×10^6 parasites and allowed to live for 21 days to achieve peak parasitemia. Fish were checked for parasitemia a week after infection. Fish were exsanguinated and parasites were immediately purified from blood. In all cases, excessive washing of parasites was avoided to prevent possibility of mRNA degradation. Total RNA was extracted using TRIzol reagent (Invitrogen) and RNA samples stored at -80°C. Total RNA was also isolated from

in vitro cultured parasites grown for 5 days to exclude possibility of bloodstream parasite differentiating into stumpy form. mRNAs were isolated from total RNAs using µMACS mRNA Isolation Kit (Miltenyi Biotec). cDNA synthesis and SSH were performed with 2 µg of mRNAs using BD PCR-Select cDNA Subtraction kit (clonetech). Differentially expressed cDNAs were directly cloned into TOPO TA vector and used to transform Top10 Competent cells. Transformants were selected on Luria bertani (LB) medium containing ampicillin (100 μ g/mL). Xgal and IPTG were added to LB plates to allow selection of white colonies, which are transformants containing inserts. About 180 colonies were selected and cultured in 200 µL of LB broth containing ampicillin in a 96 well culture plates. Cultures were used directly as templates in PCR reactions to further screen for transformants containing inserts. Clones containing inserts from gel electrophoresis of PCR products were used to inoculate 2 mL LB broth containing ampicillin. Plasmids were purified (Qiagen) and sequenced directly using vectorspecific primers. Sequences were compared to databases at GeneDB or NCBI to identify the gene fragments. List of genes identified in the SSH study is shown in Table 3.7.

3.13.1 *In silico* analysis of differentially expressed gene fragments identified in suppression subtraction hybridisation

Gene fragments obtained by sequencing were used to identify the respective genes. A 50% identity cut off was used for gene identification. In some cases, protein sequences of gene fragments were used in database searches. Where sequence fragments show no identity to databases sequences, domains and signatures present in the fragments were identified and used for gene identification. In most cases, gene fragments show identity to more than one parasite genes suggesting that the differentially expressed transcript could be any of those genes. Of the 120 sequenced clones, about 20 clones show no significant homology to any database sequence suggesting they could be T. carassii specific genes. About 40 differentially expressed transcripts show identity to uncharacterised parasites genes and are excluded from the list. Some identified genes were found in more than one clone. Since the cDNAs were digested with Rsa1 during SSH procedure, genes such as surface antigen, mucins, histones, and hsp40 which were identified in many clones could be different fragments of the same T. carassii gene. This indicates the protein product of these genes may be important for parasite immune escape strategies. The result of *in silico* analysis is hampered by low sequence homology between the identified sequences and the database sequences. This made gene identification difficult.

3.14 Statistical analysis

Statistical analysis was performed using one-way analysis of variance (ANOVA) and Tukey's test. Probability level of P < 0.05 was considered significant. In all figures, error bars represent standard errors calculated from standard deviation and sample size. For gene expression analysis, four individual fish (n = 4) were used because less variability in gene expression was observed in previous studies conducted in our laboratory. For functional assays, between four

to seven individual fish were used since these assays are subject to much higher levels of variability as shown by previous studies in our laboratory.

Components of Trypanosoma carassii TDL-15 Medium^a

	1	
Reagents/Solutions	Amount/L	Specifications
Milli-Q water	300 mL	Endotoxin-free
Hank's Solution	40 mL	10X, no Ca^{2+} or Mg^{2+}
MEM amino acid solution ^b	12.5 mL	50X
MEM non-essential amino	12.5 mL	100X
acid solution ^b		
NaHCO ₃	1.26 g	
NaOH	30 mL	1 N
Sodium pyruvate solution ^b	12.5 mL	10 mM
MEM vitamin solution ^b	10 mL	100X
Nucleic Acid Precursor	10 mL	2.5 mM
Solution ^c		
L-glutamine Solution	10 mL	200 mM
Gentamicin solution ^b	1 mL	50 mg/mL
2- Mercaptoethanol solution	1 mL	50 mM (fresh)
HEPES	2 g	
Insulin	0.005 g	
GFL-15 Medium ^d	500 mL	

^a Developed by Wang and Belosevic [16]

^b Supplied by Gibco/Invitrogen

^c 2.5 mM each of : adenosine, cytidine, hypoxanthine, thymidine, uridine.

^d 50% Leibovitz-15 and 50% Dulbecco's Modified Eagles Medium

Reagents/Solutions	Amount/2L
Hepes	7 g
KH ₂ PO ₄	0.688 g
K ₂ HPO ₄	0.570 g
NaOH	0.750 g
NaHCO ₃	0.340 g
L-glutamine	0.5844 g
Insulin	0.01 g
10X Hank's balanced salt solution	80 mL
MEM amino acid solution ^a	25 mL
MEM non-essential amino acid solution ^a	25 mL
Sodium pyruvate solution ^a	25 mL
MEM vitamin solution ^a	20 mL
Nucleic Acid Precursor Solution ^b	20 mL
2- Mercaptoethanol solution	7 μL
GFL-15 Medium ^c	1 L

Components of MGFL-15 Medium

^a Supplied by Gibco/Invitrogen

^b 0.067 g adenosine, 0.061 g cytidine, 0.031 g hypoxanthine, 0.061 g thymidine,

0.061 g uridine in 100 mL.

^c GFL-15 medium: 1 package of dry powder each of Leibovitz-15 and Dulbecco's Modified Eagles Medium in 2 L.

Primes used for molecular cloning of Trypanosoma carassii genes

Primers	Sequences (3'- 5')
Primers for amplification of Tcahsp70 fragments:	
Hsp70 forward	GATACAGTTTCTGTACTATATTG (Spliced leader)
Hsp70 reverse	TGCCGCCCACCAGCACCACATCAT (MS peptide)
Hsp70 3' RACE	GCTGACGATCGATGGGCGGCATCTTCGAG

Primers for cloning Tcahsp70 for expression in prokaryotic system

SUMOHsp70 forward	ATGGCCTACGAAGGTGCGATTG
SUMOHsp70 reverse	TCAGTCAACTTCCTCCACCTTC
Vector-specific primers:	
M13 forward	GTA AAA CGA CGG CCA G
M13 reverse	ACA GCT ATG ACC ATG ATT AC
SUMO forward	AGA TTC TTG TAC GAC GGT ATT AG
T7 reverse	TAG TTA TTG CTC AGC GGT GG

Primers for amplification of TcaCRT gene fragments:

CRT forward	(same as hsp70 forward)
CRT Reverse	TCGGGTTCGGAATCATCGGCGCTTC (MS peptides)
Primers for 3' RACE	GACTGGCCGATTGTTCAGCCCAAGG
Primers for cloning Tca	CRT for expression in prokaryotic system:
SUMOCRTforward	ATGCGTGCATCACTTCTCCTTATTG
SUMOCRTreverse	TTAGAGATCACCCTCCTCTTCCGCA

Primers for amplification of Tcagp63 fragments:

Gp63 forward	(same as hsp70 forward)
3' RACE primer	CACCGATGACGAGGGAAAGGGCCCAACAGC

Quantitative real time primer sequences

Primer	Sequence (5'-3')
CCL-1 forward	AAG GTC ACC GAA CCC ATC AG
CCL-1 reverse	TCG TCA CAT GAT GGC CTT CA
CXCL-8 forward	CTG AGA GTC GAC GCA TTG GAA
CXCL-8 reverse	TGG TGT CTT TAC AGT GTG AGT TTG G
EF-1α forward	CCG TTG AGA TGC ACC ATG AGT
EF-1α reverse	TTG ACA GAC ACG TTC TTC ACG TT
gp91 phox forward	CCC ATC ACC TGT TCA TCG TCT T
gp91phox reverse	TAG TCT GGC CTC GCA CGA TA
IFN _y forward	GAA ACC CTA TGG GCG ATC AA
IFNy reverse	GTA GAC ACG CTT CAG CTC AAA CA
IL-1β-1 forward	GCG CTG CTC AAC TTC ATC TTG
IL-1β-1 reverse	GTG ACA CAT TAA GCG GCT TCA C
IL-1β-2 forward	GAT GCG CTG CTC AGC TTC T
IL-1β-2 reverse	AGT GGG TGC TAC ATT AAC CAT ACG
IL-10 forward	CAAGGAGCTCCGTTCTGCAT
IL-10 reverse	TCGAGTAATGGTGCCAAGTCATCA
IL-12-p35 forward	TGT TTT ACG TGC ATT CCT TTG G
IL-12-p35 reverse	GGC GCC TGA AAA AAA TAC GA
IL-12-p40 forward	CTT CAG AAG CAG CTT TGT TGT TG
IL-12-p40 reverse	CAG TTT TTG AGA GCT CAC CGA TAT C
iNOS isoform A forward	TTG GTA CAT GGG CAC TGA GAT T
iNOS isoform A reverse	CCA ACC CGC TCA AGA ACA TT
iNOS isoform B forward	CAT CTT CCA TCC GAC CCT AGT G
iNOS isoform B reverse	AAA GCT ACG GAA GGG AGC AAT
p22 phox forward	TGG ACC CCT GAC CAG AAA CT
p22 phox reverse	AAC ATG AAC CCC CCT GGA A

p40 phox forward	TCC AAG AGC GGG AAT CAT G
p40 phox reverse	GTC GAT GCC CTC TGG CTG TA
p47 phox forward	CCA GGA ATG GGA CAC GAT CT
p47 phox reverse	GAG GAG AGC CTG AGT TTG CAA
p67 phox forward	TGC CTG GCA ACA TTG TCT TC
p67 phox reverse	CCC GCT TCT CAT TGA AAA CAA
TGFβ forward	GTA CAC TAC GGC GGA GGA TTG
TGFβ reverse	CGC TTC GAT TCG CTT TCT CT
TNFα-1 forward	CAT TCC TAC GGA TGG CAT TTA CTT
TNFα-1 reverse	CCT CAG GAA TGT CAG TCT TGC AT
TNF α -2 forward	TCA TTC CTT ACG ACG GCA TTT
TNFα-2 reverse	CAG TCA CGT CAG CCT TGC AG

Molecular Weight (KDa)	Protein Identity
36	Activated Kinase C receptor
52	Elongation factor 1α
55	Alpha and beta tubulin
55	Caleticulin
72	Hsp70
81	Hsp 83
95	Elongation factor
100	Pyruvate dikinase

Trypanosoma carassii antigens identified in ES products by MS.
Table 3.6

Spot Number	Molecular Weight (KDa)	Protein Identity
1.	55	Tubulin alpha
2.	55	Tubulin alpha
3.	55	Tubulin beta
4.	58	Hsp60
5.	60	Hsp60
6.	78	Glucose regulated protein 78
7.	70	Hsp 70
8.	70	Hsp 70
9.	50	Calreticulin
10.	45	Hsp70
11.	45	Hsp 70
12.	40	Elongation factor 2
13.	36	Tubulin beta chain
14.	50	Enolase
15.	20	Tryparedoxin peroxidase
16.	54	Enolase
17.	17	Cyclophilin A
18.	40	MDR Efflux
19.	50	Enolase
20.	50	Actin
21.	40	Elongation factor 2
22.	40	Elongation factor 2
23.	36	Fructose-bisphosphate aldolase
24.	36	Enolase

T	••		••••	• • • • • •	•		P 1.	1 1/0
Trunanacoma	AMPACCII Y	rotoing	idont	itiad in	curtaca	nrotoin	traction	
I I VIJUILIJN(JIILU	L <i>UTU</i> NN <i>L</i> L I	JI ULCIIIS.	IUCIII		SULLAUE	UIULCIII	II AULIUII	
						P- 0000000		

Table 3.7

Differentially expressed Trypanosoma carassii genes identified by SSH.

Sequences	Possible Genes
1.	Surface antigen, Cysteine peptidase
2.	Histone H1, Cysteine peptidase, Integral membrane protein
3.	Surface antigen protein 2, Transialidase
4.	DnaJ protein i.e HSP40
5.	Mucin associated surface protein
6.	Cytochrome C oxidase Cox1 homologue
7.	Mucin associated surface protein, Interspersed repeat antigen
8.	Cysteine protease, Serine/threonine protein kinase
9.	Peptidase
10.	Mucin associated surface protein, Interspersed repeat antigen,
11.	Histone
12.	Surface antigen, Mucin TcSMUGS,
13.	Surface antigen, Mucin TcMUC1
14.	Ribosomal protein L15

_

3.15 References

- Asea, A., S. K. Kraeft, E. A. Kurt-Jones, M. A. Stevenson, L. B. Chen, R. W. Finberg, G. C. Koo, S. K. Calderwood. 2000. HSP70 stimulates cytokine production through a CD14-dependant pathway, demonstrating its dual role as a chaperone and cytokine. Nature Medicine 6: 435-442.
- Bienek, D. R., D. A. Plouffe, G. F. Wiegertjes, M. Belosevic. 2002. Immunization of goldfish with excretory/secretory molecules of *Trypanosoma danilewskyi* confers protection against infection. Developmental Comparative Immunology 26: 649–657
- Bienek, D. R. 2001. The biological and immunological aspect of the hostparasite relationship of goldfish, *Carassius auratus*, infected with *Trypanosoma danilewskyi*. PhD thesis, Department of Biological Sciences, University of Alberta, Edmonton.
- Bienek, D. R., and M. Belosevic. 1997. Comparative assessment of growth of *Trypanosoma danilewskyi* (Laveran & Mesnil) in medium containing fish or mammalian serum. Journal Fish Disease 20: 217–221.
- Bordier, C. 1981. Phase separation of integral membrane proteins in Triton X-114 solution. Journal of Biological Chemistry 256: 1604-1607
- Ferreira, V., C. Valck, G. Sanchez, A. Gingras, S. Tzima, M. C. Molina, R. Sim, W. Schwaeble, and A. Ferreira. 2004. The classical activation pathway of the human complement system is specifically inhibited by calreticulin from *Trypanosoma cruzi*. Journal of Immunology **172**: 3042-3050.
- Grayfer, L. 2011. The of cytokine regulation of macrophage antimicrobial responses of the goldfish (*Carassius auratus* L). PhD thesis, Department of Biological Sciences, University of Alberta, Edmonton.
- Grayfer, L., and M. Belosevic. 2009. Molecular characterization, expression and functional analysis of goldfish (*Carassius aurutus* L.) interferon gamma. Developmental Comparative Immunology 33: 235-246.
- Hey, A. S., T. G. Theander, L. Hviid, S. M. Hazrati, M. Kemp, and A. Kharazmi. 1994. The major surface glycoprotein (gp63) from *Leishmania*

major and *Leishmania donovani* cleaves CD4 molecules on Human T cells. Journal of Immunology **152:** 4542-4558

- Katzenback, B. A., D. A. Plouffe, G. Haddad, M. Belosevic. 2008. Administration of recombinant parasite β-tubulin to goldfish (*Carassius auratus* L) confers partial protection against challenge with *Trypanosoma danilewskyi* Laveran and Mesnil,1904. Veterinary Parasitology 151: 36–45.
- 11. Laemmli, U. K. 1970. Cleavage of structural proteins during the assembly of the head of bacteriophage T4. Nature **227:** 680-685.
- Neumann, N. F., D. R. Barreda, and M. Belosevic. 2000. Generation and functional analysis of distinct macrophage sub-populations from goldfish (*Carassius auratus* L.) kidney leukocyte cultures. Fish and Shellfish Immnuology **10:** 1-20.
- Oladiran A, Belosevic M. *Trypanosoma carassii* calreticulin binds host complement component C1q and inhibits classical complement pathwaymediated lysis. Developmental and Comparative Immunology 2010; 34: 396-405.
- Oladiran, A., and M. Belosevic. 2009. *Trypanosoma carassii* hsp70 increases expression of inflammatory cytokines and chemokines in macrophages of the goldfish (*Carassius auratus* L.). Developmental and Comparative Immunology 33: 1128–1136.
- Wang, R., and M. Belosevic. 1995. The in vitro effects of estradiol and cortisol on the function of a long-term goldfish macrophage cell line. Developmental and comparative Immonology 19: 327-336.
- Wang, R., and M. Belosevic. 1994. Cultivation of *Trypanosoma* danilewskyi (Laveran & Mesnil 1904) in serum-free medium and assessment of course of infection in goldfish, *Carassius auratus (L.)*. Journal of Fish Disease **17:** 47–56.
- 17. Wilson, M. R., A. Wang, W. W. Fish, and G. W. Warr. 1985. Anomalous behaviour of goldfish IgM heavy chain in sodium dodecyl sulphate

polyacrilamide gel electrophoresis. Comparative Biochemical Physiology **82B:** 41-49.

 Woo, P. T. K. 1979. The hematocrit centrifuge for the detection of trypanosomes in blood. Canadian Journal of Zoology. 47: 921-923.

CHAPTER 4

THE COURSE OF INFECTION AND EXPRESSION ANALYSIS OF INFLAMMATORY AND ANTIMICROBIAL GENES IN THE GOLDFISH (Carassius auratus L.) INFECTED WITH Trypanosoma carassii¹.

4.0 Introduction

T. carasssii can naturally infect a number of fresh water fish including common carp (*Cyprinus carpio*), tench (*Tinca tinca*) and eel (*Anguilla* species) [23]. Experimentally, the parasite was shown to be infective to other fish species such as tin foil barb (*Barbus schwanenfeldi*), brown bullhead (*Ictalurus nebulosus*), and goldfish (*Carassius auratus*) [22, 60]. *T. carassii* is transmitted by blood sucking leeches of the genera *Hemiclepis* and *Pisciola* [47, 60]. In experimental studies, parasites isolated from fish can be used to establish infection in a new host. Also, parasite cultured *in vitro* can be used to establish new infection [9, 57, 60].

T. carassii is an extracellular protozoan parasite and morbidity and mortality caused by this parasite can be significant especially in aquaculture setting, where prevalence of infection can approach 100% [21, 40]. Goldfish are highly susceptible to trypanosome infection and while 60–80% mortality has been

¹A version of this chapter has been published: Oladiran, A., D. Beauparlant, and M. Belosevic. 2011 The expression analysis of inflammatory and antimicrobial genes in the goldfish (*Carassius auratus* L.) infected with *Trypanosoma carassii*. Fish and Shellfish Immunology **31:** 606-613.

documented and fish that survive the infection become immune to homologous challenge [59, 60]. The nature of the protective immune responses induced by *T*. *carassii*, as well as parasite molecules that drive such responses, are yet to be fully elucidated. In addition, the host factors that contribute to induction of protective immunity and those that govern outcome of infection remain to be fully elucidated.

The role of cytokines in the control of trypanosome infections has been an area of intense research. For example, *T. brucei brucei* produces a trypanosome lymphocyte triggering factor (TLTF), which triggers CD8⁺ cells to produce IFN- γ and TGF- β [5, 6, 39]. The trypanocidal effect of TNF- α has been reported [24], while IL-4 has been shown to promote antibody production important during the elimination phase of the infection [7]. IFN- γ produced by NK cells during *T. cruzi* infection activates phagocytic cells to destroy internalised parasites through production of reactive nitrogen intermediates [3, 11, 15]. Recently, it was reported that *T. carassii* infection of carp caused an induction of IFN- γ 2 and IL-23, suggesting a role for a Th17-type immune response [49].

Parasites modulate host cytokine network to enhance their survival in the host. It is well documented that parasites or parasite-derived molecules are capable of inducing inflammatory response [20, 27, 38, 42, 49, 54]. Several studies using parasite model systems have shown an association between disease progression and inflammatory responses. For example, a decreased Th2 cytokine production and parallel increase in the Th1 cytokine levels were shown to correlate with resistance to *Leishmania* spp. infections [8, 34, 48, 53]. In

Schistosoma masoni infected mice, a down-regulation of Th1 cytokine production is accompanied by induction of Th2 response [43]. Mice infected with *T. cruzi* have decreased ability to produce interleukin-2 (IL-2) and exhibit reduction in the expression of IL-2 receptor [36, 50]. In addition, *T. cruzi* infected mice were shown to produce higher amounts of the pro-inflammatory cytokine IFN- γ during the second week of infection [36, 55]. IFN- γ is one of the central cytokines that promotes antimicrobial effector responses of macrophages resulting in the inhibition of parasite replication [15]. Similar results have been reported for IL-10 knockout mice infected with *T. cruzi*. These mice exhibit lower parasitemia in blood and tissues and higher levels of IFN- γ and nitric oxide production by spleen cells compared to wild type mice [1].

During the early stages of infection of mice with African trypanosomes, higher levels of pro-inflammatory cytokines IFN- γ and TNF- α have been reported [12, 56]. Similarly, the infection of vervet monkeys with *T. brucei rhodesiense* induced higher levels of IFN- γ , TNF- α and soluble TNF receptor 1 in serum but not in the cerebrospinal fluid of the animals [28]. It has also been postulated that in Aftican trypanosomiasis, the survival of the host was dependent on a shift in the production of pro-inflammatory cytokines (IFN γ and TNF α), to the production of anti-inflammatory cytokines (TGF β and IL-10) [37].

The expression profiles of cytokines induced in the host during the acute and elimination phases of infection of goldfish with *T. carassii* and their roles in host defense have not been fully elucidated. My working hypothesis was that *T. carassii* induce Th1 type immune response to evade Th2 type response, which is required for their elimination, or mixed Th1/Th2 responses to suppress specific or general immune response. In this study, I examined the expression patterns of a panel of pro-inflammatory genes encoding IFN- γ , TNF α 1 and TNF α 2, IL-1 β -1 and IL-1 β -2, IL-12-p35 and IL-12-p40, CCL1, CXCL8, anti-inflammatory genes encoding IL-10 and TGF β and antimicrobial genes encoding iNOS A and iNOS B and those encoding the NADPH oxidase components.

4.1 Experimental Design

4.1.1 Assessment of T. carassii growth in vitro

Trypanosomes were cultured in 25 cm² tissue culture flasks. Briefly, 1×10^5 trypanosomes/mL of medium were added to 25 cm² tissue culture flasks, $20^{\circ}C \pm 2^{\circ}C$. The number of parasites in each culture (n = 4) was determined every two days for 8 days by performing hemocytometer counts on diluted samples. To determine the generation time (g) of the parasites, the following formula was used:

$$g = t_2 - t_1 / \log e^{n^2/n^2}$$

Where t_1 = time of inoculation; t_2 = end time point; n1 = number of parasites in the inoculum; n2 = number of parasites at the end point [14].

4.1.2 Course of infection in goldfish infected with *T. carassii*

Before experimental infections of fish with *T. carassii*, blood sample was taken from each fish and examined for the possible presence of natural hemoflagellate infections. The goldfish used in all studies in this thesis, were not

naturally infected with hemoflagellates. The cultured trypanosomes $(6.25 \times 10^6/\text{fish})$ were used to inoculate individual fish intraperitoneally (n = 10). We have previously shown that all fish inoculated with this dose of parasites became infected. Non-infected control fish received an equal volume of serum-free culture medium. On days 3, 7, 14, 21, 28, and 56 pi, blood samples were collected and number of parasites/mL of blood was determined using a hemocytometer or hematocrit centrifugation technique, when parasites numbers were below detection limit of the hemocytometer [58]. In our system, the detection limit of hemocytometer is 10^3 parasites.

4.1.3 Assessment of the effects of *T. carassii* infection on expression of pro-inflammatory and antimicrobial genes of goldfish.

Spleen, liver and kidney tissues were removed from infected (n = 4) and non-infected (n = 4) fish on days 7, 14, 21, 28 and 56 pi, and were immediately flash frozen in liquid nitrogen and stored at -80°C until RNA isolation. Total RNA was extracted from goldfish tissues and RNA from the different tissues was reverse transcribed into cDNA. The purity of RNA was confirmed by calculating the ratio of the absorbance at 260 nm and 280 nm. Also, equal amounts of RNA and cDNA were used. Relative expression was determined in relation to elongation factor-1 alpha (EF-1 α). All expression data were normalized using the baseline expression in non-infected control fish for each gene. Results were reported as fold change in expression relative to non-infected control for each gene.

4.2 Results

4.2.1 The course of *T. carassii* infection in the goldfish

Image of *in vitro* grown parasite is shown in Fig 4.1. All fish infected with *in vitro* cultured *T. carassii* and their blood was positive for parasites on day 3 pi. In the early stages of infection, I observed a continuous rise in parasitemia from day 3 (pi) with peak parasitemia between days 14 and 21 pi (Table. 4. 1). After day 21 pi, the mean number of parasites gradually declined (days 28 to 56 pi) (Table. 4.1) and all fish remained infected at day 56 pi. The first phase of infection (days 3-21) is subsequently referred to as the acute phase while the second phase (days 28-56) is referred to as elimination phase. No mortality was observed in both non-infected control fish and infected fish.

Assessment of parasite growth *in vitro* showed that parasite proliferated in TDL-15 medium supplemented with 10% heat-inactivated goldfish serum (Fig. 4.2). The generation time of parasites was found to be 47.1 hours based on the above formula and the average number of parasites from the cultures on day 8. This represents a greater than 8-fold increase in parasite concentration over the course of growth (Fig. 4.2).

4.2.2 *T. carassii* induces differential cytokine gene expression during the acute and elimination phases of the infection in the goldfish

To understand the immune response elicited during *T. carassii* infection of goldfish, I measured the expression of genes encoding pro-inflammatory cytokines CCL1, CXCL8, IFN- γ , TNF α 1 and TNF α 2, IL-1 β -1 and IL-1 β -2, IL-

12-p35 and IL-12-p40, and anti-inflammatory cytokines TGFβ, and IL-10 using quantitative PCR. I compared immune gene expression in the kidney, spleen and liver in infected and non-infected goldfish. All of the fish that were experimentally infected with *T. carassii* had parasites in their blood. In the kidney, I observed significant up-regulation of IL-1β-2, IFN- γ and IL-10 by day 7 pi (Fig. 4.3). By day 14 pi, IL-1β-2 mRNA level returned to normal while IFN- γ and IL-10 mRNA levels were significantly higher in infected compared to noninfected fish (Fig. 4.3). With the exception of IL-1β-2 and CXCL8 (day 28 pi), all cytokines were significantly up-regulated by day 21 and 28 pi (Fig. 4.3). All cytokines were either down-regulated or returned to normal levels by day 56 pi (Fig. 4.3).

In the spleen, I observed up-regulation of IFN- γ , IL-1 β -1 and IL-10 during the acute phase of infection, days 7 and 14 pi (Fig. 4.4). There were also significant increases in the mRNA levels of CCL1, IL-12-p35, IL-12-p40 and TNF α 2 on day 14 pi. By day 21 pi, cytokine mRNAs returned to normal levels with the exception of IL-10, which was still significantly up-regulated (Fig. 4.4). A significant down-regulation of mRNA levels of majority of cytokine genes was observed on day 28 pi, and by day 56 pi the mRNA levels of all cytokines examined, except for IL-10, were similar in infected and non-infected fish (Fig. 4.4).

In the liver, I observed a significant up-regulation of IL-1 β -1 by day 7 pi, IL-1 β -1 mRNA level returned to the level observed in non-infected fish by day 14 pi (Fig. 4.5). In addition, the mRNA levels of CCL1, IFN- γ and IL-10 were

significantly higher in infected compared to non-infected fish on day 14 pi. A significant up-regulation of all cytokines was observed (with the exception of IL-1 β -1) on day 21 pi, whereas only CXCL8 and IL-10 mRNA levels were significantly higher in infected fish on day 28 pi. The cytokines mRNA levels returned to levels in non-infected fish by day 56 pi with the exception of IL-10 (Fig. 4.5).

4.2.3 Infection of goldfish with *T. carassii* induces delayed expression of iNOS genes

Nitric oxide production regulated by iNOS genes is up-regulated during trypanosome infection in mammals [16]. For this reason, I examined the expression of iNOS genes in *T. carassii* infected and non-infected fish. I observed a delayed (day 14 or 21 pi) but a significant up-regulation of iNOS genes (both isoforms A and B) in the kidney and liver of infected fish (Fig. 4.6). I also observed a lesser but significantly different increase in expression of iNOS genes in the spleen on days 14 and 21 pi, compared to non-infected animals (Fig. 4.6). These results indicate that *T. carassii* infection induced enhanced iNOS gene expression that was delayed during the critical period of establishment of parasites in the host (first week pi).

Finally, I examined the effects of *T. carassii* infection of goldfish on expression of NADPH oxidase gene components. Both phox40 and phox67 showed non-significant increases in expression in the kidney of infected goldfish on day 21 pi (Fig. 4.7)

4.3 Discussion

The course of infection reported in this thesis is consistent with previously observed course of infection in *T. carassii*-infected goldfish [10, 46]. Also, the generation time determined is consistent with previously reported observation for parasite grown under the same condition [14, 46]. The number of parasite in the inoculum has been suggested to be the major determinant of mortality in *T. carassii*-infected fish [57, 59]. Although a high dose of parasites was used in this study, no mortality was observed. The *in vitro* cultured parasites were highly infective since all inoculated fish developed trypanosome infections as early as day 3 pi.

The inability of *in vitro* grown parasites to cause mortality may be due to many factors among which is the source of the infecting parasites since parasites can lose virulence after prolonged *in vitro* cultivation. However, *T. carassii* grown in culture supplemented with 5% carp serum are indistinguishable from parasites isolated from infected fish with respect to morphology, surface structure and reactivity with monoclonal antibody [40]. It is unlikely that *in vitro* cultured parasites have decreased infectivity since the course of infection reported in this thesis is consistent with previous observations [10, 46]. Another reason for decreased virulence of *T. carassii* may be due to the adaptation of the developmental form of the parasite used to infect fish. For example, African trypanosomes have different developmental forms adapted to live in different hosts [29, 30]. Different developmental forms of *T. carassii* have also been observed [47], but it is not clear how these different forms are adapted to live in fish and leech hosts. There is evidence that *in vitro* grown bloodstream forms of *T. carassii* decrease from 80% to 60%, after one week of cultivation as shown by morphometric studies [10]. These observations suggest that a lower number of virulent parasites may be present in the inoculums since I used parasites obtained from 5-7 day-old cultures to infect fish.

The ability to grow *T. carassii in vitro* and the ability of these parasites to cause an infection in the goldfish provides an excellent experimental model system to study the immune evasion mechanism in this host-parasite association.

Cytokines produced by exposure to parasites or parasite-derived molecules determine whether a Th1, Th2 or Th17 type response is elicited. The levels of IL-12 or IL-4 early during infection determine whether Th1 or Th2 type immune response predominates, while Th17 type response is regulated by the production of IL-23 [14, 33, 35, 44]. The elaboration of Th1 and Th2 T-cell subsets, eventually results in efficient host defence against intracellular and extracellular parasites, respectively. Since *T. carassii* is an extracellular parasite, the induction of IFN- γ (a Th1 cytokine) may cause reduction in Th2-type cytokine production and affect host's humoral immune responses, an immune evasion mechanism similar to that shown for *S. mansoni* [43]. It is well established that anti-parasite antibody responses are essential for *T. carassii* clearance and maintenance of long lasting resistance to re-infection [17, 18, 41, 59].

From the literature, it is clear that exclusively extracellular trypanosomes, such as *T. brucei*, have adapted to exclusively live in the blood of their hosts. The key to that adaptation is that these parasites express on their surface Variable Surface Glycoproteins (VSGs), and that they are able to evade host immune responses in the blood (primarily anti-parasite antibody responses of the host) by expressing different VSGs on their surface (there are many as 1200 genes that encode VSGs). In contrast, intracellular trypanosomes, such as *T. cruzi*, do not express VSGs on their surface, instead their surface coat is very rich in sialic acid residues. From an evolutionary perspective, although *T. carassii* is considered to be an extracellular parasite at present, it does not express VSGs on its surface (like *T. brucei*), but it does have a surface coat that is very rich in sialic acid residues like *T.cruzi* [2, 41].

I observed that *T. carassii* induced different cytokine profiles in the three tissues of the goldfish during the trypanosome infection. Interestingly, *T. carassii* induced higher expression of both pro-inflammatory and anti-inflammatory cytokines during the acute phase of the infection, while the mRNA levels of these cytokine genes either returned to normal levels or were down-regulated during the elimination phase of infection, with notable exception of the IL-10. In addition, the expression of majority of the cytokine genes appeared to be sustained longer in the kidney of the infected fish (until day 28 pi), compared to the cytokine mRNA levels in the spleen (day 21 pi). An up-regulation of IL-1 β observed during the acute phase of infection was expected since IL-1 is an important mediator of inflammatory response and tissue damage [4]. Furthermore, I observed that IFN- γ and IL-10 were significantly up-regulated in the spleen of infected fish by day 7 pi. These results support the observation of previous studies

using mammalian model systems [11, 36, 55] and suggest a role for both IFN- γ and IL-10 in regulation of *T. carassii* infection in goldfish.

I found a delayed (day 14 pi) and significant up-regulation of the antimicrobial genes encoding iNOS isoforms, especially on day 21 pi in all tissues examined. It has been reported that *T. carassii* does not induce significant NO response in the carp [51], while parasite antigen (hsp70) has been shown to induce NO response of goldfish macrophages *in vitro* [38]. *T. carassii* is also susceptible to NO [51], suggesting that a mechanism may exist *in vivo*, to dampen a significant NO response particularly during the establishment of *T. carassii* infection in the goldfish. It has been established that while *T. brucei*-infected mice exhibit pro-inflammatory cytokine responses and NO response [52], the infected humans have only a marginal increase in plasma nitrite levels [26]. Interestingly, humans infected with African trypanosomes, also have significantly elevated levels of Th1 cytokine, IFN₇ [26].

The increased expression of pro-inflammatory genes during the course of the infection paralleled the up-regulation of anti-inflammatory cytokine genes, such as IL-10, suggesting Th1/Th2 cross-regulation may be important in the outcome of *T. carassii* infection in the goldfish. The regulatory T-cells (Tregs), which produce IL-10 and TGF- β are induced by bacterial, viral and parasite antigens. The Tregs can suppress protective Th1 responses thereby encouraging parasite survival [13, 19, 25, 31, 32]. In support of this are reports that demonstrated that antigens of *Plasmodium falciparum* [45] and *Onchocerca*

volvulus [13] induce IL-10-producing T cells that caused a down-regulation of T-cell proliferation.

I reported for the first time a comprehensive analysis of expression of genes that encode pro- and anti-inflammatory cytokines during both acute and elimination phases of a protozoan infection in fish. My findings indicate that IFN- γ and IL-10 may play significant role in the progression of host immune responses in fish trypanosomiasis.



Figure 4.1. Image of *in vitro* **cultivated** *T. carassii*. Wright's-stained smear of *T. carassii* grown for 6 days in culture medium supplemented with 10% goldfish serum. Adapted from [10].

Days post-infection	Number of Trypanosomes/mL
	(log_{10}) (Mean ± SEM)
3	4.32 ± 0.82
7	5.18 ± 0.76
14	6.02 ± 1.01
21	5.92 ± 1.17
28	5.07 ± 1.13
56	4.12 ± 0.90

Table 4.1. Parasitemia in goldfish infected with *T. carassii*. Fish (n = 6) were inoculated with 6.25 x 10^6 *in vitro* grown *T. carassii*. Blood samples were collected from individual fish at the indicated time period and the number of parasites was enumerated. The results are mean number of parasites/mL of blood \pm SEM.



Figure 4.2. Growth of *T. carassii in vitro* in medium supplemented with 10% heat-inactivated goldfish serum. Trypanosomes were inoculated at 1×10^5 cells/mL into 25 cm² culture flasks (n = 4). Number of parasites was enumerated at the indicated time period. Each dot represents mean number parasites/mL of medium ± SEM.



Figure 4.3. Quantitative expression analysis of goldfish immune genes in the kidney of *T. carassii*-infected and non-infected fish. The reported expression was relative to EF-1 α . The expressions were normalized against those observed in non-infected control fish respectively for each gene. Statistical analysis was performed using ANOVA (n = 4 for each experimental group/observation period).



Figure 4.4. Quantitative expression analysis of goldfish immune genes in the spleen of *T. carassii*-infected and non-infected fish. The reported expression was relative to EF-1 α . The expressions were normalized against those observed in non-infected control fish respectively for each gene. Statistical analysis was performed using ANOVA (n = 4 for each experimental group/observation period).



Figure 4.5. Quantitative expression analysis of goldfish immune genes in the liver of *T. carassii*-infected and non-infected fish. The reported expression was relative to EF-1 α . The expressions were normalized against those observed in non-infected control fish respectively for each gene. Statistical analysis was performed using ANOVA (n = 4 for each experimental group/observation period).



Figure 4.6. Quantitative expression analysis of goldfish iNOS genes in *T*. *carassii*-infected and non-infected fish. The reported expression was relative to EF-1 α . The expression data were normalized against those observed in non-infected control fish respectively for each gene in the different tissues. Statistical analysis was performed using ANOVA (n = 4) for both experimental groups/observation period) and the results were deemed to be significant at *p* <0.05. (*) denotes significantly different (*p* <0.05) from the respective controls.





4.4 References

- Abrahamsohn, I. A., and R. L. Coffman. 1996. *Trypanosoma cruzi*: IL-10, TNF, IFN-y, and IL-12 regulate innate and acquired immunity to infection. Experimental Parasitology 84: 231-244.
- Agüero, F., V. Campo, L. Cremona, A. Jäger, J.M Di Noia, P. Overath, D.O Sánchez, and A.C. Frasch. 2002. Gene discovery in the freshwater fish parasite *Trypanosoma carassii*: identification of transialidase-like and mucin-like genes. Infection and Immunity **70**: 7140-7144.
- Antunez, M. I., and R. L. Cardoni. 2000. IL-12 and IFN-γ production, and NK cell activity, in acute and chronic experimental Trypanosoma cruzi infections. Immunology letters 71:103-109.
- Arend, W. P. 2002. The balance between IL-1 and IL-1Ra in disease. Cytokine Growth Factor Review 13: 323-340
- Bakhiet, M., T. Olsson, C. Edlund, B. Hojeberg, K. Holmberg, J. Lorentzen, and K. Kristensson. 1993. *Trypanosoma brucei brucei* derived factors that triggers CD8⁺ lymphocytes to interferon gamma secretion: purification, characterization and protective effects in vivo by treatment with monoclonal antibodies against the factor. Scandinavian Journal of Immunology **37**: 165-178
- Bakhiet, M., T. Olsson, J. Mhlanga, P. Buscher, N. Lycke, P. H. van der Meide, and K. Kristensson. 1996. Human and rodent IFN-γ as a growth factor for *Trypanosoma brucei brucei*. European Journal of Immunology 26: 1359-1364.
- Bakhiet, M., L. Jansson, P. Buscher, R. Holmdahl, K. Kristensson, and T. Olsson. 1996. Control of parasitemia and survival during *Trypanosoma brucei brucei* infection is related to strain dependent ability to produce interleukin-4. Journal of Immunology 157: 3518-3526
- Belosevic, M., D. S. Finbloom, P. H. Van Der Meide, M. V. Slayter, and C. A. Nacy. 1989. Administration of monoclonal anti-IFN- antibodies in vivo abrogates natural resistance of C3H/HeN mice to infection with *Leishmania major* Journal of Immunology 143: 266-274.

- Bienek, D. R., and M. Belosevic. 1997. Comparative assessment of growth of *Trypanosoma danilewskyi* (Laveran & Mesnil) in medium containing fish or mammalian serum. Journal of Fish Diseases 20: 217-221.
- Bienek, D. R. 2001. The biological and immunological aspect of the hostparasite relationship of goldfish, *Carassius auratus*, infected with *Trypanosoma danilewskyi*. PhD thesis. Department of Biological sciences, University of Alberta: Edmonton
- Cardillo, F., J. C. Voltarelli, S. G. Reed, J. S. Silva. 1996. Regulation of *Trypanosoma cruzi* infection in mice by gamma interferon and interleukin 10: role of NK cells. Infection and Immunity 64: 128-134.
- Darji, A., A. Beschin, M. Sileghem, H. Heremans, L. Brys, and P. De Baetselier. 1996. *In vitro* simulation of immunosuppression caused by *Trypanosome brucei*: active involvement of gamma interferon and tumor necrosis factor in the pathway of suppression. Infection and Immunity 64: 1937-1943.
- Doetze, A., J. Satoguina, G. Burchard, T. Rau, C. Loliger, B. Fleischer, and A. Hoerauf. 2000. Antigen-specific cellular hyporesponsiveness in a chronic human helminth infection is mediated by Th3/Tr1-type cytokines IL-10 and transforming growth factor-β but not by a T(h)1 to T(h)2 shift. International Immunology **12:** 623–630
- Dong, C., and R. A. Flavell. 2001. Th1 and Th2 cells. Current Opinion in Hematology 8: 47-51.
- Gazzinelli, R. T., I. P. Oswald, S. Hieny, S. L. James, and A. Sher. 1992. The microbicidal activity of interferon-y treated macrophages against *Trypanosoma cruzi* involves an L-arginine-dependent, nitrogen oxidemediated mechanism inhibitable by IL-10 and transforming growth factor-B. European Journal of Immunology **22**: 2501-2506.
- Gutierrez, F. R. S., T. W. P. Mineo, W. R. Pavanelli, P. M. Guedes , J. S. Silva. 2009. The effects of nitric oxide on the immune system during *Trypanosoma cruzi* infection. Memorial Institute Oswaldo Cruz 104: 236-45.

- Islam, A. K., and P. T. Woo. 1991. *Trypanosoma danilewskyi* in *Carassius auratus*: the nature of protective immunity in recovered goldfish. Journal of Parasitology 77: 258–262
- Katzenback, B. A., D. A. Plouffe , G. Haddad, M. Belosevic. 2008. Administration of recombinant parasite β-tubulin to goldfish (*Carassius auratus* L) confers partial protection against challenge with *Trypanosoma danilewskyi* Laveran and Mesnil,1904. Veterinary Parasitology 151: 36–45.
- Levings, M. K., R. Sangregorio, F. Galbiati, S. Squadrone, R. de Waal Malefyt, and M. G. Roncarolo. 2001. IFN-α and IL-10 induce the differentiation of human type 1 T regulatory cells. Journal of Immunology 166: 5530–5539
- 20. Liu, Y., E. Ragaa, Z. Li, L. Nuortio, A. Mustafa, and M. Bakhiet. 1999. Interferon-γ and interleukin-12 genes are preferentially expressed during early experimental African trypanosomiasis and suppressed by denervation of the spleen. Scandinavian Journal of Immunology **50**: 485-491.
- 21. Lom, J., and I. Dykova. 1992. Protozoan parasites of fishes. Developments in aquaculture and fisheries science. Amsterdam: Elsevier.
- Lom, J. 1973. Experimental infection of goldfish with blood flagellates, in International Congress of Protozoology.
- Lom, J. 1979. Biology of fish trypanosomes and trypanoplasms, in Biology of Kinetoplastida, W.H.R. Lumsden and D.A. Evans, Editors., Academic Press: New York. P. 269-337.
- Lucas, R., S. Magez, R. De Leys, L. Fransen, J. P. Scheerlinck, M. Rampelberg, E. Sablon, and P. De Baestselier. 1994. Mapping the lectinlike activity of tumor necrosis factor. Science 263: 814-817
- 25. MacDonald, A. J., M. Duffy, M. T. Brady, S. McKiernan, W. Hall, J. Hegarty, M. Curry, and K. H. Mills. 2002. CD4 T helper type 1 and regulatory T cells induced against the same epitopes on the core protein in

hepatitis C virus infected persons. Journal of Infectious Disease **185:** 720–727

- MacLean, L., M. Odiit, and J. M. Sternberg. 2001. Nitric oxide and cytokine synthesis in human African trypanosomiasis. Journal of Infectious Disease 184: 1086-1090.
- 27. Magez, S., B. Stijlemans, M. Radwanska, E. Pays, M. A. J. Ferguson, and P. De Baetselier. 1998. The glycosyl-inositol-phosphate and dimyristoyl glycerol moieties of the glycosylphosphatidylinositol anchor of the trypanosome variant-specific surface glycoprotein are distinct macrophage-activating factors. Journal Immunology **160**: 1949-1956.
- 28. Maina, N., J. M. Ngotho, T. Were, J. K. Thuita, D. M. Mwangangi, J. M. Kagira, M. Ndung'u, and J. Sternberg. 2004. Proinflammatory cytokine expression in the early phase of *Trypanosoma brucei rhodesiense* infection in vervet monkeys (*Cercopithecus aethiops*). Infection Immunity **72**: 3063-3065.
- Matthews, K. R., and K. Gull. 1994. Cycles within cycles: The interplay between differentiation and cell division in *Trypanosoma brucei*. Parasitology Today 10: 473-476.
- 30. Matthews, K. R., J. R. Ellis, and A. Paterou. 2004. Molecular regulation of the life cycle of African trypanosomes. Trends in Parasitology **20:** 40-47.
- McGuirk, P., and K. H. G. Mills. 2002. Pathogen-specific regulatory T cells provoke a shift in the Th1/Th2 paradigm in immunity to infectious diseases. Trends in immunology 33: 450-455.
- 32. McGuirk, P., C. McCann, and K. H. Mills. 2002. Pathogen-specific T regulatory 1 cells induced in the respiratory tract by a bacterial molecule that stimulates interleukin-10 production by dendritic cells: a novel strategy for evasion of protective T helper type 1 responses by *Bordetella pertussis*. Journal of Exprimental Medicine **195**: 221–231
- Mosmann, T. R., and S. Sad. 1996. The expanding universe of T-cell subsets: Th1, Th2 and more. Immunol Today 17: 138-146.

- 34. Muller, I., J. A. Garcia-Sanz, R. Titus, R. Behin, and J. Louis. 1989. Analysis of the cellular parameters of the immune responses contributing to resistance and susceptibility of mice to infection with the intracellular parasite, *Leishmania major*. Immunology Review **112**: 95-113.
- 35. Murray, J. S. 1998. How the MHC selects Th1/Th2 immunity. Immunology Today **19:** 157-163.
- Nabors, G. S., and R. L. Tarleton. 1991. Differential control of IFN-γ and IL-2 production during *Trypanosoma cruzi* infections. Journal of Immunology 146: 3591-3598.
- 37. Namangala, B., W. Noel, P. De Baetselier, L. Brys, and A. Beschin. 2001.
 Relative contribution of interferon-γ and interleukin-10 to resistance to murine African trypanosomiasis. Journal of Infectious Disease 183: 1794-1800
- 38. Oladiran, A., and M. Belosevic. 2009. *Trypanosoma carassii* hsp70 increases expression of inflammatory cytokines and chemokines in macrophages of the goldfish (*Carassius auratus* L.). Developmental and Comparative Immunology **33:** 1128–1136.
- 39. Olsson, T., M. Bakhiet, B. Hojeberg, A. Ljungdahl, C. Edlund, G. Andersson, H. P. Ekre, W. P. Fung-Leung, T. Mak, H. Wigzell, U. Fiszer, and K. Kristensson. 1993. CD8 is critically involved in lymphocyte activation by a *Trypanosoma brucei brucei* released molecule. Cell **72**: 715-727
- Overath, P., J. Ruoff, Y. D. Stierhof, J. Haag, H. Tichy, I. Dykova, and J. Lom. 1998. Cultivation of bloodstream forms of *Trypanosoma carassii*, a common parasite of freshwater fish. Parasitology Research 84: 343–347.
- 41. Overath, P., J. Haag, M. G. Mameza, and A. Lischke. 1999. Freshwater fish trypanosomes: definition of two types, host control by antibodies and lack of antigenic variation. Parasitology **119**: 591–601.
- Paulnock, D. M., and S. P. Coller. 2001. Analysis of macrophage activation in African trypanosomiasis. Journal Leukocyte Biology 69: 685-690.

- 43. Pearce, E. J., P. Caspar, J. M. Grzych, F. A. Lewis, A. Sher. 1991.
 Downregulation of Th1 cytokine production accompanies induction of Th2 responses by parasitic helminth, *Schistosoma masoni*. Journal of Experimental Medicine 173: 159-166.
- Peck, A., and E. D. Mellins. 2010. Precarious balance: Th17 cells in host defense. Infection and Immunology 78: 32-38.
- 45. Plebanski, M., K. L. Flanagan, E. A. Lee, W. H. Reece, K. Hart, C. Gelder, G. Gillespie, M. Pinder, A. V. Hill. 1999. Interleukin10-mediated immunosuppression by a variant CD4 T-cell epitope of *Plasmodium falciparum*. Immunity 10: 651–660
- 46. Plouffe, D. A. 2005. Immunity of goldfish *Carassius auratus* (L.), to *Trypanosoma danilewskyi* (Laveran and Mesnil, 1904). PhD thesis.
 Department of Biological sciences, University of Alberta: Edmonton.
- 47. Quadri, S. S. 1962. An experimental study of the life cycle of *Trypanosoma danilewskyi* in the leech, *Hemiclepis marginata*. Journal of Protozoology 9: 254-258.
- Reed, S. G., and P. Scott. 1993. T-cell and cytokine responses in leishmaniasis. Current Opinion Immunology 5: 524-531.
- 49. Ribeiro Carla, M. S., J. S. L. Pontes Maria, S. Bird, M. Chadzinska, M. Scheer, B. M. Lidy Verburg-van Kemenade, F. J. Savelkoul Huub, and G. F. Wiegertjes. 2010. Trypanosomiasis-induced Th17-like immune responses in carp. PLos One 5: e13012.
- So. Rottenberg, M., C. Lindqvist, A. Koman, E. L. Segura, A. Orn. 1989. Modulation of both interleukin-2 receptor expression and interleukin-2 production during experimental *Trypanosoma cruzi* infection. Scandinavian Journal of Immunology **30**: 65-72.
- Saeij, J. P. J., W. B. Van Muiswinkel, A. Groeneveld, and G. F. Wiegertjes. 2002. Immune modulation by fish kinetoplastid parsites: role for nitric oxide. Parasitology 124: 77-86
- 52. Schleifer, K. W., and J. M. Mansfield. 1993. Suppressor macrophages in African trypanosomiasis inhibit T cell proliferative responses by nitric

oxide and prostaglandins. Journal of Immunology Immunology **151:** 5492-5503.

- 53. Scott, P. 1989. The role of Th1 and Th2 cells in experimental cutaneous leishmaniasis. Experimental Parasitology **68:** 369-372.
- 54. Sharafeldin, A., M. Hamadien, A. Diab, H-L. Li, F-D. Shi, and M. Bakhiet. 1999. Cytokine profiles in the central nervous system and the spleen during the early course of experimental African trypanosomiasis. Scandinavian Journal of Immunology 50: 256-261.
- 55. Silva, J. S., P. J. Morrissey, K. H. Grabstein, K. M. Moller, D. Anderson, S. G. Reed. 1992. Interleukin 10 and interferon-y regulation of experimental *Trypanosoma cruzi* infection. Journal of Experimental Medicine **175**: 169-174
- Sternberg, J. M. 1998. Immunobiology of African trypanosomiasis. Chemical Immnnology 70: 186-199.
- 57. Wang, R., and M. Belosevic. 1994. Cultivation of *Trypanosoma danilewskyi* (Laveran & Mesnil 1904) in serum-free medium and assessment of course of infection in goldfish, *Carassius auratus (L.).* Journal of Fish Disease **17:** 47–56.
- 58. Woo, P.T.K. 1969. The haematocrit centrifuge for the detection of trypanosomes in blood. Canadian Journal of Zoology **47**: 921-923.
- 59. Woo, P. T. K. 1981. Acquired immunity against *Trypanosoma danilewskyi* in goldfish, *Carassius auratus*. Parasitology **83:** 343–346.
- Woo, P.T.K. and G.A. Black. 1984. *Trypanosoma danilewskyi*: Host specificity and host's effect on morphometrics. Journal of Parasitology **70**: 788-793.

CHAPTER 5

IDENTIFICATION OF HEAT SHOCK PROTEIN 70 (hsp70) AND OTHER ANTIGENS IN EXCRETORY/SECRETORY PRODUCTS AND SURFACE PROTEIN FRACTION FROM *Trypanosoma carassii*¹.

5.0 Introduction

Excreted/secreted (ES) and surface molecules of protozoan parasites are important for establishing infection, protecting parasites from early attack of host immune system, and acting as invasive as well as evasive agents [10, 41, 19]. These molecules can increase parasite survival and help maintain long-lasting infection by interfering with macrophage antimicrobial functions, cytokine production, antigen presentation, and effector cell activation [19, 36]. These effects are achieved by variety of mechanisms including down-regulation of gene expression, protein degradation or post-translational modification, interference with host cell signaling, and activation of suppressive pathways and molecules [11, 17, 19, 36]. Many of the parasite proteins that contribute to immune evasion or modulation have been identified in both ES products and on parasites surface [8, 30, 34, 49].

The heat shock proteins (hsps) are highly conserved molecules and are present in subcellular compartments in eukaryotes and prokaryotes. Under normal

¹A version of this chapter has been published: Oladiran, A., and M. Belosevic. 2009. *Trypanosoma carassii* hsp70 increases expression of inflammatory cytokines and chemokines in macrophages of goldfish (*Carassius auratus L*). Developmental and Comparative Immunology **33**: 1128-1136.

physiological conditions, hsps are expressed at low levels but during conditions of stress a vast array of physiological, environmental and pathological stimuli can induce significant up-regulation of hsp mRNAs [29].

The heat shock response is a general homeostatic mechanism that protects cells or organisms from damaging effects of stressful stimuli. Hsps have been termed molecular chaperones because they play crucial roles in protein folding and unfolding, protein translocation, assembly and disassembly of protein complexes. However, there is increasing evidence for the involvement of hsps in antigen processing and presentation and in other immunological processes. For example, the formation of multimeric molecules such as T-cell receptors (TCR), immunoglobulins (Ig) and major histocompatibility complex (MHC) requires functional hsps [52].

During infection, pathogens are faced with an array of stressful conditions such as changes in temperature, pH and immune responses of the host. To survive inside the host, pathogens must activate several immune evasion strategies including synthesis of hsps [52]. The importance of hsps during infection has been reported [9]. For example, mutation in *Salmonella typhimurium* HtrA gene, which encodes a chaperone results in increased susceptibility to killing by activated macrophages and produces a non-virulent bacteria *in vivo* [26]. Although a high degree of similarity exists between parasite and host hsps, parasite hsps have been reported to be immunogenic probably because of selective immune responses to the non-homologous regions in the C-terminus of hsps sequences [14, 32, 47]. There is about 73% amino acid sequence similarity between human and *Trypanosoma cruzi* hsp70, yet *T. cruzi* hsp70 induces significant antibody response during human infection [14]. Surprisingly, sera from *Leishmania donovani* infected patients recognize *L. donovani* hsp70 and hsp90 but not *T. cruzi* hsps despite more than 80% amino acid identity between these proteins [12]. This suggests that hsps may be exploited for serodiagnostic purposes for mixed infections in endemic countries.

The role of hsps in induction of proliferative responses in immune cells, as well as their immunostimulatory activities, is currently generating interests. Hsps have been reported to directly activate macrophages to secrete cytokines, stimulate proliferation of NK cells and induce maturation of dendritic cells in mammals [3, 16, 28, 33, 40]. These effects are believed to be due to interaction of hsps with specific receptors on target cells [4, 5, 48]. Trypanosomes and related parasites are known to possess all major families of hsps with hsp40, hsp60 and hsp70 family members present in large numbers indicating they play important roles in parasite biology [15]. Unfortunately, the precise role of these parasite hsps in modulation of host immune responses by the parasite is yet to be elucidated. Hsp70 from *Plasmodium falciparum* has been reported to be a major immunogen in protein fraction shown to experimentally protect *Saimiri sciureus* monkeys [6].

The extracellular existence of *T. carassii* predisposes it to direct contact with components of both the cellular and humoral branches of the immune system. Goldfish are highly susceptible to trypanosome infection and fish that survive the infection become immune to homologous challenge [50, 51]. Also,

157
passive transfer studies have shown that immunity to re-infection can be transferred to naïve goldfish using plasma or purified IgM from a recovered host [27, 31, 37]. Resistance to re-infection was long lasting and was believed to be due to the sequestration of parasites in the internal organs resulting in constant priming of the immune system [25, 50, 51]. Protective antibodies have been induced in naïve goldfish after immunization with parasite excretory/secretory products [7] or parasite beta tubulin [27]. These results provides strong evidence for antibody-mediated control of *T. carassii* infection in goldfish, and molecules in ES products as well as surface protein fractions may be crucial for vaccine development to combat this infection.

In this chapter, I report on identification of an immunogenic hsp70 from *T*. *carassii*. My working hypothesis was that *T. carassii* hsp70 modulates host immune defense by enhancing cytokine gene expression and antimicrobial responses of goldfish macrophages. *T. carassii* hsp70 was recognized by serum from infected fish among other excreted/secreted (ES) molecules and was present in solubilized membrane protein fraction. Furthermoree, recombinant parasite hsp70 enhanced expression of pro-inflammatory cytokines and induced nitric oxide response of goldfish macrophages.

5.1 Experimental Design

5.1.1 Proteomic analysis of ES products and surface protein fraction.

ES products and membrane protein fractions were prepared from *in vitro* cultured bloodstream parasites. ES proteins separated by SDS-PAGE were

immunoblotted with immune sera collected from infected fish. ES were also separated with SDS-PAGE and stained with Coomasie blue. Proteins that reacted with immune sera were excised from SDS-PAGE gels and identified by MS. *T. carassii* surface protein fractions were separated by two-dimensional electrophoresis (2-DE) and selected proteins were identified by mass spectrometry.

5.1.2 Molecular cloning and expression of *T. carassii* hsp70

Protein band corresponding to 68-70KDa antigen detected on immunoblot of ES product with sera from infected fish was identified by MS as *T. carassii* hsp70. Hsp70 fragment was amplified using primers designed to *T. carassii* spliced leader sequence and peptides identified in MS studies (IFDLGGGTFDVTLL and VVLVGGSTRIPKVMQ). The full sequence of Tcahsp70 was amplified, cloned and expressed in prokaryotic system. Purified rTcahsp70 was used to immunize rabbits for production of polyclonal antiserum.

5.1.3 Assessment of effects of recombinant *T. carassii* hsp70 on cytokine gene expression in goldfish macrophages

Macrophage cultures prepared from four individual fish (n = 4) were seeded into at a density of 1×10^6 cells per well into individual wells of 96-well plates. Cells were treated with medium alone or 5000 ng/mL of *T. carasii* hsp70 (rhsp70) for 3 hours and 6 hours. Total RNA was isolated from cells and reverse transcribed into cDNA. cDNAs were used for Q-PCR and relative expression was determined in relation to EF-1 α . All expression data were normalized using the baseline expression at 0 hour time point.

To show that rTcahsp70 effects were due to cell surface signaling, macrophages pre-treated with either culture medium only or 10 mM pronase for 30 min [3] were washed twice with incomplete medium and incubated with 5000 ng/mL rTcahsp70. Total RNAs isolated at 0, 3 and 6 hours post treatment were used in Q-PCR.

5.1.4 Assessment of effects of recombinant *T. carassii* hsp70 on antimicrobial response of goldfish macrophages

Individual cultures of primary macrophages obtained from kidney of goldfish were seeded into individual wells of 96-well plates at a density of 3 x 10^5 cells per well. Cells were then treated with culture medium alone (control) or with either heat-killed *Aeromonas salmonicida* or different concentrations of rTcahsp70 (500 ng/mL, 5000 ng/mL or 10000 ng/mL) and incubated at 20°C for 72 h. Nitrite production was determined using the Griess reaction. To ascertain that the results were due to rTcahsp70 effects, *E. coli* bacterial lysates were prepared and subjected to the same purification protocol as rTcahsp70-containing lysates. Volumes of purified *E. coli* lysate equivalent to volumes of rTcahsp70 were used in NO assay were assayed.

5.2 Results

5.2.1 Identification of *T. carassii* antigens in ES products

Previously, it was reported that sera from infected fish recognized several high molecular weight proteins, 102-104, 68-72 and 50-59 KDa present in ES supernatants of *in vitro* grown *T. carassii* [7]. The 50-59 KDa proteins were identified as *T. carassii* alpha and beta tubulins [39]. I repeated these experiments and my analysis of MS results of ES antigens led to identification of the 68-72 KDa antigen as parasite hsp70. *T. carassii* antigens identified in ES products so far are shown in Table 3.5.

5.2.2 Proteomic analysis of *T. carassii* surface protein fraction

I used Triton X-114 to solubilize parasite membrane proteins and analyzed the solubilized proteins by 2-DE. The reproducibility of my sample preparation, IEF, gel electrophoresis and gel staining was tested by running 4 replicate gels with proteins prepared from different batches of parasites cultured on different days. In all cases, approximately 300 protein spots were detected using same detection parameters. This suggests that the majority of the proteins in solubilized membrane fraction could be resolved with a high degree of reproducibility (Fig. 5.1). I selected 30 protein spots for mass spectrometry analysis based on the results presented in Table 3.5 that indicated that goldfish immune serum recognized primarily higher molecular weight proteins (55-97KDa) in immunoblotting of ES products. The full list of protein identified in surface protein fraction is shown in Table 3.6. Proteins such as tubulin, hsps and enolase were detected in more than one spot. This probably reflects the highly abundant nature of these proteins and their degradation products. Also, several isoforms of these proteins may have been produced. Their abundant nature suggests they may play important roles in parasite interaction with the host.

5.2.3 T. carassii hsp70 (Tcahsp70) is immunogenic

The first attempt at identifying *T. carassii* ES proteins came from the observation that sera from infected fish recognized several high molecular weight proteins in ES supernatants of *in vitro* grown *T. carassii* [7]. By using a more concentrated ES preparation, I showed that additional proteins were recognized by immune serum (Table 3.5). Protein band corresponding to the 68-78KDa antigen excised from a Coomassie stained SDS-PAGE gel was identified as parasite hsp70 using mass spectrometry (Fig. 5.2A). Tcahsp70 was cloned and its full sequence was determined. Alignment of Tcahsp70 with hsp70 from other trypanosomatids as well as sequences from mammalian and non-mammalian vertebrates is shown in Fig.5 3. Fig. 5.4 shows the protein structure of *T. carassii* hsp70 and T. cruzi hsp70. Recombinant parasite hsp70 was produced in prokaryotic system and purified using the N-terminal His-tag. Figure 5.2C shows detection of rTcahsp70 using anti-His antibody. Interestingly, the T. carassii recombinant hsp70 was also recognized by fish immune serum as shown in Figure 5.2D, suggesting proper folding of the recombinant molecule.

5.2.4 *T. carassii* hsp70 is present in surface protein fraction

Parasite hsp70 have been reported in the soluble antigens and in the secretome of *Leishmania donovani* [24, 46]. Because *T. carassii* hsp70 was found in the ES preparations, I examined whether *T. carassii* hsp70 was present in solubilized surface protein fraction of the parasite. Parasite hsp70 were detected in more than one spot (Fig. 5.5) in 2-D gel of solubilized surface proteins suggesting degradation products or different forms of this molecule.

5.2.5 Recombinant parasite hsp70 induces expression of pro-inflammatory genes in goldfish macrophages

The ability of recombinant hsp70 (rhsp70) to directly activate macrophages to secrete cytokines, stimulate proliferation of NK cells and induce maturation of dendritic cells has been reported for mammalian and bacterial hsp70 [3, 16, 40]. For this reason, I examined whether rTcahsp70 can modulate proinflammatory gene expression of goldfish macrophages. Macrophage proinflammatory genes examined include cytokines (IFN γ , TNF α -1, TNF α -2; IL-1 β -1, IL-1 β -2; IL-12-p35, IL-12-p40), chemokines (CXCL-8, CCL-1), and iNOS isoforms A and B at different times after treatment with rhsp70. Treatment of macrophages with rTcahsp70 results in up-regulated expression of IFN γ , TNF α 1 and TNF α 2 (Fig. 5.6 A and B) and the expression of IL-1 β -1, IL-1 β -2 and IL-12p35 (Fig. 5.6 C and D). I also quantified the expression of chemokines CCL-1 and CXCL-8 (IL-8) in macrophages activated with rTcahsp70. rTcahsp70 upregulated expression of both chemokines (Fig. 5.7) That hsp70 effects were induced following interaction of hsp70 with receptor(s) on the surface of activated goldfish macrophages was supported by the observation that pre-treatment of macrophages with pronase abrogated the expression of select immune genes (Fig. 5.8).

5.2.6 Recombinant parasite hsp70 induces nitric oxide response of goldfish macrophages

Because T. carassii hsp70 may be secreted during infection or expressed on the parasite surface, I examined whether the rTcahsp70 induced nitric oxide response in activated goldfish macrophages. Primary macrophages obtained from goldfish kidney were used to establish cultures from individual fish and treated with culture medium (control), either heat-killed A. salmonicida or rTcahsp70. Macrophages were treated with 5000 ng/mL rhsp70 because this concentration has been shown to be effective in a previous report [40]. As shown in Figure 5.9, T. carassii rhsp70 induced increased expression of both iNOS isoforms as well as a strong nitrite production by activated goldfish macrophages (Fig. 5.10). The bacterial lysates subjected to the same purification regimen as those containing rTcahsp70 were also tested for their ability to induce nitric oxide response in goldfish macrophages (Fig 5.11). One way analysis of variance indicated that rTcahsp70 treatment induced significant (P < 0.05) nitric oxide response of goldfish macrophages when compared to either medium control or bacterial lysate groups.

5.3 Discussion

The identity of proteins secreted/excreted by T. carassii as well as those expressed on the parasite surface is of particular interest to us because of the possible roles these molecules might play in host-parasite interactions. It has been previously shown that immunization of fish with ES products from *in vitro* cultured parasites increased fish resistance to challenge infection [7]. Parasite alpha and beta tubulins were identified in the ES product and immunization studies showed that recombinant beta tubulin partially protected fish against a challenge infection [27, 39]. I [35] reported that T. carassii hsp70 is also one of the major proteins in ES supernatants and that it was identified in the solubilized parasite membrane fraction. Although it is not clear how T. carassii may export hsp70 to its surface, my observations are consistent with reports from other studies that identified parasite hsp70 in soluble antigen preparations and secretome of L. donovani [24, 46]. Absence of a clear N-terminal signal sequence in T carassii hsp70, may suggest the existence of a non-classical secretory pathway for this protein. The fact that T. carassii hsp70 was secreted and expressed on the parasite surface, suggests that it may play a role in the hostparasite interactions. Indeed hsp70 has been reported to be an important antigen in several infectious diseases [52].

During infection, fish macrophages respond by up-regulating the expression and secretion of cytokines including TNF- α , IL-1 and enhanced nitric oxide response [13, 42, 43, 44, 45]. Activation of macrophages by parasite antigens is common in trypanosome infection [22, 23, 38]. My results further

support these earlier studies and suggest interaction between T. carassii hsp70 and goldfish macrophages potentially leading to development of Th1-like immune response. Recombinant hsp70 from T. carassii activated goldfish macrophages and up-regulated expression of cytokines IFNy, TNFa-1, TNFa-2, IL-1β-1, IL-1β-2; IL-12-p35 and chemokines CXCL-8, CCL-1. The expressions of iNOS isoforms A and B were also up-regulated but at different levels after treatment with rTcahsp70. I did not see a significant increase in expression for IL-12-p40 at the time points examined. It is possible that this cytokine is expressed at later time points as shown previously in goldfish macrophages stimulated with recombinant goldfish interferon gamma [18] Previous reports have shown possible links between cytokine expression and protozoan infection in fish. Trypanoplasma *borreli* induced differential expression of TNF- α with a higher expression of TNF α -2 in carp head kidney phagocytes *in vitro* and up-regulated IL-1 β mRNA *in* vivo [32-34]. T. borreli also stimulates nitrite production by carp phagocytes in vitro [45]. In contrast, these authors reported that T. carassii organisms when added to macrophages did not induce nitric oxide response in vitro and that there was no change in serum nitrate level in fish infected with this parasite [44]. I have also observed that T. carassii induces ROI response of goldfish monocytes but failed to induce NO response of goldfish macrophages. In this study, I showed that T carassii recombinant hsp70 induced a strong nitric oxide response in goldfish macrophages and was demonstrated to be present in ES supernatants generated *in vitro*. My observations are consistent with those reported for T. *cruzi*, which showed that *T. cruzi* surface proteins (GPIs) are able induce a potent

nitric oxide response of mouse macrophages [2]. Interestingly, *T. carassii* and *T. cruzi* have very similar surface architecture [1, 30] that may account for differential induction of nitric oxide response of macrophages by *T. borreli* and *T. carassii*.

My results indicate a novel function of trypanosome hsp70 in ectothermic vertebrates, as a modulator of host immune response. T. carassii recombinant hsp70 induced increased expression of pro-inflammatory cytokines and chemokines in goldfish macrophages. The recombinant hsp70-induced cytokine expression was abrogated by pronase treatment of macrophages suggesting that rTcahsp70 acted through receptor(s) on macrophage surface. This recognition and signal transduction has been reported to occur partly through CD14 pathway [3]. It is also possible that hsp70 may use different receptors on different immune cells, since hsp70 activation of NK-cell mediated killing was reported to be dependent on the interaction of hsp70 with C-type lectin receptor CD94 [20, 21]. The effects of *T. carassii* hsp70 reported here may result in non-specific stimulation of immune responses that may contribute to general immunosuppression, which is one of the hallmarks of trypanosome infections in mammals. Given that T. carassii is an extracellular parasite, stimulation of proinflammatory immune response by parasite antigens may down-regulate host humoral immune responses thereby ensuring their establishment and survival in the host. We [27], and others [25, 51] have shown that humoral immune responses play a major role in the elimination and the maintenance of long-lasting resistance to reinfection against T. carassii.

Considering that hsps are highly conserved molecules, it is somewhat surprising that *T. carassii* hsp70 activated goldfish macrophages. Although *T. carassii* and goldfish hsp70 sequences are similar (68% similarity), there are notable differences between the host and parasite hsp70 sequences at the Cterminal end of the proteins. *T. carassii* hsp70 possess five GMPG repeats while goldfish hsp70 contains only one repeat. Similarly, *T. cruzi* hsp70 contains at least ten GMPG repeats and human hsp70 has no GMPG repeats at the C-terminal end of the protein. It is possible that the GMPG repeats may be major antigenic determinants of parasite hsp70. A further investigation of this repeat sequence is warranted to understand its importance in host-parasite interactions.







Figure 5.2. Immunogenicity of T. carassii hsp70

Filtered ES product was separated by SDS-PAGE and transferred to nitrocellulose (A) Immunoblotting ES products performed with serum from recovered fish diluted 1:25 in blocking buffer (56 days post infection). The 68-70kDa band (arrow) was identified as *T carassii* hsp70. (B) Immunoblot of ES products using serum from non-infected fish diluted 1:25 in blocking buffer. (C) Immunoblot of recombinant *T. carassii* hsp70 recognized by anti-His antibody. (D) Immunoblot of recombinant goldfish hsp70 recognized by immune goldfish serum (56 days post infection).

T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	- MAYEGA GILDLGTYYE YYGVYQNERVEI IAHDQORRTYPS YVAFTDTERLICDAAKNO - MTYEGA GILDLGTYYE YYGVYQNERVEI IAHDQORRTYPS YVAFTDTERLICDAAKNO - MTYFGA GILDLGTYYE YYGVYQNERVDI IAHDQORRTYPS YVAFTDERLICDAAKNO MSDAKOYA GILDLGTYYE YYGVYQNERVDI IAHDQORRTYPS YVAFTDTERLICDAAKNO - MSDAKOYA GILDLGTYYE YYGYYQNERVEI IAHDQORRTYPS YVAFTDTERLICDAAKNO - MAGAAN GILDLGTYYE YYGYYQNERVEI IAHDQORRTYPS YVAFTDTERLICDAAKNO - MAGAAN GILDLGTYYE YYGYYQNERVEI IAHDQORRTYPS YVAFTDTERLICDAAKNO - MAGAAN GILDLGTYYE YYGYYDIGYYEI IAHDQORRTYPS YVAFTDTERLICDAAKNO - MAGAAN	58 58 58 60 58
T.cruzi T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	VANNPTNTYFDARRLIGRKFSESVOSDMRIMFFKVTSKGDDRFYIQVQFHGETKTNPE VANNPINTYFDARRLIGRKFSDYVOSDMRIMFFKVTTKGDDRFYIQVQFRGETKTPNE VANNPINTYFDARRLIGRKFSDYVOSDMRIMFFKVTTKGDDRFYIQVQFRGETKTPNE VANNPINTYFDARRLIGRKFSDYVOSDMRIMFFKVTKGDDRFYIQVGRGETKTPNE VANNPINTYFDARRLIGRKFSDYVOSDMRIMFFQVTSDGC-RFXQVEYKGENKTYPE VANNPINTYFDARRLIGRKFGDYVOSDMRIMFQVISDGC-RFXQVEYKGENKTYFF VANNPINTYFDARRLIGRKFGDYVOSDMRIMFFQVISDGC-RFXQVEYKGENKTYFF VANNPINTYFDARRLIGRKFFGDYVOSDMRIMFFQVISDGC-RFXQVEYKGENKTYFF	118 118 118 118 119 109 117
T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	EISSWULVKKKEVAESYLGKPVKKAVVTVPAYPNDSOROATKDAGTIAGLEVLAI INEPT EISSWULVKKEIASYLGKVKKAVVTVPAYPNDSOROATKDAGTIAGLEVLAI INEPT EISSWULKKKEVAESYLGKVKKAVVTVPAYPNDSOROATKDAGTIAGLEVLAI INEPT EISSWULKKKEIASYLGKVVKAVVTVPAYPNDSOROATKDAGTIAGLEVLAI INEPT EISSWULKKKEIASYLGKVVTVAVITVDAYPNDSOROATKDAGTIAGLEVLAI INEPT EISSWULKKKEIASYLGKVVTVAVITVDAYPNDSOROATKDAGTIAGLEVLAI INEPT	178 178 178 178 179 169 177
T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	ADAILAYGLOKVEDAKERNVL FFDLGGGTFDVTLLFTDGRHLAGENGDERNHASWNGLKOP AAAIAYGLOKVEDGKERNVL FFDLGGGTFDVTLFTDGGTFEVKATNGDYLLGGEDFDNR AAAIAYGLOKADBGKERNVL FFDLGGGTFDVTLFTDGGTFEVKATNGDYLLGGEDFDNR AAAIAYGLOKADBGKERNVL FFDLGGGTFDVTLFTLGGTFEVKATNGDYLLGGEDFDNR AAAIAYGLOKTADBGKERNVL FFDLGGGTFDVTLFTLGGTFEVKATNGDYLLGGEDFDNR AAAIAYGLORTAGG ERNVL FFDLGGGTFDVTLFTLGGTFEVKATNGDYLLGGEDFDNR AAAIAYGLORTAGG ERNVL FFDLGGGTFDVTLFTLGGTFEVKATNGDYLLGGEDFDNR	238 238 238 238 238 238 228 236
T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	PRCALIREFYRINKGKDFBINNORALRRLHTACERISKRILBSAAQATTEIDALFORMFOA LUSEIFYDEFYRINKGKDFBINNORALRRLHTACERISKRILBSAAQATTEIDALFORMFOA NUSEIFYDEFYRINKGKDLTSGORALRRLHRACERAKKTLBSBAOATEIDALFON LUAEIFTEFYRINKGKDLBSULALRRLHRACERAKKTLBSBOABTEIDALFFIN NUNEFYEFYRINKGKDLBSULARLHRALRTACERAKKTLBSBOABTEIDALFFIN LUXEIFYDEFYRRKKK-KDISONKIAVRALRTACERAKKTLBSBOABTEIDSLFEGIDFYT LUXEIFYDEFYRRKKK-KDISONKIAVRALRTACERAKKTLBSBTOASLEIDSLFEGIDFYT	298 298 298 297 287 295
T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	TITRARFEELCGOLFRGTOPVERVLODARMOKRAVULVVLGGSTRIFKVMCVSDFFG TITRARFEELCGOLFRGTLOPVERVLODARMOKRAVULVVLVGGSTRIFKVMCVSDFFG TITRAFFEELCGOLFRSTJOPVERVLODARMOKRSVILVVLVGGSTRIFKVMCVSDFFG SITRAFFEELCGOLFRGTLOPVERVLODARMOKRSVILVVLVGGSTRIFKVMCVSDFFG SITRAFFEELCGOLFRGTLEPVEKALRDARMOKSQINUVVLVGGSTRIFKIQKLOPFFN SITRAFFEELCGOLFRGTLEPVEKALRDARMOKSQINUVVLVGGSTRIFKIQKLOPFFN SITRAFFEELCGOLFRSTLEPVEKALRDARMOKSQINUVVLVGGSTRIFKIQKLOPFFN	358 358 358 358 357 347 355
T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	GRELNRS INFDERAVA VGANVQGF ILFOGRSKOTEGLLLLOVTPLTLGIFTAGGUMTALIR GRELNRS INFDERAVAGANVQGF ILFOGRSKOTEGLLLLOVTPLTLGIFTAGGUMTALIR GRELNRS INFDERAVATOANVQAFILFOGRSKOTEGLLLLOVAPLTLGIFTAGGUMTALIR GRELNRS INFDERAVATGANVQAFILFOGRSKOTEGLLLLOVAPLTLGIFTAGGUMTALIR REDLINES INFDERAVATGANVQAFILFOGRSKOTEGLLLLOVAPLSLGIFTAGGUMTALIR GRELNRS INFDERAVATGANVQAFILFOGRSKOTEGLLLLOVAPLSLGIFTAGGUMTALIR GRELNRS INFDERAVATGANVQAFILFOGRSKOTEGLLLLOVAPLSLGIFTAGGUMTALIR GRELNRS INFDERAVATGANVQAFILFOGRSKOTEGLLLLOVAPLSLGIFTAGGUMTALIR GRELNRS INFDERAVATGANVQAFILFOGRSKOTEGLLLLOVAPLSLGIFTAGGUMTALIR GRELNRS INFDERAVATGANVQAFILFOGRSKOTEGLLLLOVAPLSLGIFTAGGUMTALIR GRELNRS INFDERAVATGANVQAFILFOGRSKOTEGLLLLOVAPLSLGIFTAGGUMTALIR	418 418 418 417 407 415
T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	RHTTIFTKKSOIFSTVADHOPGVIIOVFSGERAHTTKOLLILGTFDLEGIPPAPRGVVDIE RHTTIFTKSOIFSTVADHOPGVIIOVFSGERAHTTKOLLIGTFDLEGIPPAPRGVPDIE RHTTIFTKSOIFSTVADHOPGVIIOVFSGERAHTKOLLIGTFDLEGIPPAPRGVPDIE RHTTIFTKROOTISTVEDHOPGVIIOVFSGERAHTKONLLGFELGIPPAPRGVPDIE RHTTIFTKOTOTISTVEDHOPGVIIOVYSGERAHTKONNLLGFELGIPPAPRGVPDIE RHSTIFTKOTOIITTYEDHOPGVLIOVYSGERAHTKONNLLGFELGIPPAPRGVPDIE	478 478 478 478 477 467 475
T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	VTPDLDANGILWYTAEEKOTGKENG VITTDKGRLBRADIERWYGAASKTEEOOKLGRER VTPDLDANGILWYSAESKOTGKENG VITTDKGRLBRADIERWYSGAASKTEAOKRGORER VTPDLDANGILWYSAESKOTGKENG VITTDKGRLBRADIERWYSGAOKGAGAG VTPDLDANGILWYSADVAETGKENG VITTDKGRLBRADIERWYGAASKTEADCKAGAER VTPDLDANGILWYSADVAETGKENG VITTDKGRLBRADIERWYGAASKTEADCKAGAER VTPDLDANGILWYTATDKETGKANKITTITDKGRLBREEIERWYGAASKTEADCKAGAER VTPDLDANGILWYTATDKETGKANKITTITDKGRLBREEIERWYGAASKTEADCYGAESYAADCEVGRE	538 538 538 538 537 527 535
T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	VERKNOLENVAFEMKNTMNDONVAGKI DENDKSTINKAVEDALOKLNNNGEATKEEVDHI IDAKNOLENVAFEMKNTVNERVAGKI EELADKATLINKAVEDALOKLNNDGEATKEEVDHI VERKNOLENVAFEMKNTVESDSNVSGKLDDOBDKATLINKE IDALEKULSSNGEATKEEVEH IDAKNOLENVAFEMKNTIDSDSNVSGKLDDOBDKATLINKE IDALEKULSSNGEATKEEVEH IAAKNSLESVAFMKNSVEDEDLKGKI SEDDKKKVIIKKOMEAVSWLENNOLADKEEVEHH IAAKNSLESVAFMKNSVEDEDLKGKI SEDDKKKVIIKKOMEAVSWLENNOLADKEEVEHH VARKALESVAFMKNSVEDEDLKGKI SEDDKKKVIIKKOMEAVSWLENNOLADKEEVEHH VARKALESVAFMKNSVEDEDLKGKI SEDDKKKVIIKKOMEAVSWLENNOLADKEEVEHH VARKALESVAFMMKSAVEDELKGKI SEDDKKKVIIKKOMEAVSWLENNOLADKEEVEHH	598 598 598 597 597 587 595
T.cruzi T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	OKELEGICTPINKKNYGEMGGGANFPCMPFGMPFGCMFGGMFGGMFGGMFGGMFG OKKLENICTPINKKNYGEMGGGMFGGMFGGMFGGMFGGMFGGMFGGMFGGMFGGMF	645 658 641 645 631 621 629
T.carassii T.cruzi L.major T.brucei C.auratus C.carpio H.sapiens	MGGARG -ASSSCRVVIEVD- 662 MCGARFSSSCRVVIEVD- 677 ACPAG- GASSCRVVIEVD- 658 MEALARRFCLKSRRLTXFFCVPTGAAVVSLSTVMVRRHIFSSSL 690	

Figure 5.3. Amino acid alignment of *T. carassii* hsp70 sequence and those from other trypanosomatids (*T. cruzi, T. brucei* and *L. major*) as well as sequences from mammalian (*Homo sapiens*) and non-mammalian (*Carassius auratus* and *Cyprinus carpio*) vertebrates. Sequences were obtained from GenBank with accession numbers *T. cruzi* (XM_812645), *L. major* (XM_001684512), *T. brucei* (XM_824105), *C. auratus* (BAC67184), *C. carpio* (AAM81603), *H. sapiens* (NP_005336). (*) indicates identical residues, (:) indicates conserved residues while (.) indicates semi-conserved residues. (-) are gaps introduced to maximize the alignment. The boxed residues represent the three signature patterns of the hsp70 family predicted by PROSITE server. The region containing the GMPG repeats were shown with a bar at the C-terminal end of the sequences to highlight the presence of this repeat in parasites but not in the host molecules.



Figure 5.4. Schematic of protein structure of *T. carassii* hsp70. *T. carassii* hsp70 was compared with *T. cruzi* hsp70. The coloured boxes represent hsp70 family signatures. The approximate locations of the hsp70 family signatures are shown.









Quantitative expression of cytokines IFN γ , TNF α -1, TNF α -2, IL-12-p35, IL-12-p40, IL-1 β -1, IL-1 β -2 were reported relative to endogenous control EF-1 α . The expression data were normalized against 0 hour time point expression level for each gene. The results are mean ± SEM of macrophage cultures established from from four individual fish (n = 4). Statistical analysis was done with one-way ANOVA. Expression levels were considered to be significant at P < 0.05. (*) denotes significantly different from the 0 hour time point.







Figure 5.8. Analysis of immune gene expression in goldfish macrophages pretreated with pronase before incubation with rTcahsp70. Quantitative expression of select genes of goldfish macrophages pre-treated with pronase before incubation with rTcahsp70. Expression of TNF α -2, IFN γ , IL-1 β -1 and iNOS A was relative to endogenous control EF-1 α . The expression data were normalized against 0 hour time point expression level for each gene. The results are mean \pm SEM of macrophage cultures established from four individual fish (n = 4). Statistical analysis was done with one-way ANOVA. None of the expression levels were significant at P < 0.05 (*) from the 0 hour time point for each gene. Although the expression of iNOS A gene is slightly up-regulated, this expression level is about 30-fold less than expression level in macrophages incubated with rTcahsp70 without pronase pre-treatment.



Figure 5.9. *T. carassii* hsp70 up-regulates iNOS gene expression of goldfish macrophages. Quantitative expression of iNOS isoforms A and B relative to endogenous control EF-1 α in goldfish macrophages following treatment with rTcahsp70. The expression data were normalized against 0 hour time point expression level for each gene. The results are mean ± SEM of macrophage cultures established from four individual fish (n = 4). Statistical analysis was done with one-way ANOVA. Expression levels were considered to be significant at P < 0.05. (*) denotes significantly different from the 0 hour time point.



Figure 5.10. T. carassii hsp70 enhances nitric oxide response of goldfish

macrophages. Nitric oxide production by goldfish macrophages was determined using the Griess reaction and nitrite concentration determined using a nitrite standard curve. The mean \pm SEM nitrite production determined for macrophage cultures established from seven individual fish (n = 7). NO assay was performed twice with same results. At concentration greater than 5000 ng/mL, rTcahsp70 effect is unchanged probably because the system is saturated. Statistical analysis was done using one-way ANOVA. (*) denotes statistically different (P < 0.05) from medium control.



Figure 5.11. Nitric oxide response of goldfish macrophages is due to

rTcahsp70. Nitric oxide production by goldfish macrophages was determined using the Griess reaction and nitrite concentration determined using a nitrite standard curve. The mean \pm SEM nitrite production determined for macrophage cultures established from seven individual fish (n = 7). *E. coli* lysate failed to enhance NO response of goldfish macrophages. Statistical analysis was done using one-way ANOVA. (*) denotes statistically different (P < 0.05) from medium control.

5.4 References

- Agüero, F., V. Campo, L. Cremona, A. Jäger, J. M. Di Noia, P. Overath, D. O. Sánchez, and A. C. Frasch. 2002. Gene discovery in the freshwater fish parasite *Trypanosoma carassii*: identification of trans-sialidase-like and mucin-like genes Infection and Immunity **70**: 7140-7144.
- Almeida, J. C., M. M. Camargo, D. O. Procopio, L. S. Silva, A. Mehlert, L. R. Travassos, R. T. Gazzinelli, and M. A. Ferguson. 2000. Highly purified glycosylphosphatidylinositols from *Trypanosoma cruzi* are potent proinflammatory agents. EMBO Journal **19**: 1476-1485.
- Asea, A., S. K. Kraeft, E. A. Kurt-Jones, M. A. Stevenson, L. B. Chen, R. W. Finberg, G. C. Koo, and S. K. Calderwood. 2000. HSP70 stimulates cytokine production through a CD14-dependant pathway, demonstrating its dual role as a chaperone and cytokine. Nature Medicine 6: 435-442.
- Basu, S., R. J. Binder, T. Ramalingam, and P. K. Srivastava. 2001. CD91 is a common receptor for heat shock proteins gp96, hsp90, hsp70, and calreticulin. Immunity 14: 303-313.
- 5. Binder, R. J., D. K. Han, and P. K. Srivastava. 2000. CD91: a receptor for heat shock protein gp96. Nature Immunology **1:** 151-155.
- Behr, C., J. L. Sarthou, J. F. Trape, M. H. Dat, J. C. Michel, G. Aribot, A. Dieye, J. M. Claverie, P. Druhihle, and P. Dubois. 1992. Antibodies and reactive T cells against the malaria heat shock protein Pf72/Hsp70-1 and derived peptides in individuals continuosly exposed to *Plasmodium falciparum*. Journal of Immunology 149: 3321-3330.
- Bienek, D. R., D. A. Plouffe, G. F. Wiegertjes, and M. Belosevic. 2002. Immunization of goldfish with excretory/secretory molecules of *Trypanosoma danilewskyi* confers protection against infection. Developmental and Comparative Immunology 26: 649-657
- Bouvier, J., R. J. Etges, and C. Bordier. 1985. Identification and purification of membrane and soluble forms of the major surface protein of *Leishmania* promastigotes. Journal of Biological Chemistry 260: 15504-15509.

- 9. Buchmeier, N. A., and F. Heffron. 1990. Induction of Salmonella stress proteins upon infection of macrophages. Science **248**: 730-732.
- Chang, K.P., S. G. Reed, B. S. McGwire, and L. Soong. 2003. *Leishmania* model for microbial virulence: the relevance of parasite multiplication and pathoantigenicity. Acta Tropica 85: 375-390.
- Contreras, I., M. A. Gomez, O. Nguyen, M. T. Shio, and R. W. McMaster. 2010. *Leishmania*-induced inactivation of the macrophage transcription factor AP-1 is mediated by the parasite metalloprotease GP63. PLos Pathogens 6: e1001148.
- de Andrade, C. R., L. V. Kirchhoff, J. E. Donelson, and K. Otsu. 1992. Recombinant Leishmania Hsp90 and Hsp70 are recognized by sera from visceral leishmaniasis patients but not Chagas' disease patients. Journal of Clinical Microbiology 30: 330-335
- Engelsma, M. Y., R. J. Stet, J. P. Saeij, and B. M. Verburg-van Kemenade.
 2003. Differential expression and haplotypic variation of two interleukinlbeta genes in the common carp (*Cyprinus carpio L.*). Cytokine 22: 21-32.
- Engman, D. M., E. A. Dragon, and J. E. Donelson. 1990. Human humoral immunity to hsp70 during *Trypanosoma cruzi* infection. Journal of Immunology 144: 3987-3991
- 15. Folgueira, C., and J. M. Requena. 2007. A postgenomic view of the heat shock proteins in kinetoplastids. FEMS Microbiology Review **31**: 359-377.
- 16. Galdiero, M., G. C. de I'Ero, and A. Marcatili. 1997. Cytokine and adhesion molecule expression in human monocytes and endothelial cells stimulated with bacterial heat shock proteins. Infection and Immunity 65: 699–707.
- Gomez, M. A., I. Contreras, M. Halle, M. L. Tremblay, R. W. McMaster, M. Olivier. 2009. *Leishmania* GP63 alters host signalling through cleavage-activated protein tryrosine phosphatases. Science Signal 2: ra58.
- Grayfer, L., and M. Belosevic. 2009. Molecular characterization, expression and functional analysis of goldfish (*Carassius aurutus* L.) interferon gamma. Developmental and Comparative Immunology 33: 235-246.

- Gregory, D. J., and M. Olivier. 2005. Subversion of host cell signalling by the protozoan parasite *Leishmania*. Parasitology 130: S27-S35
- Gross, C., D. Hansch, R. Gastpar, and G. Multhoff. 2003a. Interaction of heat shock protein 70 peptide with NK cells involves the NK receptor CD94. Biological Chemistry 384: 267-279.
- 21. Gross, C., I. G. Schmidt-Wolf, S. Nagaraj, R. Gastpar, J. Ellwart, L. A. Kunz-Schughart, and G. Multhoff. 2003b. Heat shock protein 70-reactivity is associated with increased cell surface density of CD94/CD56 on primary natural killer cells. Cell Stress Chaperones 8: 348-360.
- Grosskinsky, C. M., and B. A. Askonas. 1983. Macrophage activation in murine African trypanosomiasis. Clinical Experimental Immunology 52: 153-143.
- Grosskinsky, C. M., and B. A. Askonas. 1981. Macrophages as primary target cells and mediators of immune dysfunction in African trypanosomes. Infection and Immunity 33: 149-155.
- 24. Gupta, S. K., B. S. Sisodia, S. Sinha, K. Hajela, S. Naik, A. K. Shasany, and A. Dube. 2007. Proteomic approach for identification and characterization of novel immunostimulatory proteins from soluble antigens of *Leishmania donovani* promastigotes. Proteomics 7: 816-823
- 25. Islam, A. K., and P. T. Woo. 1991. *Trypanosoma danilewskyi* in *Carassius auratus*: the nature of protective immunity in recovered goldfish. Journal of Parasitology 77: 258-262
- 26. Johnson, K. I. Charles, G. Dougan, D. Pickard, P. O'Gaora, G. Costa, I. Miller, and C. Hormaeche. 1991. The role of a stress-response protein in *Salmonella typhimurium* virulence. Molecular Microbiology 5: 401-407.
- 27. Katzenback, B. A., D. A. Plouffe, G. Haddad, and M. Belosevic. 2008. Administration of recombinant parasite β-tubulin to goldfish (*Carassius auratus* L) confers partial protection against challenge with *Trypanosoma danilewskyi* Laveran and Mesnil, 1904. Veterinary Parasitology **151**: 36-45.
- Kuppner, M. C., R. Gastpar, S. Gelwer, E. Nossner, O. Ochmann, A. Scharner, and R. D. Issels. 2001. The role of heat shock protein (hsp70) in

dendritic cell maturation: hsp70 induces the maturation of immature dendritic cells but reduces DC differentiation from monocyte precursors. European Journal of Immunology **31:** 1602-1609.

- 29. Lindquist, S., and E. A. Craig. 1988. The heat shock proteins. Annual Review of Genetics **22**: 631-677.
- 30. Lischke, A., C. Klein, Y. D. Stierhof, M. A. Hempel, A. Mehlert, I. C. Almeida, M. A. Ferguson, P. Overath. 2000. Isolation and characterization of glycosylphosphatidylinositol-anchored, mucin-like surface glycoproteins from bloodstream forms of the freshwater-fish parasite *Trypanosoma carassii*. Biochemistry Journal **345**: 693-700.
- 31. Lom, J., and I. Dykova. 1992. Protozoan parasites of fishes Developments in aquaculture and fisheries science, Elsevier, Amsterdam.
- 32. MacFarlane, J., M. L. Blaxter, R. P. Bishop, M. A. Miles, and J. M. Kelly. 1990. Identification and characterisation of a *Leishmania donovani* antigen belonging to the 70-kDa heat-shock protein family. European Journal of Biochemistry **190**: 377-384.
- 33. Multhoff, G., L. Mizzen, C. C. Winchester, C. M. Milner, S. Wenk, G. Eissner, H. H. Kampinga, B. Laumbacher, and J. Johnson. 1999. Heat shock protein 70 (Hsp70) stimulates proliferation and cytolytic activity of natural killer cells. Experimental Hematology 27: 1627-1636.
- 34. Norris, K.A., J. E. Schrimpf, and M. J. Szabo. 1997. Identification of the gene family encoding the 160-kilodalton *Trypanosoma cruzi* complement regulatory protein. Infection and Immunity 65: 349-357
- Oladiran, A., and M. Belosevic. 2009. *Trypanosoma carassii* hsp70 increases expression of inflammatory cytokines and chemokines in macrophages of goldfish (*Carassius auratus L*). Developmental and Comparative Immunology 33: 1128-1136.
- 36. Olivier, M., D. J. Gregory, and G. Forget. 2005. Subversion mechanisms by which *Leishmania* parasites can escape the host immune response: a signalling point of view. Clinical Microbiology Reviews 18: 293-305.

- 37. Overath, P., J. Haag, M. G. Mameza, and A. Lischke. 1999. Freshwater fish trypanosomes: definition of two types, host control by antibodies and lack of antigenic variation. Parasitology **119**: 591-560.
- Paulnock, D. M., and S. P. Coller. 2001. Analysis of macrophage activation in African trypanosomiasis. Journal of Leukocyte Biology 69: 685-690.
- 39. Plouffe, D. A., and M. Belosevic. 2006. Antibodies that recognize alpha- and beta-tubulin inhibit in vitro growth of the fish parasite *Trypanosoma danilewskyi*, Laveran and Mesnil, 1904. Developmental and Comparative Immunology **30**: 685-97.
- Retzlaff, C., Y. Yamamoto, P. S. Hoffman, H. Friedman, and T. W. Klein. 1994. Bacterial heat shock proteins directly induce cytokine mRNA and interleukin-1 secretion in macrophage cultures. Infection and Immunity 62: 5689-5693
- 41. Rittig, M.G., and C. Bogdan. 2000. *Leishmania*-host-cell interaction: complexities and alternative views. Parasitology Today **16:** 292-297
- 42. Saeij, J. P., R. J. Stet, B. J. de Vries, W. B. van Muiswinkel, and G. F. Wiegertjes. 2003a. Molecular and functional characterization of carp TNF: a link between TNF polymorphism and trypanotolerance? Developmental and Comparative Immunology 27: 29-41
- 43. Saeij, J. P., B. J. de Vries, and G. F. Wiegertjes. 2003b. The immune response of carp to *Trypanoplasma borreli*: kinetics of immune gene expression and polyclonal lymphocyte activation. Developmental and Comparative Immunology 27: 859-874.
- 44. Saeij, J. P., W. B. Muiswinkel, A. Groeneveld, and G. F. Wiegertjes. 2002. Immune modulation by fish kinetoplastid parasites: a role for nitric oxide. 124: 77-86.
- 45. Saeij, J. P., R. J. Stet, A. Groeneveld, L. B. Verburg-van Kemenade, W. B. van Muiswinkel, G. F. Wiegertjes. 2000. Molecular and functional characterization of a fish inducible-type nitric oxide synthase. Immunogenetics 51: 339-346

- 46. Silverman, J. M., S. K. Chan, D. P. Robinson, D. M. Dwyer, D. Nandan, L. J. Foster, and N. E. Reiner. 2008. Proteomic analysis of the secretome of *Leishmania donovani*. Genome Biology **9**: R35.1-R35.21.
- 47. Skeiky, Y. A., D. R. Benson, J. A. Guderian, J. A. Whittle, O. Bacelar, E. M. Carvalho, and S. G. Reed. 1995. Immune responses of leishmaniasis patients to heat shock proteins of Leishmania species and humans. Infection and Immunity 63: 4105-4114
- 48. Sondermann, H., T. Becker, M. Mayhew, F. Wieland, and F. U. Hartl. 2000. Characterization of a receptor for heat shock protein 70 on macrophages and monocytes. Biological Chemistry **381**: 1165-1174.
- 49. Turner, C.M., and J. D. Barry. 1989. High frequency of antigenic variation in *Trypanosoma brucei rhodesiense* infections. Parasitology **99:** 67-75.
- 50. Wang, R., and M. Belosevic. 1994. Cultivation of *Trypanosoma danilewskyi* (Laveran & Mesnil 1904) in serum-free medium and assessment of course of infection in goldfish, *Carassius auratus* (L.). Journal of Fish Disease 17: 47-56
- 51. Woo, P. T. 1981. Acquired immunity against *Trypanosoma danilewskyi* in goldfish, *Carassius auratus*. Parasitology **83**: 343-346
- 52. Zugel, U., and S. H. E. Kaufmann. 1999. Role of heat shock proteins in protection from and pathogenesis of infectious diseases. Clinical Microbiology Review 12: 19-39.

CHAPTER 6

GLYCOPROTEIN 63 (gp63) OF Trypanosoma carassii SUPPRESSES ANTIMICROBIAL RESPONSES OF GOLDFISH (Carassius auratus L.) MACROPHAGES¹

6.0 Introduction

Glycoprotein 63 (gp63) is a metalloprotease belonging to the metizincin class and is characterized by a conserved motif HEXXH, an N-terminal propeptide that maintains the protein in an inactive form that is lost upon activation [44]. The pro-peptide is cleaved from the N-terminal end of nascent gp63 by autocatalytic mechanism that unblocks the active site through cysteine switch mechanism [10, 62]. Mature protease consists of N-terminal domain comprising the catalytic component of the zinc protease, the central domain and C-terminal domain containing the glycosylphosphatidylinositol (GPI) anchor addition site [43, 66]. The protein is synthesized as an N-glycosylated pro-enzyme [41, 44].

Gp63 is also known as major surface protease (MSP) or leishmanolysin and is a major membrane glycoprotein of *Leishmania* spp. [8, 20]. This molecule plays many roles in host-parasite interactions and has been shown to be important for entry of promastigote stage of the parasite into host macrophages and survival inside macrophages [13, 38, 58]. Gp63 is involved in the attachment of the

¹A version of this chapter has been submitted for publication and is under review: Oladiran A and M. Belosevic. Glycoprotein 63 (Gp63) of *Trypanosoma carassii* suppresses antimicrobial responses of goldfish (*Carassius auratus* L.) macrophages.

promastigote to host cell surface receptors, such as the fibronectin receptor [13, 58]. The protease activity of gp63 contributes to the resistance of promastigotes to complement-mediated lysis [9] while immunization with recombinant protein partially protects animals after challenge infection [1, 14, 53, 65]. The findings of a recent report indicate that gp63 is also capable of generating a partial Th1 type protective response in *Leishmania donovani*-infected mice [33]. The immunized animals had high levels of IFN-y and IL-2 and low levels of IL-4 and IL-10 indicating generation of a protective Th1 response and suppression of Th2 type immune response [33]. The Leishmania parasites interfere with host immunity using sophisticated mechanisms involving impairment of host cell signal transduction such as inhibition of JAK/STAT signalling and inactivation of transcription factors AP-1, NF-_kB, and CREB. These activities result in subversion of macrophage anti-microbial functions, including cytokine-induced functions and nitric oxide (NO) production [26, 52]. Gp63 has been demonstrated to be centrally crucial to the processes that lead to impairment of host cell functions. Gp63 is involved in cleavage of NF-_kB, AP-1, protein tryrosine phosphatases (PTP), and actin cytoskeleton regulators [15, 22, 25, 29]

Trypanosome homologues of leishmania gp63 have been identified although the contribution of trypanosome gp63 to host-parasite interactions and immune modulation is yet to be fully elucidated. In *T. brucei*, genes encoding gp63 have been reported and these gene show differential expression in the bloodstream form parasites [18, 36]. This differential expression has led to suggestion that gp63 may protect bloodstream trypanosomes from complement lysis by impairing a protein processing functions [18, 23, 36]. Trypanosome gp63 is also required for removal of variable surface glycoproteins (VSGs) from the surface of bloodstream trypanosomes [23, 36]. *T. cruzi* gp63 genes have been divided into four novel groups based on sequence features [40]. Some *T. cruzi* gp63 (Tcgp63) were differentially expressed in the different life cycle stages and antibodies raised against a Tcgp63 inhibited the entry of trypomastigotes into cells as shown in neutralization assays [16, 35]. These results suggest involvement of Tcgp63 in host-parasite interaction and evasion of phagocytic killing.

T. carassii shares extracellular existence with African trypanosomes although its surface architecture closely resembles T. cruzi [37]. Transmission of T. carassii from infected fish to naive hosts by leech vectors occurs through blood sucking, another characteristic shared with T. brucei. A number of economically important fish can be infected and 100% prevalence of T. carassii infection have been reported [39, 55]. However, fish that survive the infection become immune to homologous challenge [63, 64], suggesting induction of protective immunity and absence of antigenic variation. Although we have identified some parasite molecules that may play role in evasion of goldfish immunity [50, 51), those involved in induction of protective immune responses are yet to be identified. In this study, I cloned and expressed T. carassii gp63 (Tcagp63) to further understand how T. carassii interacts with the fish host and modulates host immunity. My working hypothesis was that Tcagp63 is used by parasite to downregulate inflammatory, antimicrobial and cytokine-induced responses thereby providing T carassii with a strategy to modulate the balance of pro- and antiinflammatory responses to ensure maintenance of long lasting infections and their transmission.

6.1 Experimental design

6.1.1 Molecular cloning and expression of Tcagp63

Total RNA extracted from *in vitro* grown parasites was reverse transcribed into cDNA. The cDNA synthesized was used in polymerase chain reactions (PCR) to amplify Tcagp63 fragment using primers designed to *T. carassii* spliced leader sequence and a highly homologous region of trypanosome gp63 sequences. The full sequence of Tcagp63 was amplified and the recombinant protein was produced using a prokaryotic expression system. Purified rTcagp63 was used to immunize rabbits for production of polyclonal antiserum.

6.1.2 Assessment of the effect of Tcagp63 on nitric oxide response of goldfish macrophages

Primary macrophages cultures obtained from kidney of goldfish (n = 6) were seeded into individual wells of 96-well plates at a density of 3 x 10^5 cells per well. Cells were incubated with culture medium alone (negative control), heatkilled *A. salmonicida* (positive control), rTcagp63 (1 µg/mL, 5 µg/mL or 10 µg/mL) plus heat-killed *A. salmonicida* at 20°C for 48 hours. Nitrite production was determined using the Griess reaction as described previously [24]. To ascertain that the results obtained were not due to other contaminating proteins, *E. coli* bacterial lysates were prepared and subjected to the same purification protocol as rTcagp63-containing lysates. Separate goldfish macrophage cultures obtained from 6 fish (n = 6) were treated with medium alone (negative control), heat-killed *A. salmonicida* (positive control), rTcagp63 (10 μ g/mL) or bacterial lysates (4.5 μ L, equivalent volumes of rTcagp63 lysates that corresponded to 10 μ g/mL of rTcagp63) plus heat-killed *A. salmonicida* and assayed for nitric oxide response using Griess reaction. Also, recombinant goldfish TGF β , purified using the same procedure as described for Tcagp63 were used in the NO assay at 10 μ g/mL.

6.1.3 Assessment of the effect of rTcagp63 on respiratory burst response of goldfish monocytes

Cultures of goldfish monocytes obtained from kidney of goldfish (n = 6) were incubated with medium alone (control), *A. salmonicida* alone or *A. salmonicida* plus different concentration of rTcagp63. In another experiment aimed at understanding how rTcagp63 exerts its effects cells were treated with culture medium alone (control), rTcagp63 (5 µg/mL or 10 µg/mL) for 2 hours and washed 3 times with incomplete medium before used to set up ROI assay. Cells were seeded into 96-well plates at a density of 3×10^5 cells per well. ROI assay was performed with cells incubated in medium alone or with rTcagp63 using heat-killed *A. salmonicida* and recombinant goldfish TNFa2 (rgTNFa2 100 ng/mL) to stimulate production of reactive oxygen species. ROI assays were incubated at 20°C for 48 h. NBT (2 mg/mL, Sigma) and PMA (final conc. 100 ng/mL, Sigma) in PBS were added to the cultures and incubate at room

temperature for an additional 30 minutes. The plates were than centrifuged at 400 x g for 10 minutes, the supernatants aspirated and cells in the pellet fixed with absolute methanol. Non-reduced NBT was removed by washing with methanol and reduced NBT was dissolved with 2 M KOH. DMSO was added to induce the colorimetric response and the plates were read at 630 nm. Absorbances from cells alone (no PMA) were subtracted from treatment values to factor in background NBT reduction.

6.1.4 Assessment of effect of rTcagp63 on immune gene expression of goldfish macrophages activated with *A. salmoncida*

Goldfish monocyte or macrophage cultures (n = 4) were seeded into individual wells of 96-well plates at a density of 1×10^6 cells per well. Cells were treated with medium alone or 10 µg/mL of rTcagp63 at 20°C for 2 hours, washed twice and then stimulated with heat-killed *A. salmonicida* at 20°C for 6 hours. Total RNA was isolated from cells and reverse transcribed into cDNA. Changes in expression pattern of select immune genes (iNOS, TNF α 1 and 2, IL10 and NADPH oxidase component genes) were examined. Relative expression was determined in relation to elongation factor 1 alpha (EF-1 α) and all expression data were normalized using the baseline expression of cells incubated in medium alone. 6.1.5 Assessment of effect of rTcagp63 on immune gene expression of goldfish macrophages

Goldfish macrophage cultures (n = 4) were seeded into individual wells of 96-well plates at a density of 1×10^6 cells per well. Cells were treated with medium alone or 10 µg/mL of rTcagp63 for 0, 3 and 6 hours at 20°C. Total RNA was isolated from the cells and reverse transcribed into cDNA. Changes in expression pattern of select immune genes (iNOS A and B, TNFa1 and 2, IL10) were examined. Relative expression was determined in relation to elongation factor 1 alpha (EF-1a). All expression data were normalized using the baseline expression of cells incubated in medium alone.

6.1.6 Assessment of interaction of rTcagp63 with goldfish macrophages

Goldfish macrophage cultures (n = 3) were established and 10^5 cells were washed twice and incubated with rTcagp63 (10 µg/mL) for 30 minutes at room temperature. Cells were washed twice in PBS and incubated with affinity purified rabbit anti-rTcagp63 (80 µg) or affinity purified rabbit IgG (80 µg) diluted in PBS for 1 hour. This is followed by washing twice in PBS before incubation with FITC-labelled goat anti-rabbit IgG (1:100) diluted in PBS for 1 hour at room temperature. Cells were washed twice in PBS and rTcagp63 binding to macrophages was analyzed using FACS Calibur flow cytometer. A total of at least 10,000 cells were analyzed in each sample.

To assess whether rTcagp63 enter and accumulates in goldfish macrophage, 10^7 cells were washed twice and incubated with rTcagp63 (10 µg/mL) for 0, 1, 2,

3 hours at 20°C. Cells were washed 3 times with PBS and lysed with 30 μ L of RIPA buffer. 30 μ L of Laemmli sample buffer was added and the mixture loaded into SDS PAGE for western blotting. Accumulated rTcagp63 was detected with rabbit anti-rTcagp63 antibody (1:2000) followed by HRP-conjugated goat anti-rabbit IgG (1:3000). Blots were stripped and re-probed rabbit anti-actin-beta (NT) zebrafish polyclonal IgG (AnaSpec Inc.) (1:1000) for detection of goldfish actin, which served as protein loading control.

To assess the effect of rTcagp63 on phospho-tyrosine protein pattern of goldfish macrophages, macrophage cultures (n = 3) were established and 10^7 cells were washed twice and incubated with medium alone (control) or with rTcagp63 (10 µg/mL) for 6 hours. Cells were stimulated with *A. salmonicida* at 20°C for 10, 15 and 20 minutes before lysed with 30 µL of RIPA buffer. 30 µL of Laemmli sample buffer was added and loaded in SDS PAGE for western blotting. Phosphotyrosine protein patterns were detected with anti-phospho-tyrosine antibody (1:4000) and HRP-conjugated goat anti-mouse monoclonal antibody (1:3000).

6.2 Results

6.2.1 Tcagp63 is similar to gp63 of other trypanosomes.

Sequence alignment (Fig. 6.1) shows Tcagp63 shares some sequence similarities with gp63 of *T. cruzi* (Tcgp63) and *T. brucei* (Tbgp63). The highest identity is seen with Tcgp63 46%. Interestingly, the catalytic site of Tcagp63 is different from that of *T. cruzi* and *T. brucei* as well as Leishmania gp63 (Lmg63). The protein structure of Tcagp63 compared with Tcgp63 and Lmgp63 is shown in
Fig. 6.2. Metalloprotease catalytic site denoted by HEXXH is different compared to Tcagp63 catalytic site (HELRTH) which has an insertion of a polar threonine residue with neutral side chains. The implications of this are unclear at this point although introduction of a larger polar residue with neutral side chains may destabilize the structure of this protein at least at the catalytic site. Destabilization of structure may affect the overall functions of the protein or results in a non-functional catalytic site. The glutamic acid residue has been shown to be important for the catalytic activity as demonstrated in mutagenesis experiment [44], while substitution of either histidine residue leads to intracellular degradation of the mutant protein [44]. Phylogenetic analysis of Tcagp63 predicted protein sequence grouped it with trypanosome gp63 sequences and showed the most similarity to *T. cruzi* gp63 (Fig 6.3).

6.2.2 rTcagp63 is unstable and undergoes rapid degradation

To assess the functions of Tcagp63, I expressed the full coding sequence in the prokaryotic system. The protein products of the full coding sequence and protein produced from Tcagp63 fragment that lacks the N-terminal signal peptide were rapidly degraded following purification. I was able to produce relatively stable recombinant Tcagp63 from fragment lacking both the signal peptide and the catalytic site (Fig 6.4 A). The protein product of this fragment was therefore used in antibody production and in subsequent functional assays. Antibody raised against rTcagp63 recognized the recombinant protein (Fig. 6.4 B) but not native gp63 from *T. carassii* lysate. It has been reported that gp63 is posttranslationally modified [10, 43, 44, 62, 66]. Post-translational modification by Nglycosylation has also been described [13] and was thought to stabilize the protein and its enzymatic activity [34]. Three potential N-glycosylation sites have been reported in gp63 proteins and analysis of these sites by site-directed substitution indicated they were important for intracellular stability of the protein [44]. However, functional gp63 proteins have also been produced in prokaryotic system [11, 57]

6.2.3 rTcagp63 abrogated *A. salmonicida*-induced nitric oxide response of goldfish macrophages.

Leishmania spp. parasites use sophisticated mechanisms to subvert macrophage anti-microbial functions including inhibition of nitric oxide (NO) production and macrophage cytokine-inducible functions [52]. These effects have been largely attributed to *Leishmania* spp. gp63 [15, 22]. For this reason, I tested the ability of rTcagp63 to inhibit *A. salmonicida*-inducible macrophage antimicrobial responses. As shown in Figure 6.5, rTcagp63 abrogated *A. salmonicida*-induced NO response of goldfish macrophages. This effect was specific to rTcagp63 since recombinant goldfish TGF β (rgTGF β) and bacteria lysate purified using the same procedure as rTcagp63 had no effect on *A. salmonicida*-induced NO response of macrophages (Fig. 6.6). Cells were viable after addition of recombinant molecules as shown by microscopy, RNA isolation and measurement of metabolic rates by MTT assay. 6.2.4 rTcagp63 abrogated *A. salmonicida*-induced respiratory burst response of goldfish monocytes

To further characterize the effects of Tcagp63 on antimicrobial responses of goldfish, I examined whether incubation of goldfish monocytes with rTcagp63 would affect their ROI response. As shown in Figure 6.7, rTcagp63 down-regulated *A. salmonicida*-induced ROI response of goldfish monocytes. The ability of rTcagp63 to inhibit ROI response was not limited to *A. salmonicida*-activated monocytes. The recombinant protein also inhibited cytokine-inducible effects. As shown in Figure 6.8, rTcagp63 also inhibited rgTNF α 2-indced ROI response of activated goldfish monocytes.

6.2.5 rTcagp63 exerts its effects partly by association with immune cells.

It has been suggested that gp63 could function by cleaving surface receptors on immune cells or by entering cells and activating protein tyrosine phosphatases leading to alteration of signalling and attenuation of antimicrobial responses [22, 30]. I investigated the mechanisms by which rTcagp63 exert its effects. Goldfish monocytes were incubated with rTcagp63 for 2 hours, washed three times and used to set up ROI assay for 48 hours. As shown in Figure 6.9, pre-incubation of cells with rTcagp63 inhibited both *A. salmonicida-* and rgTNF α 2-induced ROI response of the monocytes. These findings suggest that rTcagp63 may use mechanism other than cleavage of surface receptors to exert its effects. However, I cannot exclude the possibility that a combination of receptor cleavage and alteration or attenuation of intracellular signalling occurred after rTcagp63 treatment of monocytes.

6.2.6 Pre-incubation of cells with rTcagp63 altered immune gene expression Since incubation of cells with rTcagp63 abrogates A. salmonicida- and cytokine-induced production of reactive nitrogen and oxygen intermediates, I examined whether pre-incubation of cells with rTcagp63 before incubation with A. salmonicida had an effect on the expression of several immune genes. Pretreatment of cells with rTcagp63 for 2 hours prior to incubation with A. salmonicida for 6 hours caused a significant decrease in expression of iNOS A, TNF α 1 and TNF α 2, compared to that in cells incubated with A. salmonicida alone (Fig. 6.10). Although I observed a decrease in the expressions of IL-10 and p47^{phox}, they were not statistically significant (Fig. 6.11). These results are consistent with previous findings that gp63 interfered with host cell signalling [22, 25, 29]. I then examine whether incubation of cells with rTcagp63 alone had any effect on immune gene expression. As shown in Fig 6.12, only slight increases in expressions of immune genes were observed when cells were incubated with rTcagp63 alone for 6 hours. These results suggest rTcagp63 have no significant effect on immune gene expression at least at the time points investigated in this study.

6.2.7 rTcagp63 associates with goldfish macrophages and altered phospho-tyrosine protein patterns of macrophages.

I examined the ability of rTcagp63 to bind to goldfish macrophages by incubating cells with rTcagp63 (10 μ g/mL) for 30 minutes. Cells were washed twice in PBS and incubated with rabbit anti-rTcagp63 (80 μ g) or rabbit IgG (80 μ g) isotype control for 1 hour. Following incubation, cells were washed twice in PBS and incubated with FITC-labelled goat anti-rabbit IgG (1:100) for 1 hour. Cells were washed twice in PBS and analyzed by flow cytometry. As shown in Figure 6.13, cells incubated with rTcagp63 followed by anti-rTcagp63 showed increased fluorescence intensity indicated by a shift in surface staining compared to cells incubated with isotype control antibody (anti-rabbit IgG). The nonspecific binding of isotype control antibody to macrophages observed (Fig. 6.13) may result from effects of rTcagp63 on the cells. Cleavage of surface molecules by rTcagp63 may expose epitopes recognized by rabbit IgG. Nonetheless, a significant shift in staining suggests binding of rTcagp63 to macrophages.

To further investigate how rTcgp63 interacted with macrophages, I incubated cells with the recombinant protein for 0, 1, 2, 3 hours before washing three times with PBS. Cells were lysed with RIPA buffer and the lysates were subjected to Western blot analysis. rTcagp63 was detected using the antirTcagp63 antibody as early as 1 hour after incubation (Fig. 6.14). The degree of rTcagp63 association with macrophages was time dependent (Fig. 6.14). I used rabbit anti-zebrafish actin to detect goldfish actin that served as loading control to ensure presence of protein in all lanes. To investigate whether rTcagp63 affected the phospho-tryrosine protein patterns of macrophages, cells incubated with either medium alone or with rTcagp63 for 6 hours were washed twice and incubated with *A. salmonicida* for the indicated time. Cells were lysed with RIPA buffer and the lysates were subjected to western blot analysis. I observed changes in phospho-tyrosine protein patterns as early as 10 minutes in cells pre-treated with rTcagp63 (Fig 6.15). I used cells incubated with medium alone as control to determine basal phosphotyrosine protein levels (Fig 6.15).

6.3 Discussion

A number of trypanosome proteins that participate in immune evasion are abundantly expressed on the parasite surface and some have been identified in ES fraction [8, 18, 37, 47, 48, 50, 51, 60]. There is evidence that gp63 a GPIanchored surface protein and can be released from parasites [17, 32, 43, 67], suggesting that it can participate in immune evasion or modulation extracellulary. In my proteomic studies of the excretory/secretory products and surface proteins of *in vitro* grown *T. carassii*, I did not identify gp63 as a constituent of these fractions [50, 51] probably due to very low levels of protein expressed on the surface of *in vitro* grown trypomastigotes. Alternatively, the low level of sequence similarity among parasite gp63 molecules especially at amino acid levels might make detection of Tcagp63 by MS difficult. Moreover, rabbit IgG antibody produced using rTcagp63 failed to detect native protein in Western blot analysis, which may be due to either very low protein levels or post-translational modification of Tcagp63.

Among parasite surface and secreted molecules, the roles of *T. brucei* VSG [59, 60], *T. cruzi* complement regulatory proteins (CRP) [47, 48], Leishmania gp63 and lipohosphoglycan (LPG) [15, 22, 25, 29, 52, 56] are well documented. The rhomboid proteases of *Entamoeba histolytica* cleave surface lectins and have been implicated in immune evasion [5]. *E. histolytica* is an extracellular parasite and must evade host immunity for successful colonization of the host. The parasite uses surface receptor capping whereby surface receptors that have been recognized by the host are shed from the parasite surface [19]. *Plasmodium* rhomboid proteases aid in host cell invasion since they cleave parasite adhesins that engage host cell receptors [3]. Successful invasion of host cells requires disengagement of adhesin-receptor interactions. These results in release of adhesins, and rhomboid proteases that are responsible for this activity have been identified in all invasive stages of the malaria life cycle [3].

Crithidia fasciculata a trypanosomatid that does not seem to have mammalian host and complete its life cycle in mosquito has been reported to possess a homologue of *Leishmania* gp63 [31]. *C. fasciculata* appears to have two developmental forms in the insect but the biological role of gp63 in this organism remains to be determined [31, 42]. It is thought that *C. fasciculata* gp63 might protect the organism from insect immune response [31]. This suggests evolutionary conservation of gp63 metalloprotease [18].

200

Replacement of the bloodstream stage VSG with procyclic stage procyclin, is a hallmark of differentiation in T. brucei. VSG is released during differentiation by at least two mechanisms, endoproteolytic cleavage by gp63 and GPI hydrolysis by endogenous bloodstream trypanosome glycosylphosphatidylinositol-specific phospholipase C (GPI-PLC) which hydrolyzes GPI anchor present on VSG [27, 28]. Electrophoretic mobility of the VSG released in culture medium can be used to distinguish between the two forms. Full-length soluble VSG results from GPI hydrolysis while truncated VSG is due to proteolysis [27]. Several genes encoding gp63 have been identified in trypanosomes, but the contribution(s) of trypanosome gp63 to immune modulation is yet to be fully investigated. T. brucei gp63 molecules show differential expression in the bloodstream form of the parasites, and are required for removing VSG from the surface of bloodstream trypanosomes [18, 23, 27, 28, 36]. It has been suggested that gp63 is up-regulated during differentiation from bloodstream parasites to procyclic forms to release VSG and is maintained on the surface of procyclic trypanosomes following differentiation [28]. T. cruzi gp63 molecules also show differential expression in the different life cycle stages and have been reported to be involved in infection of host cells [16, 35]. Although T. *carassii* is an extracellular parasite of ectothermic vertebrates, it has a surface coat that is rich in sialic acid residues like T. cruzi that is primarily an intracellular parasite [2]. It is clear that Tcagp63 is not involved in infection of host cells or the cleavage of VSG molecules, since T. carassii does not have VSGs on its surface [2, 54].

I showed that rTcagp63 associates with goldfish macrophages and that cells treated with the recombinant protein exhibited altered phospho-tyrosine protein patterns. These findings are similar to those that showed that the incubation of macrophages with *L. major* or recombinant gp63 from *L. major* lead to internalisation of native and recombinant proteins within 1 hour of incubation [22]. Internalised gp63 interacted with and activated protein tyrosine phosphatases causing down-regulation of JAK/STAT and MAPK signalling and inhibition of macrophage functions [7, 45, 21].

rTcagp63 abrogated A. salmonicida- and rgTNFα2-induced production of reactive oxygen and nitrogen intermediates of goldfish monocytes and macrophages, respectively. These effects may result from alteration in signalling events. Attenuation of signalling or alteration in signalling events will affect activity of transcription factors and consequently gene transcription. I observed significant decrease in the expression of iNOS isoform A that encodes the enzyme (iNOS) responsible for the production of reactive nitrogen intermediates. In addition, the mRNA levels of TNF α 1 and TNF α 2 in macrophages pre-incubated with rTcagp63, and then activated by A. salmonicida were significantly reduced. These observations are consistent with previous finding that *Leishmania* gp63 altered host cell signalling through cleavage of protein tyrosine phosphatase [22] and that it mediated the inactivation of p38 mitogen-activated protein kinase [29] and transcription factors [15]. Since incubation of macrophages with the recombinant protein followed by washing to remove the protease before used for ROI assay failed to reverse the effects of rTcagp63, rTcagp63 likely mediated

these effects by associating with macrophages and interfering with signal transduction. This undoubtedly leads to attenuation or alteration of signalling and inhibition of the activity of transcription factors and eventual impairment of downstream macrophage antimicrobial functions.

The recombinant Tcagp63 used in this study lacks the metalloprotease catalytic site. However, Tcagp63 catalytic site denoted by HELRTH is different from the putative metalloprotease catalytic site denoted by HEXXH. Although Tcagp63 catalytic site contains the important residues required for catalytic activity and stabilization of the protein (Section 6.2.1), insertion of a larger polar threonine residue with neutral side chains may destabilize the structure of Tcagp63 catalytic site. It is possible that Tcagp63 catalytic site is inactive or that this protein retained other functions that are independent of the catalytic site.

This is the first report showing that a trypanosome gp63 down-regulates pro-inflammatory responses of macrophages. I showed previously that a balanced Th1/Th2 response is important for the outcome of *T. carassii* infection in the goldfish [49], and that one of the immune evasion strategies of the parasite is the secretion of hsp70 that causes up-regulation of macrophage antimicrobial responses [51]. The results reported here indicate *T. carassii* gp63 may be one of the molecules responsible for dampening pro-inflammatory responses induced by Tcahp70 and related molecules. This hypothesis is supported by observations that both Th1 and Th2-type cytokine genes were up-regulated in *T. carassii*-infected goldfish compared to non-infected controls [49]. Taken together, my findings suggest that *T. carassii* has the ability to regulate and balance pro- and anti-

inflammatory responses of the host during infection enabling it to evade immune responses and persist in its ectothermic host. However, these activities are likely to result in inability of the fish host to mount a strong humoral response to eliminate the parasites thereby leading to generalised immunosuppression. These findings suggest that induction of immunosuppression may be an important immune evasion and modulation strategy of *T. carassii*. Immunosuppression is common in mammalian trypanosome infections [4, 61] and untimely death of host may occur due to secondary infections.

-MOUTMI ---T.cruzi -LIVILLOCV9 16 MLTTHFRCCISPRVSGAYSLFPLFLPCIKRKRLMMLPACVIPMHGALKLAILLMLVWCCS 60 T.brucei -----TAVLLLVLWST 17 ------IAVAFALVCCVV 19 T.carassii L.major . . GSVAVAEH-----HCISDEIEKKVG-SRTTAVVLELPTRESGMMRALTASAPEWAP 66 T.cruzi LCLAKSGD-----RCMFDEIAAKAGRPRVLALRRTKAGMENVKYDRTGSVDPEWOH 111 T.brucei T.carassii ECSALLEY-----RCIHDKITREWDKKNSTAYILPLTSGDQ---KGADAAGDAFAP L.major GPGAAOGHPERADSEEPRCGFDELEAHTIGTRVSGISR--VELPTGELVVAAAATGALOP 77 :* *:: . . . T.cruzi VRFQLFTEDLNDPSRYCTAEGQIRPDFTG-GTVECKERDILTEEKKSIILNSLIPRALKM 125 T.brucei IRIVVFAEDMKDRSRYCTSAGOERPTFFG-ETATCSOEDILTAAKRDIAVTKLLPSAVOM 170 T.carassii IRIKVFSEDINNASRYCTSEGQSRPTFKDTSLATCSSHDILTSAKKDILLNYLIPSAIQL IRIAVFTDDISNSSQHCTASCQSRPNFRG-SRVTCSAAEVLTRAKKRVLLELLIPSAVQL :*::*::*::*::*::*:*******...**:******...** L.major 136 * . *: : : T.cruzi HTDRLLVEPLMGRVIVP--EFLSGACAQFTIPSSHHTEGVFGADMYLYVSAAPIKGSTLA 183 HMDRLLVDPITEPLVFP--PFDGSVCSEFKVPSSHFSEGVPDADMVMYAAAGPTPEGVAA 228 HAERLLVVPLQGAVKITRDILVGNPCSLFSIPPEHFTTGATDGDMFIYAAAGPTNLPEVA 185 T.brucei T.carassii L.major HQERLNVQRVNGNIVVDSSIQKDRVCGQFSIREEHMKTGVKDADFVLYMSAAPTSGSVIA 196 *. *.: .* . *. ..*: :* :*.* :** : ÷ . . WAVACSALPD-GRPVVGVVNYGPRSVTDSEHSVRALVHEIAHALGFTLEIMEERNMLKEF 242 T.cruzi T.brucei WATGCITLDD-GRAVAGVTNLGPGSISLSETSIRTAAHEIAHILGFNFRAMNDAGMVQRI 287 T.carassii WIFRCFFSEIYERPLVGAMNISPHHASRGHFYIRTVAHELRTHLDSLIVYLYVWNSRDQL 245 WALKCONFDN-GRPSVGVVNISPKYIAADPKTVRVIAHEVLHALGFSRSVFQERNMLAMA 255 L.major *. .*. * .* :*. .**: ÷ . *. : T.cruzi LNVRGKASVLOVSSPKTVEKTREHFNCVTATGMELEDEGGERTASSHWKRRNAKDELMAG 302 T.cruzi T.brucei T.carassii PGVRGKVDVTLISSPRTLQKAREHYNCPDAPGMELEDEGGSGTALSHWERRNAKDEIMSG 347 S-IRGKSYISYVTSPMTRERAOOHYNCSTAIGMELEDEGGTGTALSHWKRRNAKDELMAG 304 L.major S-FRSKGPSPVICSEKVVAKAQQHYGCKTQAFMELEDTGDIDDASSHWKRRNAKDELMAG . **..* **** * * ***:******:*:* .*.* T.cruzi LSGIGYYTALTMAALEDTGFYKANWGMEEPMSWGNNSGCALLTEKCVMNGVTKYPEMFCT 362 T.cruzi T.brucei T.carassii ISSPGRYTALTMAAFEDLGYYRGAWGSEEPMGWGNNSGCELLNESCLVNGVTAHPDMFCN 407 ITGAGYYTALTMAAFEDLDFYRANWGKEEVMEWGRDASCDFLTEKCITKGVTAHPDMFCT 364 FSGVGIYSALTIAAMEDTGYYQGNYAKAEPMAYGHDAGCKLSSDQCVTNSTSQIPGMFCD 374 L.major ... * *:***:**:** . : * : . * * :*.:..* : ...*: :... T.cruzi ${\tt AESRLLSCTSDRLALGYCALKLYDAPLPPQFQYFSNPKLGGSPDLLMDFCPYIEEYANAG}$ 422 T.cruzi T.brucei T.carassii ETVSKLVCNSERDGLGRCNVIKHENPLPPQYHYFSDPSRGAPSHLLMDYCPSIDAFSNTP 467 EKLSTLTCTSDRQNLGVCNIKKYDSILPEEFQYFKDPSVGTEHDVLMDFCPYITPRSDTG 424 APDAPWSCTSDRLGVGRCILTSHKSNLPTYF0YFSDPRLGGP-DPLMDFCPVVEVAEGTM L.major 433 *.*:* :* * : :. ** ::**.:* * . ***:** : CSDGNATDMRGSRVGPTSKCLK---GDGLADFMGFIGDVCAEVSCDKGEVSVRYLGDDAW 479 T.cruzi T.brucei T.carassii L.major CADGETKEMRGSLIGPSSMCLK---AEGLRDSOGVIGDVCADVRCDGGEVSIRYLGDDAW 524 CMDGRLKHMPGSVVGPAARCVK---GSSLVYYGKPIGDVCVHVRCDNETLFISHYGTQEW L.major CA-ATTNALKGSVYGVMSRCVDTPVGFSMDDSAVROHGICVEVQCDSTKYYIKANGASAF 492 . : ** : *:. . .: . : * . . * :

 T.cruzi
 HKCPEGSSITPTGL---FMKGRILCPKYDDVCIVIDTING-----TGDVSSLLSAFPP
 529

 T.brucei
 HPCPEGSHIKPTTT---FTDGVIVCPTYSEVCIKATVVVR-----PSSASYRSSVPQS
 574

 T.carassii
 IECPAGSGITPSAP--FSGGQIVCPQYHEVCIKATVVVR-----PSSASYRSSVPQS
 574

 L.major
 CCCPPGSTYNLSTLSPSFSKGYLVCPSYESVCAIKINASLYEEYSGFLTDHSVAGVLTSV
 552

 ** **
 * * ::** * **
 **

IPLIILVLIFISMF 543 LLLTLFAIVYAAC- 587 T.cruzi T.brucei T.carassii SVVSIVFFVFL--L.major KAVVAVLLVVLFMV 566 . . .

Figure 6.1. Amino acid alignment of *T. carassii* gp63 (Tcagp63) sequence and those from *T. cruzi*, *T. brucei* and *Leishmania major*. Sequences were obtained from GenBank accession numbers *T. cruzi* (XP_817187), *T. brucei* (XP_846998), *L. major* (XP_001684335). (*) indicates identical residues, (:) indicates conserved residues while (.) indicates semi-conserved residues. (-) are gaps introduced to maximize the alignment. The predicted cleavage site for Tcagp63 signal peptide and pro-region respectively are indicated with arrow. A bar denotes the putative metalloprotease zinc binding site.



Figure 6.2. Schematic of protein structure of *T. carassii* gp63. *T. carassii* gp63 was compared with *T. cruzi* and *L. major* CRT sequences. The coloured boxes represent the putative gp63 zinc metalloprotease catalytic site. The approximate locations of the catalytic sites are shown.



Figure 6.3. Phylogenetic analysis of the *T. carassii* gp63 demonstrated closest relationship to *T. cruzi* and *T. brucei* gp63 sequences. Protein sequences used and their accession numbers are: *T. cruzi* (XP_817187), *T. brucei* (XP_846998), *T. congolense* (CCC95446), *T. vivax* (CCC53361), *L. major* (XP_001684335), *L. mexicana* (CAA45733), *L. donovani donovani* (ACT31401), *L. infantum* (XP_003392667), *L. braziliensis* (XP_001562828), *T. carsssii* Cathepsin L (ABQ23397), *T. cruzi* cruzipain (AAF75547), *T. brucei* cysteine peptidase (XP_845218). *T. carsssii* Cathepsin L, *T. cruzi* cysteine protease and *T. brucei* cysteine peptidase were used as out-groups.



Figure 6.4. Western blot analysis of rTcagp63. Western blot analysis of rTcagp63. rTcagp63 was expressed in prokaryotic system and purified as described. rTcagp63 is recognized by Mouse anti-histidine monoclonal antibody (1:5000) (A) and Rabbit anti-rTcagp63 affinity purified IgG polyclonal antibody (1:2000) raised against rTcagp63 (B).











Figure 6.7. Recombinant *T. carassii* gp63 abrogated *A. salmonicida*-induced ROI response of goldfish monocytes. ROI (mean \pm SEM) assay was performed with cells from 6 individual fish (n = 6) incubated in medium alone or with rTcagp63 using heat-killed *Aeromonas salmonicida* to stimulate production of reactive oxygen species. ROI assays were incubated at 20°C for 72 hours. NBT (2 mg/mL, Sigma) and PMA (final conc. 100 ng/mL, Sigma) in PBS was added to the cultures and incubated at room temp for an additional 30 minutes. The supernatants were aspirated and cells in the pellet fixed with absolute methanol. Non-reduced NBT was removed by washing with methanol and reduced NBT was dissolved with 2 M KOH. DMSO was added to induce the colorimetric response and the plates were read at 630 nm. Statistical analysis was done using one-way ANOVA. (*) denotes statistically different (P < 0.05) from medium control. (+) denotes statistical difference (P < 0.05) between experimental groups.



Figure 6.8. Recombinant *T. carassii* gp63 abrogated rgTNFa2-induced ROI response of goldfish monocytes. ROI (mean \pm SEM) assay was performed with cells from 6 individual fish (n = 6) incubated in medium alone or with rTcagp63 using rgTNFa2 (100 ng/mL) to stimulate production of reactive oxygen species. ROI assays were incubated at 20°C for 72 hours. NBT (2 mg/mL, Sigma) and PMA (final conc. 100 ng/mL, Sigma) in PBS was added to the cultures and incubated at room temp for an additional 30 minutes. The supernatants were aspirated and cells in the pellet fixed with absolute methanol. Non-reduced NBT was removed by washing with methanol and reduced NBT was dissolved with 2 M KOH. DMSO was added to induce the colorimetric response and the plates were read at 630 nm. Statistical analysis was done using one-way ANOVA. (*) denotes statistically different (P < 0.05) from medium control. (+) denotes statistical difference (P < 0.05) between experimental groups.



Figure 6.9. Pre-incubation of cells with rTcagp63 and subsequent washing failed to reverse A. salmonicida and $TNF\alpha2$ -induced ROI response of goldfish monocytes. ROI (mean± SEM) assay was set up with cells from 6 individual fish (n = 6) incubated in medium alone or with rTcagp63 (5 µg/mL and 10 µg/mL) for 2 hours. Cells were washed to remove the protease and used to set up ROI assay. Cells were stimulated with heat-killed *Aeromonas salmonicida* or rgTNF α 2 (100 ng/mL) for production of reactive oxygen species. ROI assays were incubated at 20°C for 72 hours. NBT (2 mg/mL, Sigma) and PMA (final conc. 100 ng/mL, Sigma) in PBS was added to the cultures and incubated at room temp for an additional 30 minutes. The supernatants were aspirated and cells in the pellet fixed with absolute methanol. Non-reduced NBT was removed by washing with methanol and reduced NBT was dissolved with 2 M KOH. DMSO was added to induce the colorimetric response and the plates were read at 630 nm. Statistical analysis was done using one-way ANOVA. (*) denotes statistically different (P < 0.05) from medium control. (+) denotes statistical difference (P < 0.05) between experimental groups.















Figure 6.13. Recombinant T. carassii gp63 was bound to goldfish

macrophages. Goldfish macrophage cultures obtained from two individual fish (n = 2) were used in all assays. 10^5 cells were washed twice and incubated with rTcagp63 (10 µg/mL) for 30 minutes. Cells were washed twice in PBS and incubated with affinity purified rabbit anti-rTcagp63 IgG (80 µg) or affinity purified rabbit IgG (80 µg) in PBS for 1 hour. Following incubation, cells were washed twice in PBS and incubated with FITC-labelled goat anti-rabbit IgG (1:100) for 1 hour. Cells were washed twice and analyzed by flow cytometry. A total of at least 10,000 cells were analyzed in each sample. 1 & 2 indicate results from separate experiments.



Figure 6.14. Recombinant T. carassii gp63 associates with goldfish

macrophages. 10^7 cells were washed twice and incubated with rTcagp63 (10 µg/mL) for 0, 1, 2, 3 hours at 20°C. Cells were washed 3 times with PBS and lysed with 30 µL of RIPA buffer. rTcagp63 in lysate was detected with rabbit anti-rTcagp63 antibody (1:2000) and HRP-conjugated goat anti-rabbit IgG (1:3000). Blots were stripped as described and reprobed with rabbit anti-zebrafish actin polyclonal antibody (1:1000) and HRP-conjugated goat anti-rabbit IgG (1:3000) for detection of goldfish actin which was used as loading control.



Figure 6.15. Recombinant *T. carassii* gp63 altered phospho-tyrosine protein pattern of goldfish macrophages. 10^7 cells were washed twice and incubated with medium alone (control) or with rTcagp63 (10 µg/mL) for 6 hours. Cells were stimulated with *Aeromonas salmonicida* at 20°C for 10, 15 and 20 minutes before lysed with RIPA buffer. Phospho-tyrosine protein patterns were analysed by western blotting with mouse anti-phosphotyrosine antibody (1:4000) and HRPconjugated goat anti-mouse monoclonal antibody (1:3000). Results are representative of three individual fish.

6.4 References

- Abdelhak, S., H. Louzir, J. Timm, L. Blel, Z. Benlasfar, M. Lagranderie, M. Gheorghiu, K. Dellagi, and B. Gicquel. 1995. Recombinant BCG expressing the *Leishmania* surface antigen Gp63 induces protective immunity against Leishmania major infection in BALB/c mice. Microbiology 141: 1585-1592.
- Agüero, F., V. Campo, L. Cremona, A. Jäger, J. M. Di Noia, P. D. Overath, D. O. Sánchez, and A. C. Frasch. 2002. Gene discovery in the freshwater fish parasite *Trypanosoma carassii*: identification of transialidase-like and mucin-like genes. Infection and Immunity **70**: 7140-7144
- Baker, R. P., R. Wijetilaka, and S. Urban. 2006. Two *Plasmodium* rhomboid proteases preferentially cleave different adhesions implicated in all invasive stages of malaria. PLos Pathogen 2: e113.
- Barrett, M. P., R. J. Burchmore, A. Stich, J. O. Lazzari, A. C. Frasch, J. J. Cazzulo, and S. Krishna. 2003. The trypanosomiases. Lancet 362: 1469-1480.
- Baxt, L.A., R. P. Baker, U. Singh, S.Urban. 2008. An *Entamoeba histolytica* rhomboid protease with atypical specificity cleaves a surface lectin involved in phagocytosis and immune evasion. Genes and Development 22: 1636-1646.
- Bienek, D. R., D. A. Plouffe, G. F. Wiegertjes, and M. Belosevic. 2002. Immunization of goldfish with excretory/secretory molecules of *Trypanosoma danilewskyi* confers protection against infection. Developmental and Comparative Immunology 26: 649–657
- Blanchette, J., N. Racette, R. Faure, K. A. Siminovitch, and M. Olivier. 1999. *Leishmania*-induced increases in activation of macrophage SHP-1 tryrosine phosphatase are associated with impaired IFN-γ-triggered JAK2 activation. European Journal of Immunology **29**: 3737-3744.
- Bouvier, J., R. J. Etges, C. Bordier. 1985. Identification and purification of membrane and soluble forms of the major surface protein of *Leishmania* promastigotes. Journal of Biological Chemistry 260: 15504-15509.

- Brittingham, A., C. J. Morrison, W. R. McMaster, B. S. McGwire, K. P. Chang, and D. M. Mosser. 1995. Role of the *Leishmania* surface protease gp63 in complement fixation, cell adhesion, and resistance to complementmediated lysis. Journal of Immunology **155**: 3102-3111.
- Button, L.L., G. Wilson, C. R. Astell, and W. R. McMaster. 1993. Recombinant *Leishmania* surface glycoprotein GP63 is secreted in the baculovirus expression system as a latent metalloproteinase. Gene 134: 75-81.
- 11. Button, L.L., N. E. Reiner, and W. R. McMaster. 1991. Modification of GP63 genes from diverse species of *Leishmania* for expression of recombinant protein at high levels in *Escerichia coli*. Molecular and Biochemical Parasitology 44: 213-224.
- Chang, K.P., S. G. Reed, B. S. McGwire, and L. Soong. 2003. *Leishmania* model for microbial virulence: the relevance of parasite multiplication and pathoantigenicity. Acta Tropica 85: 375-390.
- Chang, C. S., and K-P. Chang. 1986. Monoclonal antibody affinity purification of a *Leishmania* membrane glycoprotein and its inhibition of leishmania-macrophage binding. Proceedings of National Academy of Science USA 83: 100-104.
- Connell, N. D., E. Medina-Acosta, W. R. McMaster, B. R. Bloom, and D. G. Russell. 1993. Effective immunization against cutaneous leishmaniasis with recombinant bacille Calmette-Guerin expressing the Leishmania surface proteinase gp63. Proceedings of National Academy of Science USA 90: 11473-11477
- Contreras, I., M. A. Gomez, O. Nguyen, M. T. Shio, and R. W. McMaster.
 2010. *Leishmania*-induced inactivation of the macrophage transcription factor AP-1 is mediated by the parasite metalloprotease GP63. PLos Pathogen 6: e1001148.
- 16. Cuevas, I. C., J. J. Cazzulo, and D. O. Sanchez. 2003. gp63 homologues in *Trypanosoma cruzi*: surface antigens with metalloprotease activity and a possible role in host cell infection. Infection and Immunity **71**: 5739-5749.

- 17. Elias, C. G. R., F. M. Pereira, B. A. Silva, C. S. Alviano, R. M. A. Soares, and A. L. S. Santos. 2006. Leishmanolysin (gp63 metallopeptidase)-like activity extracellulary released by *Herpetomonas samuelpessoai* Parasitology **132**: 37-47.
- El-Sayed, N. M., and J. E. Donelson. 1997. African trypanosomes have differentially expressed genes encoding homologues of the *Leishmania* GP63 surface protease. Journal of Biological Chemistry 272: 26742-26748.
- Espinosa-Cantellano, M., and A. Martinez-Palomo. 1994. *Entamoeba histolytica:* Mechanism of surface receptor capping. Experimental Parasitology **79:** 424-435
- 20. Etges, R., J. Bouvier, and C. Bordier. 1986. The major surface protein of *Leishmania* promastigotes is anchored in the membrane by myristic acid labelled phosphoplipid. EMBO Journal 5: 597-601.
- Forget, G., D. J. Gregory, L.A. Whitcombe, and M. Olivier. 2006. Role of host protein tyrosine phosphatase SHP-1 in *Leishmania donovani*-induced inhibition of nitric oxide production. Infection and Immunity **74:** 6272-6279.
- 22. Gomez, M. A., I. Contreras, M. Halle, M. L. Tremblay, R. W. McMaster, and M. Olivier. 2009. *Leishmania* GP63 alters host signalling through cleavage-activated protein tryrosine phosphatases. Science Signal **2:** ra58.
- Grandgenett, P. M., K. Otsu, H. R. Wilson, M. E. Wilson, J. E. Donelson.
 2007. A function for a specific zinc metalloprotease of African trypanosomes. PLoS Pathogen 3: 1432-1445
- 24. Grayfer, L., and M. Belosevic. 2009. Molecular characterization, expression and functional analysis of goldfish (*Carassius aurutus* L.) interferon gamma. Developmental and Comparative Immunology 33: 235–246.
- 25. Gregory, D. J., I. Contreras, G. Forget, and M. Olivier. 2008. A novel form of NF-_kB is induced by *Leishmania* infection: Involvement in

macrophage gene expression. European Journal of Immunology **38:** 1071-1081

- 26. Gregory, D. J., and M. Olivier. 2005. Subversion of host cell signalling by the protozoan parasite *Leishmania*. Parasitology **130**: S27-S35
- 27. Gruszynski, A. E., F. J. van Deursen, M. C. Albareda, A. Best, K. Chaudhary, L. J. Cliffe, et al. 2006. Regulation of surface coat exchange by differentiating African trypanosomes. Molecular and Biochemical Parasitology 147: 211-223.
- Gruszynski, A. E., A. DeMaster, N. M. Hooper, and J. D. Bangs. 2003.
 Surface coat remodelling during differentiation of *Trypanosoma brucei*. Journal of Biological Chemistry 278: 24665-24672
- 29. Halle, M., M. A. Gomez, M. Stuible, Shimizu, W. R. McMaster, M. Olivier, and M. L. Tremblay. 2009. The *Leishmania* surface protease GP63 cleaves multiple intracellular proteins and actively participates in p38 mitogen-activated protein kinase inactivation. Journal of Biological Chemistry **284:** 6893-6908.
- 30. Hey, A. S., T. G. Theander, L. Hviid, S. M. Hazrati, M. Kemp, and A. Kharazmi. 1994. The major surface glycoprotein (gp63) from *Leishmania major* and *Leishmania donovani* cleaves CD4 molecules on Human T cells. Journal of Immunology 152: 4542-4558
- 31. Inverso, J. A., E. Medina-Acosta, J. O'Connor, D. G. Russell, and G. A Cross. 1993. *Crithidia fasciculata* contains a transcribed leishmania surface proteinase (gp63) gene homologue. Molecular and Biochemical Parasitology 57: 47-54
- Jaffe, C. L., and D. M. Dwyer. 2003. Extracellular release of the surface metalloprotease gp63, from Leishmania and insect trypanosomatids. Parasitology Research 91: 229–237.
- 33. Kaur, T., Sobti, R. C., and S. Kaur. 2011. Cocktail of gp63 and Hsp70 induces protection against *Leishmania donovani* in BALB/c mice. Parasite Immunology 33: 95-103.

- 34. Kink, J. A., and K-P. Chang. 1988. N-glycosylation as a biochemical basis for virulence in *Leishmania mexicana amazonensis*. Molecular and Biochemical Parasitology 27: 181-190
- 35. Kulkarni, M. M, C. L. Oloson, D. M. Engman, and B. S. McGwire. 2009. *Trypanosoma cruzi* gp63 proteins undergo stage-specific differential posttranslational modification and are important for host cell infection. Infection and Immunity 77: 2193–2200
- LaCount, D. J., A. E. Gruszynski, P. M. Grandgenett, J. D. Bangs, and J. E. Donelson. 2003. Expression and function of the *Trypanosoma brucei* major surface protease (GP63) genes. Journal of Biological Chemistry 278: 24658-24664.
- 37. Lischke, A., C. Klein, Y. D. Stierhof, M. A. Hempel, A. Mehlert, I. C. Almeida, M. A. Ferguson, P. Overath. 2000. Isolation and characterization of glycosylphosphatidylinositol-anchored, mucin-like surface glycoproteins from bloodstream forms of the freshwater-fish parasite *Trypanosoma carassii*. Biochemistry Journal **345**: 693-700.
- 38. Liu, X., and K. P. Chang. 1992. Extrachromosomal genetic complementation of surface metalloproteinase (gp63)-deficient *Leishmania* increases their binding to macrophages. Proceedings of National Academy of Science USA 89: 4991-4995.
- 39. Lom, J., and I. Dykova. 1992. Protozoan parasites of fishes. Developments in aquaculture and fisheries science. Amsterdam: Elsevier 1992.
- 40. Ma, L., K. Chen, Q. Meng, Q. Liu, P. Tang, S. Hu, and J. Yu. 2011. An evolutionary analysis of trypanosomatid Gp63 proteases. Parasitology Research DOI: 10.1007/s00436-011-2348-x
- 41. Macdonald, M. H., C. J. Morrison, and W. R. McMaster. 1995. Analysis of the active site and activation mechanism of the *Leishmania* surface metalloproteinase gp63. Biochimica and Biophysica Acta 1253: 199-207.
- Medina-Acosta, E., S. M. Beverley, and D. G. Russell. 1993. Evolution and expression of the Leishmania surface proteinase (gp63) gene locus. Infectious Agents and Disease 2: 25-34.

- 43. McGwire, B. S., W. A. O'Connell. K. P. Chang, and D. M. Engman. 2002. Extracellular release of the glycosylphosphatidylinositol (GPI)-linked Leishmania surface metalloprotease, gp63, is independent of GPI phospholipolysis: implications for parasite virulence. Journal of Biological Chemistry 277: 8802-8809.
- 44. McGwire, B. S., and K. P. Chang. 1996. Posttranslational regulation of a *Leishmania* HEXXH metalloprotease (gp63). The effects of site-specific mutagenesis of catalytic, zinc binding, N-glycosylation, and glycosyl phosphatidylinositol addition sites on N-terminal end cleavage, intracellular stability, and extracellular exit. Journal of Biological Chemistry **271:** 7903-7909.
- 45. Nandan, D., and N. E. Reiner. 2005. *Leishmania donovani* engages in regulatory interference by targeting macrophage protein tyrosine phosphatase SHP-1. Clinical Immunology **114**: 266-277.
- 46. Neumann, N. F., D. R. Barreda, and M. Belosevic. 2000. Generation and functional analysis of distinct macrophage sub-populations from goldfish (*Carassius auratus* L.) kidney leukocyte cultures. Fish Shellfish Immunology **10**: 1-20.
- 47. Norris, K. A. 1998. Stable transfection of *Trypanosoma cruzi* epimastigotes with the trypomastigote-specific complement regulatory protein cDNA confers complement resistance. Infection and Immunity 66: 2460-2465.
- 48. Norris, K. A., J. E. Schrimpf, and M. J. Szabo. 1997. Identification of the gene family encoding the 160-kilodalton *Trypanosoma cruzi* complement regulatory protein. Infection and Immunity 65: 349-357
- 49. Oladiran, A., D. Beauparlant, and M. Belosevic. 2011. The expression analysis of inflammatory and antimicrobial genes in the goldfish (*Carassius auratus L.*) infected with *Trypanosoma carassii*. Fish Shellfish Immunology **31**: 606-613.
- 50. Oladiran, A., and M. Belosevic. 2010. *Trypanosoma carassii* calreticulin binds host complement component C1q and inhibits classical complement

pathway-mediated lysis. Developmental and Comparative Immunology **34:** 396-405.

- 51. Oladiran, A., and M. Belosevic. 2009. *Trypanosoma carassii* hsp70 increases expression of inflammatory cytokines and chemokines in macrophages of the goldfish (*Carassius auratus* L.). Developmental and Comparative Immunology **33**: 1128–1136.
- 52. Olivier, M., D. J. Gregory, and G. Forget. 2005. Subversion mechanisms by which *Leishmania* parasites can escape the host immune response: a signalling point of view. Clinical Microbiology Review 18: 293-305.
- 53. Olobo, J. O., C. O. Anjili, M. M. Gicheru, P. A. Mbati, T. M. Kariuki, J. I. Githure, D. K. Koech, and W. R. McMaster. 1995. Vaccination of vervet monkeys against cutaneous leishmaniosis using recombinant *Leishmania* 'major surface glycoprotein' (gp63). Veterinary Parasitology **60**: 199-212.
- 54. Overath, P., J. Haag, M. G. Mameza, and A. Lischke. 1999. Freshwater fish trypanosomes: definition of two types, host control by antibodies and lack of antigenic variation. Parasitology **119**: 591–601.
- 55. Overath, P., J. Ruoff, Y. D. Stierhof, J. Haag, H. Tichy, I. Dykova, J. Lom. 1998. Cultivation of bloodstream forms of *Trypanosoma carassii*, a common parasite of freshwater fish. Parasitology Research 84: 343–347.
- 56. Puentes, S. M., D. M. Dwyer, P. A. Bates, K. A. Joiner. 1989. Binding and release of C3 from *Leishmania donovani* promastigotes during incubation in normal human serum. Journal of Immunology **143**: 3743-3749
- 57. Sachdeva, R., A. C. Banerjea, N. Malla, and M. L. Dubey. 2009. Immunogenicity and efficacy of single antigen Gp63, polytope and polytopeHSP70 DNA vaccines against visceral leishmaniasis in experimental mouse model. PLos One 4: e7880.
- 58. Soteriadou, K. P., M. S. Remoundos, M. C. Katsikas, A. K. Tzinia, V. Tsikaris, C. Sakarellos, and J. Tzartos. 1992. The Ser-Arg-Tyr-Asp region of the major surface glycoprotein of *Leishmania* mimics the Arg-Gly-Asp-Ser cell attachment region of fibronectin. Journal of Biological Chemistry 267: 13980-13985

- 59. Taylor, J. E., and G. Rudenko. 2006. Switching trypanosome coats: what's in the wardrobe? Trends in Genetics **22:** 614-620
- 60. Turner, C.M., and J. D. Barry. 1989. High frequency of antigenic variation in *Trypanosoma brucei rhodesiense* infections. Parasitology **99:** 67-75.
- 61. Uzonna, J. E., R. S. Kaushik, Y. Zhang, J. R. Gordon, and H. Tabel. 1998. Experimental murine *Trypanosoma congolense* infections. II. Role of splenic adherent CD3+Thy1.2+ TCR-alpha beta- gamma delta- CD4+8and CD3+Thy1.2+ TCR-alpha beta- gamma delta- CD4-8- cells in the production of IL-4, IL-10, and IFN-gamma and in trypanosome-elicited immunosuppression. Journal of Immunology **161:** 6189-6197.
- 62. Van Wart, H. E., and H. Birkedal-Hansen. 1990. The cysteine switch: a principle regulation of metalloproteinase activity with potential applicability to the entire matrix metalloproteinase gene family. Proceedings of National Academy of Science USA 87: 5578-5582.
- 63. Wang, R., and M. Belosevic. 1994. Cultivation of *Trypanosoma* danilewskyi (Laveran & Mesnil 1904) in serum-free medium and assessment of course of infection in goldfish, *Carassius auratus (L.)*. Journal of Fish Disease 17: 47–56.
- 64. Woo, P.T., 1981. Acquired immunity against *Trypanosoma danilewskyi* in goldfish, *Carassius auratus*. Parasitology **83:** 343–346.
- 65. Xu, D., S. J. McSorley, S. N. Chatfield, G. Dougan, F. Y. Liew. 1995. Protection against *Leishmania major* infection in genetically susceptible BALB/c mice by gp63 delivered orally in attenuated *Salmonella typhimurium* (AroA- AroD-). Immunology **85:** 1-7.
- 66. Yao, C., Donelson, J.E., Wilson, M.E., 2003. The major surface protease (MSP or GP63) of *Leishmania* sp. Biosynhesis, regulation of expression, and function. Molecular and Biochemical Parasitology 132: 1-16.
- 67. Yao, C., K. G. Leidal, A. Brittingham, D. E. Tarr, J. E. Donelson, and M. E. Wilson. 2002. Biosynthesis of the major surface protease GP63 of Leishmania chagasi. Molecular and Biochemical Parasitology 121: 119–128.

CHAPTER 7

Trypanosoma carassii CALRETICULIN BINDS HOST COMPLEMENT COMPONENT C1q AND INHIBITS CLASSICAL COMPLEMENT PATHWAY-MEDIATED LYSIS¹.

7.0 Introduction

Calreticulin (CRT) is a highly conserved multifunctional protein originally identified as a major calcium binding protein of the endo/sarcoplasmic reticulum [24]. CRT has since been reported in every cell of the higher organisms with exception of the erythrocytes. CRTs possess lectin-like chaperone activity, participate in lytic activity of performs from T and natural killer cells, modulate gene expression, enhance phagocytosis of apoptotic cells, inhibit tumoral growth, mediate autoimmunity and inhibit C1q-dependent complement activation [13, 23]. All CRT proteins contain three structural domains: a globular N-terminal domain, a proline rich P domain and an acidic C-terminal domain. The N-terminal domain is involved in protein-protein interactions, RNA and autoantibody binding ([8, 9, 14]. The proline rich P domain binds Ca^{2+} with high affinity and low capacity while the C-terminal domain, which is the least conserved domain among CRTs, binds Ca^{2+} with low affinity [5, 24]. The CRT sequence starts with a signal peptide and ends with KDEL or related endoplasmic reticulum retention sequence [2, 14, 23].

¹A version of this chapter has been published: Oladiran, A., and M. Belosevic. 2010. *Trypanosoma carassii* calreticulin binds host complement component C1q and inhibits classical complement pathway-mediated lysis. Develomental and Comparative Immunology **34:** 396-405.

The role of CRT in host-parasite interactions has recently become an important area of research. CRT molecules from a number of human parasites have been cloned and these sequences display about 50% identity with human CRT [10]. Parasite CRTs bind host C1q and inhibit C1q-dependent complement activation [10, 15, 25, 34]. It has been reported that *Haemonchus contortus* CRT binds host C-reactive protein (CRP) and is present in the ES molecules of the adult worms [34, 35]. These studies indicate parasite CRT may inhibit activation of the classical pathway of complement by binding C1q or CRP. The binding of CRT to CRP, which has anti-parasite as well as procoagulant activities may prevent blood clotting and allow the parasite to feed longer on its host. The ectoparasite *Amblyomma americanum* secretes CRT during feeding [12] suggesting that CRT may use its anti-coagulant ability to prevent blood clotting and allow the parasite to feed longer on the host or divert host anti-parasite responses.

The presence of CRT in the penetration gland cells of schistosome cercariae suggests that this molecule may be important during penetration of host skin [16]. Parasite CRT is important in worm infection and has been reported to induce Th2 type immune response in mice infected with *Heligmosomoides polygyrus* [31]. Experimental vaccination with recombinant *Necator americanus* CRT in absence of adjuvant reduced worm burden by 43-49% in lungs of infected mice compared to control mice [38]. Among protozoan parasites, *T. cruzi* calreticulin (TcCRT) has been fairly characterized. TcCRT binds human C1q and specifically inhibits classical pathway of complement [2, 10]. TcCRT colocalized
with human C1q on parasite surface and antibody against TcCRT inhibits its interaction with human C1q [1, 10]. Taken together, these studies highlight the importance of parasite CRT during infection.

We have reported previously that T. carassii is resistant to complementmediated lysis but removal of parasite surface proteins with trypsin increases their susceptibility to lysis in *in vitro* studies. Lysis occurred through the alternative complement pathway of goldfish since specific antibody was not required and magnesium but not calcium ion was required for lysis. Parasites regained resistance to lysis 6-24 hours following cultivation in the absence of trypsin and resistance to lysis was abrogated with a protein synthesis inhibitor, puromycin [30]. Although T. carassii shares similar surface architecture with T. cruzi, this fish parasite is also related to T. brucei by its extracellular existence. The extracellular nature of T. carassii suggests that host humoral responses may also play roles in parasite lysis *in vivo* through the classical pathway of complement. Thus, there is need to further investigate the importance of this pathway in control of infection and how parasite may evade it. This is the first report on the molecular characterization of CRT molecule of T. carassii (TcaCRT). My working hypothesis was that TcaCRT binds goldfish complement component C1q and inhibits the classical complement pathway. I showed that TcaCRT is located on parasite surface and is present in the supernatants of cultured parasites. Recombinant TcaCRT binds goldfish C1q in vitro and inhibits classical pathwaymediated lysis of sensitized sheep erythrocytes.

7.1 Experimental design

7.1.1 Molecular cloning of *T. carassii* calreticulin

TcaCRT was originally identified in surface protein fraction separated by 2-dimensional gel electrophoresis (2-DE) and analyzed by mass spectrometry (MS). TcaCRT peptides identified in MS analysis are (FYADAEK, CGGGYIK, YWLMFGPDR, EAPMIPNPK, QIPNPAYK). These peptides matched TcCRT with accession number (XM_799098). CRT fragment was amplified using primers designed to *T. carassii* spliced leader sequence and peptides identified in MS studies. The full sequence was amplified and the recombinant protein was produced in prokaryotic system. Purified rTcaCRT was used to immunize rabbits for production of polyclonal antiserum.

7.1.2 Assessment of TcaCRT localization using immunofluorescence and confocal microscopy.

The *in vitro* grown parasites were washed twice in serum-free medium and re-suspended to a concentration of 1×10^6 /mL. 100 µL of parasites were incubated with 80 µg of rabbit anti-rTcaCRT IgG, rabbit anti-goldfish CSF IgG (isotype control) or with PBS for 24 hours at 20°C. Parasites were then spun onto poly-l-lysine coated glass slides using a Cytospin 2 at 55 x *g*, for 7 minutes. The parasites were fixed to the slides in methanol at -20 °C for 30 minutes and permeabilized using 1% Triton X-100 for 30 minutes at room temperature. Following permeabilization, the slides were washed 3 x for 20 minutes each in Milli-Q water. The slides were then blocked with 0.5% BSA in PBS for 30

minutes at room temperature. The slides were then washed 3 times for 20 minutes each in Milli-Q water prior to incubation with FITC-conjugated goat anti-rabbit IgG 1:100 dilution for 2 hours at room temperature. The cells were subsequently washed 3 times for 20 minutes each in Milli-Q water and observed using DIC imaging with a Leica confocal microscope.

7.1.3 Examination of interaction of goldfish complement component C1q with immobilized rTcaCRT.

To study the interaction between goldfish C1q and parasite CRT, 5 mg of purified rTcaCRT was coupled to CNBr-activated Sepharose (GE Healthcare) according to instructions provided by the manufacturer. Three mg of C1q containing serum fraction was loaded onto CRT-Sepharose column equilibrated with 20 mM Tris–HCl (pH 7.5) containing 10 mM CaCl₂. The mixture was incubated at room temperature for 1 hour with gentle rocking to allow proteinprotein interaction. Following incubation, the column was washed 4 times with 2 mL PBS. The bound proteins were subsequently eluted by increasing NaCl concentration (150 mM, 250 mM, 500 mM) and analyzed by SDS-gel electrophoresis. Proteins were transferred to nitrocellulose membrane and C1q was detected with anti-carp C1q-A antiserum using Western blot.

Binding of human C1q to immobilized rTcaCRT was examined to test whether CRT-C1q interaction is an evolutionarily conserved event. Varying concentrations of rTcaCRT diluted in coating buffer (15 mM of Na₂CO3, 35 mM of NaHCO₃, pH 9.6) were coated onto the wells of a microtitre plate (Corning)

232

100 μ L per well for 6 hours at room temperature (22°C to 25°C). Control wells received buffer alone or buffer with 3% w/v BSA. After each step, wells were washed 3 times with 0.05% Tween in PBS. Nonspecific binding sites were blocked with 3% w/v BSA in PBS for 1 h at room temperature. Following blocking, 500 ng of pure human C1q (Sigma-Aldrich) in 100uL in PBS/1% w/v BSA/0.05% Tween 20 was added to each well and the plate was incubated at 4°C overnight. To detect bound C1q, affinity-purified rabbit anti-human C1q antiserum (DAKO) 1:1500 dilution was added in 100 µL final volume and the plates incubated at room temperature for 2 hours. Wells were washed with PBS-Tween followed by the addition of 100μ L affinity-purified goat anti-rabbit IgGperoxidase conjugate (DAKO) 1:3000 dilution. Plate was kept at room temperature for 2 hours followed by washings with PBS-Tween. The bound peroxidase activity was measured by adding OPD (Sigma-Aldrich) as a substrate. The color produced was read at 492 nm in a microplate reader according to manufacturer's instruction.

Co-immunoprecipitation assay was performed to further show that TcaCRT interacts with goldfish C1q. Fifty µg of anti-rTcaCRT or rTcahsp70 was coupled to agarose resin according to manufacturer's instruction (Pierce). Fifty µg of rTcaCRT was incubated with 50 µg of C1q-containing fraction eluted from CNBr-activated sepharose column with 500 mM NaCl in the presence of 20 mM Tris–HCl (pH 7.5) containing 10 mM CaCl₂ for 6 hours at 4°C. The proteins were then incubated with the prepared agarose resin for 1 hour at room temperature. The columns were washed and bound proteins were eluted and separated by SDS- PAGE. Proteins were transferred to nitrocellulose membranes and eluted proteins were detected by Western blot.

7.1.4 Haemolytic assays

To assess the effect of parasite CRT on activation of classical pathway of complement, washed sheep erythrocytes (SRBC) re-suspended in 0.9% NaCl were sensitized by adding an equal volume of 1:250 diluted decomplemented rabbit anti-sheep erythrocyte antiserum in PBS containing 1 mM CaCl₂. Cells were washed thrice with 0.9% NaCl by centrifugation at 400 x *g* for 10 min. A total 10^7 sensitized erythrocytes in 0.9% NaCl containing 1 mM CaCl₂ was added to wells of a microtitre plate containing varying concentrations of TcaCRT or BSA mixed with fresh normal rabbit serum (1/100 dilution) in a total assay volume of 200 µL. The plate was incubated at 37°C for 45 minutes and then kept at 4°C for 3 hours. Haemoglobin in the supernatant was measured by adding 50 µL of supernatant to new wells containing 50 µL 0.9% NaCl and absorbance was measured at 415 nm. Total haemolysis was measured by lysing 10^7 sensitized erythrocytes with water and haemolytic activity was expressed as per cent of the total haemolysis.

7.2 Results

7.2.1 TcaCRT is present in membrane protein fraction and may be secreted

I [28] reported previously the identification of *T. carassii* hsp70 in ES products and surface protein fraction. TcaCRT was also identified in the surface

234

protein fraction (Fig. 7.1), suggesting surface expression of this molecule and a possibility that it may be secreted. I [27] cloned TcaCRT and expressed the recombinant protein in prokaryotic system. Alignment of TcaCRT with CRTs from other trypanosomatids is shown (Fig. 7.2). TcaCRT shares 62% similarity with TcCRT. The protein structure of TcaCRT compared with TcCRT is shown in Fig. 7.3. The recombinant protein, ES products and whole cell lysate were separated using SDS-PAGE and CRT was detected using Western blot with mouse anti-polyhistidine antibody (Fig. 7.4A), and rabbit anti-rTcaCRT antibody (Fig. 7.4B). Rabbit anti-rTcaCRT antibody reacted with TcaCRT in ES products (Fig. 7.4C) and native TcaCRT in whole cell lysate (Fig. 7.4D). Identification of TcaCRT in ES products was consistent with reports from other parasites that identified CRT in the ES products [11, 15, 32, 35]. Secreted *T. carassii* CRT has been suggested to play a role in immune evasion or immune modulation by parasites.

7.2.2 TcaCRT is expressed on parasite surface

That TcaCRT is identified in ES and surface protein fractions suggests surface expression of this molecule. To further show that TcaCRT is expressed on the parasite surface and may act as receptor for host C1q *in vivo*, I used indirect immunofluorescence. Parasites were incubated with rabbit anti-TcaCRT antibody, isotype control antibody or PBS. Bound antibodies were detected with FITC-conjugated goat anti-rabbit IgG (1:100). The result of confocal microscopy indicates TcaCRT is expressed on the surface of the parasites (Fig. 7.5C).

7.2.3 Anti-carp C1q-A antibody cross-reacts with goldfish C1q.

Interaction between parasite CRT and host C1q have been shown for *T*. *cruzi* and several heminths that infect humans [10, 15, 34, 35]. To assess possible interactions between goldfish C1q and TcaCRT, I examined whether carp anti-C1q-A antibody cross-reacted with goldfish C1q. C1q-containing serum fraction was prepared and protein fraction was separated by SDS-PAGE. Western blot of nitrocellulose containing serum proteins indicate anti-carp C1q-A antiserum recognized goldfish C1q (Fig. 7.6).

7.2.4 rTcaCRT binds goldfish C1q in vitro

I showed that rTcaCRT immobilized on activated sepharose retained structural conformation necessary for interaction with host C1q (Fig. 7.7A). Goldfish C1q bound immobilized CRT, as well as other serum proteins. Nonspecifically bound proteins were removed by washing the column 4 times with PBS and with 150 mM NaCl. C1q was specifically eluted at higher salt concentration (500 mM) and recognized by anti-carp C1q-A antiserum (Fig 7.7B). C1q is a large molecule of about 460KDa composed of 18 polypeptide chains held together by disulfide bonds [4, 33]. The individual polypeptide chains are tightly held in the C1q complex and are difficult to separate from this complex. Since protein separation was performed on SDS-PAGE under denaturing and reducing condition, it is likely that the three prominent protein bands seen on the stained blot are different subunits of C1q. I used anti-carp-C1q-A antibody, which was raised against carp C1q-A for detection and this antibody may not recognize C1qB and C1q-C. Alternatively, these bands may be degradation products of C1q, which can no longer be recognized by anti-carp C1q-A antibody. Calreticulin also binds other serum proteins including members of the collectin family mannose binding protein (MBP), conglutinin and lung surfactant protein (SP-A). The binding of host CRP with a molecular weight of 24KDa and blood clotting protein Factor X with molecular weight of 55KDa to immobilized parasite CRT has been reported [34, 35].

The binding of other serum proteins to immobilized rTcaCRT may prevent rTcaCRT from binding most of the C1q in the serum fraction, which may explain the detection of C1q in the unbound fraction. To further demonstrate that CRT bound host C1q, I employed far Western blot technique. C1q-containing serum fractions were separated using native PAGE and transferred to nitrocellulose membranes. Blots were incubated with rTcaCRT or PBS as control. Following washes with TTBS and TBS, bound rTcaCRT was detected using rabbit anti-rTcaCRT IgG, while non bound rTcaCRT was detected on blots incubated with PBS alone (Fig. 7.8A and B).

Co-immunoprecipitation experiments were done to examine the interaction between rTcaCRT and goldfish C1q. Rabbit anti-rTcaCRT IgG was coupled to agarose resin and used to precipitate rTcaCRT and goldfish CIq. Immunoprecipitated proteins were separated by SDS-PAGE and transferred to nitrocellulose membranes. Blots were probed with rabbit anti-carp C1q-A antiserum (1:1500), mouse monoclonal anti-polyhistidine antibody (1:5000) and rabbit anti-rTcaCRT antibody (1:1500). Blots were washed and proteins bands detected with alkaline phosphatase conjugated goat anti-rabbit IgG (1:3000) or goat anti-mouse IgG (1:1500). C1q in the immunoprecipitate was detected using anti-carp C1q-A antiserum (Fig. 7.9, lane 1). TcaCRT in the immunoprecipitate was detected with anti-histidine antibody (Fig. 7.9, lane 2) and anti-rTcaCRT antibody (Fig. 7.9, lane 3). No proteins were detected in elution from the control resin (Fig. 7.9, lane 4) and elution from anti-rTcahsp70 column (Fig. 7.9, lane 5).

7.2.5 rTcaCRT binds human C1q

Heligmosomoides polygyrus is a parasitic nematode of mice whose CRT molecule has been reported to bind human C1q [31]. For this reason, I investigated whether TcaCRT can interact with human C1q using ELISA. Result of ELISA shown in Figure 7.10 indicates that like *H. polygyrus* CRT, the *T carassii* recombinant CRT bound human C1q, suggesting an evolutionarily conserved interaction irrespective of species.

7.2.6 rTcaCRT inhibits C1q-dependent complement lysis

The implications of CRT-C1q interaction was investigated using classical complement pathway-dependent haemolysis of sensitized sheep erythrocytes (Fig. 7.11). I used 1/100 dilution of normal rabbit serum as source of complement since majority of C1q is associated with C1r and C1s under normal physiological conditions. Under these conditions, we observed 80-100% haemolysis in distilled water. Lysis of sensitized sheep erythrocytes was significantly inhibited in the presence of rTcaCRT as shown by decreased haemolytic activity (15.33% to

13.62%) compared to BSA control that showed no significant decrease in haemolytic activity (94.33% to 79.29%) (Fig. 7.11). Addition of anti-rTcaCRT restored hemolysis to about 80% suggesting that inhibition of hemolysis is due to rTcaCRT.

7.3 Discussion

During host-parasite interaction, important roles of parasite molecules present in ES may include helping parasites to establish infection, protect parasite from host immune molecules by binding those molecules away from parasites, divert immune responses away from parasites by causing non-specific stimulation of immune response and tissue penetration [11, 28, 32]. T. carassii is an extracellular pathogen and molecules secreted by parasites may play roles in evasion or modulation of host immunity. It has been demonstrated in previous studies that in vitro shed ES antigens from T. carassii confer protection against challenge infection [6]. I [27, 28] originally identified parasite TcaCRT in 2-DE and mass spectrometry analysis of parasite surface protein fractions suggesting surface expression of this molecule. For this reason, I examined the ES products of parasites and identified secreted TcaCRT in the ES products. In T. cruzi cell surface expressed TcCRT has been shown to interact with host C1q [10], providing the parasites with a strategy to inhibit C1q-dependent activation of the classical pathway. The ability of CRT molecules of human helminth parasites to interact with host C1q the recognition molecule of the classical pathway of complement has been documented [15, 25]. In this study, I demonstrated that

CRT from the fish parasite *T. carassii* interacted with host C1q and inhibited C1qdependent haemolytic activity *in vitro*. I [27] showed that TcaCRT was expressed on parasite surface and that it may provide the parasites with an elegant strategy to evade complement-mediated lysis. This is the first report of CRT-C1q interaction in ectothermic vertebrates.

CRT was also shown to interact with CRP [34], and since CRP may activate the classical complement pathway by binding to C1q, this may be another mechanism by which CRT may influence complement-mediated lysis. As an acute phase protein, CRP has been shown to exhibit antiparasite effects. For example, the incubation of adult *Hymenolepis diminuta* in serum containing CRP resulted in a loss of motility and damaged tegument, reminiscent of damage caused by complement [36]. CRP-depleted serum showed no antiparasitic activity while de-complemented serum was fully active suggesting that this effect was CRP-specific. These findings suggest CRT-CRP interactions may have important physiological role in parasite survival and we have initiated studies to further characterize CRP-CRT interactions in goldfish infected with *T. carassii*.

Although recombinant TcaCRT failed to bind calcium probably due to improper folding resulting from expression in prokaryotic system, it retained the structural architecture required for interaction with C1q, as indicated by the inhibition of C1q-mediated haemolysis, following addition of rTcaCRT. Also, functional forms of CRT have been produced in prokaryotic system [5].

I [27] showed that TcaCRT interacted with goldfish C1q and inhibited classical complement pathway-mediated haemolysis of sensitized sheep

erythrocytes. TcaCRT was secreted by the in vitro-grown trypanosomes and indirect immunofluorescence showed that this molecule is abundantly expressed on the parasite surface. The presence of TcaCRT on parasite surface may provide parasite with an elegant means to inhibit C1q-mediated activation of classical complement pathway. Additionally, secreted CRT molecules may perform other functions such as binding of host molecules away from the parasites, non-specific stimulation of immune responses or immunosuppression. That TcaCRT also bound human C1q suggests an evolutionarily conserved interaction between these two molecules. This observation was further supported by our haemolytic assay results, demonstrating that the interaction of rTcaCRT with C1q in normal rabbit serum inhibited classical complement pathway. Parasite CRTs have been reported to possess multiple C1q-binding motifs, which are thought to enhance specific interaction with host C1q molecules [25], and it was suggested that multiple C1q-binding motifs in parasite CRTs may be an evolutionary strategy to compensate for multipoint interaction between antibody and C1q to ensure complete inactivation of the classical complement pathway [25]. The presence of such multiple C1q motifs in parasite CRTs may allow CRTs to bind diverse C1q molecules.

It has been documented that complement-mediated lysis of *T. cruzi* was primarily due to the action of alternative compement pathway since serum depletion of factor B and Properdin abrogated lysis of IgG-coated trypanosomes [20]. The classical pathway appeared to be ineffective in parasite lysis, but its role was to amplify the lytic effects of the alternative complement pathway. Sera from individuals that are genetically deficient in the classical pathway component C2 lysed 50% of IgG-coated parasites compared to controls treated with normal sera and addition of increasing amount of purified C2 to the deficient sera restored lysis to levels comparable to that of controls [20]. We have previously shown that trypsin treatment of T. carassii increases their susceptibility to lysis by the alternative complement pathway of goldfish, suggesting a similar mechanism to lysis of *T. cruzi*. Parasites regained resistance to lysis 6-24 hrs after cultivation in the absence of trypsin and resistance to lysis was abrogated with puromycin a protein synthesis inhibitor [30]. The expression of CRT on the parasite surface may be the reason why the classical compement pathway appears to play a minor role in control of T. cruzi. Similarly, it is probable that T. carassii CRT present on the surface of the trypanosomes may inhibit the classical complement pathway. This would provide the parasite with a strategy to interfere with host complement system at the earliest stage of complement activation. T. carassii is an extracellular parasite like the African trypanosomes but does not exhibit antigenic variation [29]. The similarity in surface architecture between T. cruzi and T. carassii suggest presence of major complement regulatory proteins in T. carassii that are similar to that of T. cruzi [22, 26]. The TcaCRT may be part of a cascade of complement binding molecules elaborated on the surface of *T. carassi* that may be used in evasion of host immune attack.

T. cruzi infected individuals produce variable levels of specific antibodies against TcCRT in their serum [3, 21] and these antibodies may inhibit TcCRT-human C1q interaction which may benefit the hosts. Indeed antibody raised

against TcCRT inhibits its interaction with human C1q *in vitro* [1]. In this study, I observed that that the addition of anti-rTcaCRT antibody restored haemolysis of sensitized erythrocytes *in vitro*. Since TcaCRT shows 62% similarity to TcCRT, it possible that TcaCRT will have same functional effects as TcCRT. Furthermore, during the course of chronic *T. cruzi* infection, vertebrate hosts produce antibodies that convert complement resistant trypomastigotes into complement sensitive parasites. These antibodies have been referred to as lytic antibodies [18] and are responsible for antibody-dependent complement lysis of parasites primarily via the alternative pathway [7, 20]. It has been demonstrated that Fab and Fab antibody fragments derived from the IgG of Chagasic patients promoted complement-mediated lysis of *T. cruzi* trypomastigites [17] indicating that the lysis-promoting activity of antibodies occur independently of Fc portion of antibodies .

At present, anti-TcCRT antibody is not classified as a lytic antibody perhaps because of the inefficient nature of the classical pathway-mediated lysis on trypanosomes. However, a number of *T. cruzi* molecules, which includes the major complement regulatory proteins that elicit production of lytic antibodies in the host, have been reported [18]. Characterization of parasite molecules that elicit lytic antibodies is hampered by the fact these antibodies are difficult to elicit by immunization with dead parasites or with purified parasite antigens [19]. However, lytic antibodies have been elicited by inoculation with attenuated *T. cruzi* trypomastigotes and with trypomastigotes ES antigens [37]. These observations suggest that molecules that elicit lytic antibodies are labile and shed proteins are important and can sufficiently confer complement resistance.





T.cruzi	MRAAIFFCALLSLAT-LSAVHGTVYFHEEFKSMEHWTTSKHRDDFGKVEISAGKFY	55
T.brucei	MLMCMRPVAVACVFVALAT-VATVHGAIHFHEKFSSIDHWTASKARSDYGKVELSAGKFY	59
T.carassii	MRASLLLIALVGFAAVLTTVSATVLFHEDFKSIDKWISSAHRDDYGAIGHSAGKFY	56

T.cruzi	ADAEKSKGLRLTEDARFYALSTAFPTPITNEKKSLVVSFSVKHEQDLKCGGGYIKLLPSM	115
T.brucei	ADAEKSKGLRLTEDARFYALSTPLPTPITNEKKDFVVSFSVKHEQDLRCGGGYIKLLPQM	119
T.carassii	ADAEKDKGLQLTQDARFYAVSAKLPTPITNDKKEFVVSFSVKHEQGLKCGGGYIKLLPTL	116
	*****.*********************************	
T.cruzi	DPEKFHGETKYWLMFGPDRCGSQN-RVHIILHYNGENREWSKRIRFPEDKLTHVYTLHIA	174
T.brucei	DPAELKGETKYWLMFGPDRCGYDK-KIHIIISYNGANREWKKRPSYPDDRLTHVYTLHIT	178
T.carassii	DPKDFHGDSKYWLMFGPDRCGYDNNKVHIILNHDGTNHQWKKKVAFPDDKLTHVYTLRIS	176
	** .::*::*:****************************	
T.cruzi	ADNSYEFFLDGESKAKGQLDGDWTLLPPREIVDETDKKPEDWVDEETMDDPEDKKPEDWD	234
T.brucei	PSNSYEFFLDGVSKEKGTLEADWDFLPEKEIDDPEDKKPADWVDVPTIDDPEDKKPEDWD	238
T.carassii	SDDSYELYVDEELKEKGSLQDDWPIVQPKEISDATDKKPEDWVDEPTMADPEDKKPEDWD	236
T.cruzi	NEPAMIPDADAKKPDDWDDAEDGPWEAPMIPNPKSKGTWKPRQIPNPAYKGVWEPRKIPN	294
T.brucei	SEPEKIVDPEAKKPEDWNDAEDGAWEAPMISNPKSKGPWAPRKIPNPAYKGPWAPRRIPN	298
T.carassii	SEPQTIPDAEAKKPEDWDDAEDGEWEAPMIPNPKSKGPWHPKQIPNPAYKGPWEARMIPN	296
T.cruzi	PDFVEDSELHKVPEPLTHVGIDVWQVESGSIFKDIMIGDDLKEVLELVEKTYGSLKKAEA	354
T.brucei	PAYKNDEELYKIPEPLTHVGIDVWQVESGSIFKDIIIGDDVKEVLDIVKSTYDGMKKAEE	358
T.carassii	PDYKPIPDLYKIPEALEYVGIDVWQVESGSIFNNIIIGDDVKEVLEIVKGTYGASKKAES	356
	• : :*:*:*:* :*************************	
T.cruzi	DALKVMEDMEKEKRKKEEEEEKEKEK-AEEEKDEEELEEKGDGDKEDL 401	
T.brucei	DALAAFEKKEEQDEKEEDK-KETDGEEDKKKKEDKSDL 395	
T.carassii	DAFEAFKKAEDAAAKAEEANTATEVTIDNGDTKDAEEEGDL 397	
	*** * ** * *** * ***	

Figure 7.2. Amino acid alignment of *Trypanosoma carassii* **calreticulin** (**TcaCRT**) **sequence and those from** *T. cruzi* **and** *T. brucei*. Sequences were obtained from GenBank accession numbers *T. cruzi* (XM_799098), *T. brucei* (XM_842477). (*) indicates identical residues, (:) indicates conserved residues while (.) indicates semi-conserved residues. (-) are gaps introduced to maximize the alignment. The possible cleavage site for TcaCRT molecule is indicated with arrow. A bar denotes the endoplasmic reticulum retention signal sequences at the C-terminal end of the molecules.



Figure 7.3. Schematic of protein structure of *T. carassii* **CRT.** *T. carassii* CRT was compared with *T. cruzi* CRT. The coloured boxes represent CRT family signatures. The approximate locations of the CRT family signatures are shown. The three domains of CRT molecules, their approximate positions and functions are indicated.



Figure 7.4. Rabbit anti-recombinant TcaCRT affinity purified rabbit IgG polyclonal antibody recognizes TcaCRT in ES products and whole cell lysate of *T. carassii*. rTcaCRT (A and B), concentrated *T. carassii* ES products (C) and whole cell lysates (D) were separated by SDS-PAGE and transferred to nitrocellulose. (A) Western blotting of rTcaCRT performed with mouse anti-polyhistidine antibody (1: 5000). (B) Western blot of rTcaCRT with rabbit anti-rTcaCRT IgG (1:1500). (C) Western blot analysis of *T. carassii* ES products using rabbit anti-rTcaCRT IgG (1: 1500). (D) Western blot of *T. carassii* whole cell lysate with rabbit anti-rTcaCRT IgG (1: 1500). The difference in size of the recombinant molecules compared to the native molecules is due to the additional vector sequence.







Figure 7.6. Rabbit anti-carp C1q-A antiserum recognizes goldfish C1q. Sera were collected form carp and goldfish blood and C1q fraction obtained by PEG precipitation. Two μ L of goldfish or carp serum (lanes A and D) and 5 μ g of goldfish or carp PEG 4000 precipitated C1q-containing fractions (lanes B and C) were separated by SDS-PAGE and transferred to nitrocellulose membrane. C1q was detected using rabbit anti-carp C1q-A antiserum (1:1500) and AP-conjugated goat anti-rabbit IgG (1:3000).



Figure 7.7. Goldfish C1q binds immobilized rTcaCRT. Following coupling of rTcaCRT to CNBr-Sepharose, C1q-containing serum fraction was added to the column. CRT-C1q column was incubated at room temperature to allow for protein interaction and column was washed with phosphate-buffered saline. C1q was eluted using 500 mM NaCl. Eluted fractions were separated by SDS-PAGE and transferred to nitrocellulose membrane. Blots were stained with colloidal gold total protein stain (Bio-Rad) (A). Western blot using rabbit anti-carp C1q-A antiserum (B). Protein concentration was similar in all lanes of the gels. Lane 1: C1q-containing serum fraction before addition to the column. Lane 2: Unbound proteins eluted from column after rTcaCRT-C1q interaction. Lanes 3 and 4: Proteins eluted using 150 mM NaCl following washes with PBS. Lane 5: Proteins eluted using 250 mM NaCl. Lanes 6 and 7: Proteins eluted using 500 mM NaCl.











Figure 7.10. *T. carassii* **CRT binds human C1q.** Varying concentrations of rTcaCRT were coated onto wells of ELISA plate as described. Control wells were coated with buffer alone or coating buffer containing BSA. Wells were blocked with PBS/1% w/v BSA/0.05% Tween 20 before incubation with pure human C1q. Bound C1q was detected using anti-human C1q IgG (1:1500) and HRP-conjugated goat anti-rabbit IgG (1:3000). Statistical analysis was done using one-way ANOVA. (*) denotes significantly different (P < 0.05) from coating buffer alone or coating buffer + BSA. The experiment was performed twice with similar result.





7.4 References

- Aguilar, L., G. Ramírez, C. Valck, M.C. Molina, A. Rojas, W. Schwaeble, V. Ferreira, and A. Ferreira. 2005. F(ab')2 antibody fragments against *Trypanosoma cruzi* calreticulin inhibit its interaction with the first component of human complement. Biological Research 38: 187-195.
- Aguillon, J. C., L. Ferreira, C. Perez, A. Colombo, M.C. Molina, A. Wallace, A. Solari, P. Carvallo, M. Galindo, N. Galanti et al. 2000. Tc45, a dimorphic *Trypanosoma cruzi* immunogen with variable chromosomal localization, is calreticulin. American Journal of Tropical Medicine and Hygiene 63: 306-312.
- Aguillón, J. C., R. Harris, M. C. Molina, A. Colombo, C. Cortés, T. Hermosilla, P. Carreño, A. Orn, and A. Ferreira. 1997. Recognition of an immunogenetically selected *Trypanosoma cruzi* antigen by seropositive chagasic human sera. Acta Tropica 63: 159-166.
- Arlaud, G. J., C. Gaboriaud, N. M. Thielens, V. Rossi, B. Bersch, J. F. Hernandez, and J. C. Fontecilla-Camps. 2001. Structural biology of C1: dissection of a complex molecular machinery. Immunology Review 180: 136-145
- Baksh, S., and M. Michalak. 1991. Expression of calreticulin in *Escherichia coli* and identification of its Ca2+ binding domains. Journal of Biological Chemistry 266: 21458-21465.
- Bienek, D. R., D. A. Plouffe, G. F. Wiegertjes, and M. Belosevic. 2002. Immunization of goldfish with excretory/secretory molecules of *Trypanosoma danilewskyi* confers protection against infection. Developmental and Comparative Immunology 26: 649-657.
- Budzko, D. B., M. C. Pizzimenti, and F. Kierszenbaum. 1975. Effects of complement depletion in experimental chagas disease: immune lysis of virulent blood forms of *Trypanosoma cruzi*. Infection and Immunity 11: 86-91.

- Cheng, S. T., T. Q. Nguyen, Y. S. Yang, J. D. Capra, and R. D. Sontheimer. 1996. Calreticulin binds hYRNA and the 52-KDa polypeptide component of the Ro/SS-A ribonucleotide autoantigen. Journal of Immunology 156: 4484-4491.
- Dedhar, S. 1994. Novel functions for calreticulin: interaction with integrins and modulation of gene expression? Trends in Biochemical Science 19: 269-271
- Ferreira, V., C. Valck, G. Sanchez, A. Gingras, S. Tzima, M. C. Molina, R. Sim, W. Schwaeble, and A. Ferreira. 2004. The classical activation pathway of the human complement system is specifically inhibited by calreticulin from *Trypanosoma cruzi*. Journal of Immunology **172**: 3042-3050.
- Guillou, F., E. Roger, Y. Mone, A. Rognon, C. Grunau, A. Theron, G. Mitta, C. Coustau, and B. E. Gourbal. 2007. Excretory-secretory proteome of larval *Schistosoma mansoni* and *Echinostoma caproni*, two parasites of *Biomphalaria glabrata*. Molecular and Biochemical Parasitology 155: 45-56.
- Jaworski, D. C., J. A. Higgins, S. Radulovic, J. A. Vaughan, and A. F. Azad. 1996. Presence of calreticulin in vector fleas (Siphonaptera) Journal of Medical Entomology 33: 482-489.
- Johnson, S., M. Michalak, M. Opas, and P. Eggleton. 2001. The ins and outs of calreticulin: from the ER lumen to the extracellular space. Trends in Cell Biology 11: 122-129
- 14. Joshi, M., G. P. Pogue, R. C. Duncan, N. S. Lee, N. K. Singh, C. D. Atreya, D. M. Dwyer, and H. L. Nakhasi. 1996. Isolation and characterization of *Leishmania donovani* calreticulin gene and its conservation of the RNA binding activity. Molecular and Biochemical Parasitology 81: 53-64.
- Kasper, G., A. Brown, M. Eberl, L. Vallar, N. Kieffer, C. Berry, K. Girdwood, P. Eggleton, R. Quinnell, and D. I. Pritchard. 2001. A calreticulin-like molecule from the human hookworm *Necator americanus*

interacts with C1q and the cytoplasmic signalling domains of some integrins. Parasite Immunology **23:** 141-152

- Khalife, J., J. L. Liu, R. Pierce, E. Porchet, C. Godin, and A. Capron. 1994. Characterization and localization of *Schistosoma mansoni* calreticulin Sm58. Parasitology **108**: 527-532.
- Kipnis, T. L., A. U. Krettli, and W. Dias da Silva. 1985. Transformation of trypomastigote forms of *Trypanosoma cruzi* into activators of alternative complement pathway by immune IgG fragments. Scandinavian Journal of Immunology 22: 217-226.
- Krautz, G. M., J. C. Kissinger, and A. U. Krettli. 2000. The targets of the lytic antibody response against *Trypanosoma cruzi*. Parasitology Today 16: 31-34
- Krettli, A. U., and Z. Brener. 1982. Resistance against *Trypanosoma cruzi* associated to anti-living trypomastigote antibodies. Journal of Immunology **128**: 2009-2012.
- Krettli, A. U., P. Weisz-Carrington, and R. S. Nussenzweig. 1979.
 Membrane- bound antibodies to bloodstream *Trypanosoma cruzi* in mice: strain differences in susceptibility to complement-mediated lysis. Clinical Experimental Immunology **37:** 416-423.
- 21. Marcelain, K., A. Colombo, M. C. Molina, L. Ferreira, M. Lorca, J. C. Aguillón, and A. Ferreira. 2000. Development of an immunoenzymatic assay for the detection of human antibodies against *Trypanosoma cruzi* calreticulin, an immunodominant antigen. Acta Tropica **75**: 291-300.
- 22. Mathieu-Daudé, F., B. Lafay, O. Touzet, J. Lelièvre, F. Parrado, M. F. Bosseno, A. M. Rojas, S. Fatha, A. Ouaissi, and S. F. Brenière. 2008. Exploring the FL-160-CRP gene family through sequence variability of the complement regulatory protein (CRP) expressed by the trypomastigote stage of *Trypanosoma cruzi*. Infection, Genetics and Evolution 8: 258-266.
- Michalak, M., E. F. Corbett, N. Mesaeli, K. Nakamura, and M. Opas.
 1999. Calreticulin: one protein, one gene, many functions. Biochemistry

Journal **344:** 281-292

- Michalak, M., R. E. Milner, K. Burns, and M. Opas. 1992. Calreticulin. Biochemistry Journal 285: 681-692.
- Naresha, S., A. Suryawanshi, M. Agarwal, B. P. Singh, and P. Joshi. 2009. Mapping the complement C1q binding site in *Haemonchus contortus* calreticulin. Molecular and Biochemical Parasitology 166 : 42-46.
- Norris, K. A., J. E. Schrimpf, and M. J. Szabo. 1997. Identification of the gene family encoding the 160-kilodalton *Trypanosoma cruzi* complement regulatory protein. Infection and Immunity 65: 349-357.
- Oladiran, A., and M. Belosevic. 2010. *Trypanosoma carassii* calreticulin binds host complement component C1q and inhibits classical complement pathway-mediated lysis. Develomental and Comparative Immunology 34: 396-405
- Oladiran, A., and M. Belosevic. 2009. *Trypanosoma carassii* hsp70 increases expression of inflammatory cytokines and chemokines in macrophages of the goldfish (*Carassius auratus* L.). Developmental and Comparative Immunology 33: 1128–1136.
- Overath, P., J. Haag, M. G. Mameza, and A. Lischke. 1999. Freshwater fish trypanosomes: definition of two types, host control by antibodies and lack of antigenic variation. Parasitology **119**: 591-601.
- Plouffe, D. A., and M. Belosevic. 2004. Enzyme treatment of *Trypanosoma danilewskyi* (Laveran &Mesnil) increases its susceptibility to lysis by the alternative complement pathway of goldfish, *Carassius auratus* (L.). Journal of Fish Disease 27: 277-285
- Rzepecka, J., S. Rausch, C. Klotz, C. Schnoller, T. Kornprobst, J. Hagen, R. Ignatius, R. Lucius, and S. Hartmann. 2009. Calreticulin from the intestinal nematode *Heligmosomoides polygyrus* is a Th2-skewing protein and interacts with murine scavenger receptor-A. Molecular Immunology 46: 1109-1119.
- Santarém, N., R. Silvestre, J. Tavares, M. Silva, S. Cabral, J. Maciel and A. Cordeiro-da- Silva. 2007. Immune Response Regulation by

Leishmania Secreted and Nonsecreted Antigens. Journal of Biomedicine and Biotechnology 85154.

- 33. Sellar, G. C., D. J. Blake, and K. B. Reid. 1991. Characterization and organization of the genes encoding the A-, B- and C-chains of human complement subcomponent C1q. The complete derived amino acid sequence of human C1q. Biochemistry Journal 274: 481-90
- Suchitra, S., K. A. Anbu, D. K. Rathore, M. Mahawar, B. P. Singh, P. Joshi. 2008. *Haemonchus contortus* calreticulin binds to C-reactive protein of its host, a novel survival strategy of the parasite. Parasite Immunology **30:** 371-374.
- 35. Suchitra, S., and P. Joshi. 2005. Characterization of *Haemonchus* contortus calreticulin suggests its role in feeding and immune evasion by the parasite. Biochimica and Biophysica Acta **1722**: 293-303.
- Taylor, K., and D. Hoole. 1997. Interactions between rat C-reactive protein and adult *Hymenolepis diminuta*. Parasitology 115: 297-302
- Umezawa, E. S., A. M. Stolf, and B. Zingales. 1993. *Trypanosome cruzi*: different surface antigens of trypomastigotes are targets of lytic antibodies. Acta Tropica 54: 41-53
- Winter, J. A., O. R. Davies, A. P. Brown, M. C. Garnett, S. Stolnik and D. I. Pritchard. 2005. The assessment of hookworm calreticulin as a potential vaccine for necatoriasis. Parasite Immunology 27: 139-146.

CHAPTER 8

GENERAL DISCUSSION

8.0 Evasion of Host Immune Responses by Parasites

Parasites are eukaryotic pathogens and broadly consist of protozoa, fungi, helminths and arthropods. They complete part or all of their lifecycle within another organism, the host. As is the case with every other pathogenic organism, parasites have evolved ways to survive the highly potent immune response encountered in their various hosts. The immune evasion was described over 100 years ago by Paul Ehrlich (Nobel Lecture 1908). Although, a relatively faint idea at the time, Ehrlich reported on the 'disappearance of receptors' in African trypanosomes. This mechanism is now known as antigenic variation. Antigenic variation has since been studied in detail in parasites and bacteria [13, 26, 53, 98].

The immune evasion strategies employed by parasites are varied and include avoidance of immune recognition, immune suppression and diversion of host immune attack mechanisms. All the major parasite groups are known to deploy immune evasion mechanisms [11, 23, 85, 91].

Parasites use armamentarium of immune evasion strategies to interfere with, disrupt or manipulate host immune defences at the different stages of immune response. Hosts deploy their immune system to prevent infections [82, 88]. Ultimately, any immune response generated against the parasite must employ mechanisms that are able to control, contain or kill the parasite. The innate immune system is the first line of defense against pathogens and includes the pattern recognition receptors (PRRs) that recognise pathogen-associated molecular patterns (PAMPs). Following detection, the host uses a number of innate immune components such as NK cells, phagocytes, complement and antimicrobial peptides as the eventual means by which the invading parasites are eliminated [59]. Innate immunity can inhibit parasite growth but it rarely eliminates successful parasites. Thus, lymphocytes of the adaptive immune system are activated by actions of innate immune cells leading to induction of T cell proliferation and antibody production by B cells [59]. These antigen-specific T and B cells are effector cells of the adaptive immunity and generate a potent reaction to eliminate infection. Evasion or modulation of adaptive immune response is therefore crucial for parasite survival and persistence [89].

Protozoan parasites that proliferate rapidly in the hosts possess strategies to subvert host innate and adaptive immune systems. These ensure survival of these parasites, persistence of infection and increase the transmission potential. The African trypanosomes use antigenic variation whereby expression of distinct surface molecules allows new variants to avoid early recognition and repopulate the vascular system of the host, and replace those parasites killed by adaptive immune system [13]. *T. cruzi* has adopted an intracellular existence to escape recognition by host immunity. The surface of *T. cruzi* is covered by mucin glycoproteins that further allow parasite to evade recognition by host immunity and aid in establishment of infection [15].

During the passage of *P. falciparum*-infected erythrocytes through the spleen, parasite-derived proteins are exported to the erythrocyte surface to avoid clearance from the blood. These proteins of which the best characterized is *P*.

falciparum erythrocyte membrane protein 1 (PfemP1), bind to specific receptors on venular endothelial cells and allow the parasite to be retained [6]. Host antibodies can recognize these proteins and prevent adherence of infected erythrocytes to different molecules on venular endothelium. However, a mechanism of clonal antigenic diversity of PfemP1 encoded by over 50 '*var*' genes per genome is required to maintain sequestration and increase transmission potential of the parasite [14]. It has been suggested that cytoadherence evolved as a mechanism of immune evasion to avoid splenic clearance but plays a secondary role in conferring pathogenicity [81]. This phenomenon, described as antigenic diversity, which involves expression of antigenically different alleles of a gene in the different *P. falciparum* is different from antigenic variation. Antigenic variation described above for the African trypanosome is a rapid, stochastic process by which a clonal parasite population switches its antigenic phenotype and has also been observed in different *Plasmodium* species [10, 42, 62, 83].

Antigenic variation has also been described as a major immune evasion strategy in *Giardia*, the flagellated protozoan parasites of small intestine that causes diarrhoea in humans and other animals. *Giardia* surface antigens known as variant-specific surface proteins (VSPs) play roles in parasite attachment and are continuously switched to evade host humoral response [63, 65]. The VSPs are highly immunogenic and cover the entire surface of the parasite acting as a dense interface between the parasite and the host [78]. At any one time, only 1 of approximately 200 VSP genes encoded by the genome is expressed on the surface of an individual parasite, although simultaneous expression of two different VSPs can occur during VSP switching [66, 80]. It has been reported that trophozoites of *Giardia* undergo antigenic variation both within the host and in cell culture [66]. *Giardia*-derived products have also been reported to contribute to immune evasion and establishment of chronic infection [24].

Slow maturing multicellular helminths do not replicate in their human host and do not use antigenic variation or switching. They have larger genome size and have evolved several alternative immune evasion mechanisms. The large size of helminth parasites provides them with a large and varied number of antigens that must be recognized by the host immune system for development of an effective immune response. Some helminths possess ability to mask themselves from an effective immune response by acquiring antigens from their host [93]. Schistosomes establish chronic infections in their host by acquiring host molecules including immunoglobulins [48, 102], major histocompatibility complex [19, 92], complement components [87, 94], C3 decay- accelerating factor [43], β_2 -microglobulin [96] and α_2 -macroglobulin [27, 49]. These parasites express a surface form of invertebrate muscle protein paramyosin which is functionally and antigenically related to human CD59, a membrane bound complement regulatory protein. Paramyosin, like CD59 binds human C8 and C9, and inhibits assembly of membrane attack complex C5b-9 [28, 77]. Paramyosin has also been found in the tegument and on the surface of *Echinococcus* granulosus [54], Taenia solium [64] and Fasciola hepatica [16].

Generally, the excretory/secretory products of worms may induce immunosuppression by impairing or modulating immune cell functions. ES products from gastrointestinal nematode *Heligmosomoides polygyrus* were shown to inhibit cytokine and chemokine production, and co-stimulatory molecules expression induced by TLR ligation in dendritic cells. These effects led to impairment of dendritic cell functions and suppression of both Th1 and Th2 immune responses [90]. Trichinella spiralis, a zoonotic nematode and food borne parasite suppresses host immune response during infection. T. spiralis infection poses serious challenge to the host since parasite lifecycle is completed in one host resulting in exposure of the host to parasite and parasite-derived molecules from all stages of development. The ES products of T. spiralis modulate macrophage function by reducing the capacity of macrophages to express LPSinduced pro-inflammatory cytokines [1]. Treatment of macrophages with T. spiralis ES alone enhances anti-inflammatory cytokine expression and molecular studies showed that T. spiralis ES products significantly inhibited LPS-induced nuclear translocation of nuclear factor-kB and phosphorylation of extracellular signal-regulated protein kinase and p38 mitogen-activated protein kinase [1]. These results suggest worm ES products participate in immune evasion by suppressing macrophage functions and pro-inflammatory responses.

Since NO plays an important role in inducing intestinal physiology and inflammation during *T. spiralis* infection [57], abrogation of LPS-induced iNOS expression in macrophages by *T. spiralis* ES products reflects a potent immunomodulatory strategy. *T. spiralis* infection has been reported to downregulate iNOS gene expression [9], suggesting parasite and parasite-derived molecules may modulate host immune responses to allow survival. Since parasite
ES products enhance anti-inflammarotory cytokine expression, ES products may modulate balance of pro- and anti-inflammatory responses thus providing parasite with a favourable environment for infection and growth [1]. *T. spiralis* ES products also induce Th2 type responses in rat bone-marrow derived dendritic cells *in vitro* and *in vivo*, while *in vivo* T-cell priming generated a mixed Th1/Th2 cytokine response [38]. Recently, it was reported that recombinant paramyosin from *T. spiralis* bound complement components C8 and C9, inhibited C9 polymerization and prevented formation of MAC [103]. Recombinant *T. spiralis* paramyosin also inhibited lysis of rabbit erythrocytes via alternative complement pathway [103].

Parasites have evolved a variety of ways to evade antimicrobial responses of macrophages and other immune cells. To understand how *T. carassii* interacts with fish macrophages, I examined the effects of *T. carassii* hsp70 on goldfish macrophage functions. I observed that Tcahsp70 activated goldfish macrophages and stimulated the production of pro-inflammatory cytokines [73]. Parasite hsp70 also up-regulated the expression of iNOS isoforms A and B, and induced a strong nitric oxide response of goldfish macrophages [73]. Trypanosomes are known to possess all major classes of Hsps [33] but the precise roles of trypanosome hsps in immune modulation are yet to be investigated. However, hsp70 of protozoan parasites have been shown to be immunogenic and induced protection in immunization studies [7, 46].

A functioning immune response depends on its signalling network. In response to an infection, host signalling molecules, such as cytokines and

chemokines, and antimicrobial molecules are produced by macrophages. Parasites interfere with immune cell signalling in many ways. For example, *Leishmania* modulate host cells by interfering with signalling mechanisms leading to impairment of immune cell functions [37, 75]. This effect is dependent on Leishmania gp63 [21, 35]. I showed that rTcagp63 down regulated Aeromonas salmonicida and recombinant goldfish TNF α 2-induced production of reactive oxygen and nitrogen intermediates. Macrophages treated with rTcagp63 also exhibited significant reduction in the expression of cytokines TNF α -1 and TNF α -2, and iNOS-A genes. Recombinant Tcagp63 was internalized by goldfish macrophages and altered phospho-tyrosine protein pattern suggesting that Tcagp63 may act by altering the signalling events, which are important in downstream macrophage antimicrobial and other cytokine-induced functions. T. brucei gp63 is required for removing VSGs from the surface of bloodstream parasites [39, 55], while T. cruzi gp63 may participate in infection of host cells [25]. However, none of trypanosome gp63 molecule reported so far have been shown to modulate immune cell functions.

T. cruzi cruzain a cysteine protease was recently reported to mediate immune evasion by cleaving signalling molecule NF-kB thereby inhibiting host cell activation [29]. Cruzain also inhibits cytokine expression in *T. cruzi*-infected macrophages treated with LPS suggesting impairment of signalling mechanisms required for macrophage activation. Cruzain gene knockout parasites activate macrophages, induced ikB phosphorylation, NF-kB nuclear translocation, cytokine expression and were unable to survive within macrophages [29]. These results further lend support to reports that parasite proteases mediate immune evasion.

It appears that T. carassii possess molecules that can induce pro- and antiinflammatory responses presenting the parasite with an excellent strategy to manipulate host inflammatory responses to their benefit. This assertion is supported by results described in this thesis. For example, I observed significant and parallel increases in mRNA levels of genes encoding pro-inflammatory and anti-inflammatory cytokines in the different tissues of *T. carassii*-infected fish compared to non-infected control [71]. Expressions of majority of genes encoding pro- and anti-inflammatory cytokines were up-regulated during the acute phase of infection (days 7 to 21 post-infection). The mRNA levels of these cytokines returned to normal levels or were down-regulated during the elimination phase of infection (days 28 to 56). These results suggest Th1/Th2-like responses may be important for controlling *T. carassii* infection in the goldfish. Alternatively, parasite and parasite-derived molecules may induce pro- and anti-inflammatory responses to suppress specific or general immune responses. I favour the latter and propose that immunosuppression could be a major evasion strategy used by T. carassii.

In vertebrates, the complement system is an important part of both innate and adaptive immunity. In innate immunity, complement system recognizes the invading pathogen, communicate with and activate the adaptive immunity [67, 99, 100]. It involves cascades of circulating and cell surface-bound proteins which are activated sequentially through three pathways known as the classical, lectin and

alternative pathways. The classical and lectin complement pathways are initiated by C1q and MBL. The alternative pathway is activated after binding of complement component C3b to microbial surfaces. The end product of all complement activation pathways is the coating of the invading organism with C3 and assembly of the membrane attack complex leading to destruction of the pathogen [56]. Regulators of complement activation down-regulate complement action at several steps in the complement cascades and prevent deleterious effects of complement activation on host cells. Because of the significant evolutionary pressure exerted on parasites by host complement system, they have evolved mechanisms to escape the deleterious effects of complement. The investigation and characterization of complement evasion strategies used by pathogens is an expanding area of research. Generally, pathogens avoid complement lysis by evading recognition by complement activators, by expressing soluble and membrane-bound molecules that inhibit or degrade complement components or by recruiting complement regulators [23, 50, 77, 97, 103]. These mechanisms have led to a multitude of complement evasion strategies by different pathogens [8, 12, 22, 32, 44, 51, 104].

Trypanosomes are resistant to complement-mediated lysis (CML) and their complement evasion strategies have been fairly studied. In *T. brucei*, it has been reported that VSG and gp63 proteases play roles in protection of parasites from CML [30, 36, 55, 95]. *T. cruzi* possess variety of complement regulatory molecules, which allow the parasite to evade CML [68, 69]. Complement-mediated lysis of trypanosomes has been reported to occur primarily via the

alternative complement pathway [52, 79]. The classical pathway appeared to be ineffective in parasite lysis, although it is believed to amplify the lytic effects of the alternative complement pathway. Calreticulin, a secreted and cell surface-bound protein is thought to be partly responsible for the inability of the classical complement pathway to lyse trypanosomes [31, 72]. Despite the importance of calreticulin as an immune evasion molecule, only *T. cruzi* calreticulin (TcCRT) has been fairly characterized among protozoa parasites. TcCRT binds human C1q and specifically inhibits classical pathway of complement [31]. I reported that *T. carassii* calreticulin is identified in both the ES and surface protein fractions and is abundantly expressed on the parasite surface [72]. Recombinant *T. carassii* calreticulin bound first complement component, C1q of the goldfish and inhibited C1q-dependent lysis, suggesting that TcaCRT may be used by the parasite to inhibit host classical complement pathway *in vivo*.

8.1 Future direction

Further characterization of *T. carassii* antigens in ES and surface protein fractions is necessary to determine identity of these antigens and investigate their role in host-parasite interactions. Although immunoblot of surface proteins separated by 2D-electrophoresis and probed with immune sera presents technical challenges, such approach may reveal candidate antigens expressed on the parasite surface. Another approach that may accelerate identification of protective antigens of *T. carassii* would be to compare leech stage parasites with those in fish using proteomics and transcriptomics. Two-dimensional electrophoresis, cDNA library construction and suppression subtraction hybridization may be used for such large-scale studies. Results of these studies can be validated using quantitative real time PCR, and western or northern blotting techniques. A European based collaborator will be required for these studies since cultivation of leech stage parasite is not allowed in Canada. (*T.carassii* is not indigenous in North America).

The expression analysis of pro- and anti-inflammatory cytokines presented in this thesis is the first step towards understanding goldfish-T. carassii interactions. It may be necessary to validate the results reported at protein levels to further understand the type of immune response generated in T. carassiiinfected fish. T. carassii infection induced expression of IFN γ and TNF α in goldfish. TNFα has been reported to cause lysis of *T. brucei* and *T. cruzi in vitro*, while treatment of *T. brucei*-infected mice with anti-TNFα antibodies increased parasitemia in the blood and tissues [58, 74]. On the other hand, IFN γ acts as a growth factor and promotes the growth of African trypanosomes [2, 3, 4]. Like T. brucei, T. carassii is an extracellular parasite and it would be interesting to investigate the effects of goldfish TNF α on this parasite. If goldfish TNF α cause lysis of *T. carassii*, this observation will further strengthen the functional significance of Tcagp63 as parasite-derived molecule that mediates immune evasion by dampening pro-inflammatory responses. That both pro- and antiinflammatory responses are produced in T. carassii-infected goldfish supports the idea that parasite may use immunosuppression as an evasion strategy.

Further investigation of innate immunity of goldfish to *T. carassii* infection should be undertaken because of heavy reliance of fish on non-specific immunity.

A more detailed examination of parasite interaction with goldfish immune cells in vitro should provide new insights into host-parasite interactions. I have observed that T. carassii induces respiratory burst response of monocytes but fails to induce nitric oxide response of goldfish macrophages *in vitro* studies. I reported that T. *carassii* induced iNOS gene expression *in vivo* [71], while it is susceptible to NO in vitro [86]. The reasons for these discrepancies are not clear at this point, however, it would be interesting to examine the effects of activated macrophages on parasites. From my observations, it is clear that T. carassii ES products contain at least one molecule that can induce significant pro-inflammatory responses and at least one molecule that can down-regulate such responses. Effects of the ES products on goldfish monocyte and macrophages functions need to be investigated. Although, I could not detect Tcagp63 in parasite ES products there is evidence for extracellular release of gp63 [45, 61]. ES fraction should be evaluated for protease activity by examining its proteolytic effects on immune molecules of the goldfish.

In mice infected with the worm *Heligmosomoides polygyrus*, worm calreticulin skews immune response toward a Th2 type phenotype [84], suggesting that parasite CRTs may possess ability to induce a Th2 type response. This hypothesis may be tested by examining cytokine gene expression in macrophages treated with rTcaCRT. Alternatively, goldfish can be immunized with rTcaCRT in the presence and absence of adjuvants, and antibody responses as well as cytokine gene expression in the different organs examined. The result of this study should provide further support for the observation that parasite induced a parallel up-regulation of pro- and anti-inflammatory cytokines in the different tissues of infected goldfish.

The general biology and functions of parasite CRTs warrants further studies. It has been reported that CRT can inhibit perforin-mediated lysis at physiologically relevant concentration leading to proposition that CRT may stabilize cell membranes to prevent formation of polyperforin pore [34]. This indicates CRT can regulate lytic functions and it would be interesting to test whether parasite CRT can perform this function or use this as a strategy for immune modulation. Additionally, CRT binds azurocidin, a secretory protein in neutrophils that has been shown to induce cytokine production in monocytes. Azurocidin-induced cytokine production is inhibited by anti-CRT antibodies suggesting that interaction of cell surface CRT with azurocidin is required for monocyte activation [70]. It is possible that parasite CRTs generally use their chaperone activity to interact with other proteins and indirectly participate in immune cell activation.

Any potential protective antigen would need to be evaluated for its ability to protect experimentally challenged animals in immunization studies. A single antigen or combination of antigens may be evaluated for their protective efficacy. Immunization studies with recombinant antigens or DNA may be considered since DNA vaccination may induce better level of protection. Experimental vaccination with recombinant *Necator americanus* CRT in absence of adjuvant reduced worm burden by approximately 49% in the lungs of infected mice compared to control mice [101]. *Leishmania* parasite hsp70 was shown to be protective in immunization studies [46]. Recently, a cocktail of *Leishmania* gp63 and hsp70 was shown to induce significant levels of protection in experimentally challenged mice [47]. The immunized animals generated high levels of IgG2a and expressed high levels of IFN γ and IL-2 but low levels of IL-4 and IL-10 [47]. These results suggest generation of a protective Th1 response and suppression of Th2 type immune response. I [73] reported that Tcahsp70 stimulated inflammatory responses of goldfish macrophages while Tcagp63 down-regulates inflammatory responses. It would be interesting to test these antigens separately and in combination in immunization studies. Also, TcaCRT another parasite chaperone may be tested together with Tcagp63 in immunization studies.

Although several hundred antimicrobial peptides have been described from diverse species, there is little information on antimicrobial peptides in fish and their effects on *T. carassii* are yet to be elucidated. These small molecules have characteristic cationic charge and amphipathic conformation, and play crucial roles in immunity [41]. Antimicrobial peptides have been shown to kill *T. brucei* and *T. cruzi* trypomastigotes *in vitro* [40, 60, 5]. To date, only a few fish antimicrobial peptides have been identified [17, 18, 20, 76]. I have attempted to clone goldfish cathelicidin using primers designed to homolgous regions of fish cethelicidin sequences with no success thus far. The effects of these antimicrobial peptides can be tested on *in vitro* grown *T. carassii*. The progress in genome sequencing should accelerate identification of more fish peptides that can be tested. Since these molecules exhibit sequence conservation, the effects of mammalian antimicrobial peptides may also be tested on *T. carassii*.

8.2 Proposed model of *T. carassii* evasion mechanisms

Based on the work described in this thesis, I propose two mechanisms of immune evasion by *T. carassii*. For the first mechanism, *T. carassii* may use hsp70 and other parasite molecules capable of inducing inflammatory responses to induce a Th1 type immune response (Fig 8.1) early during infection (from days 1-21 pi). This will results in down-regulation of Th2 cytokine secretion and antibody production, which is required for parasite elimination. This environment will favour parasite establishment and maintenance of infection. TcaCRT is also expected to function during early stages of infection to prevent classical complement pathway-mediated lysis. Later in infection (between days 21-28 dpi), parasites may produce Tcagp63 to dampen inflammatory responses and prevent inflammation-associated tissue damage to the host. However, this is to the detriment of parasites, as this would allow host to generate Th2 type responses and produce parasite specific antibodies that will eliminate the parasites.

In the second mechanism, both Tcahp70 and Tcagp63 may be produced during early stages of infection (days 1-21 pi) (Fig. 8.1) leading to mixed Th1/Th2 responses causing general immunosuppression. This will also enhance parasite establishment in the host. Both mechanisms will ensure that parasites maintain a chronic infection to ensure their transmission to a new host.

The roles of TcaCRT in Th1/Th2 responses deserve further investigation. Calreticulins have anti-complement effects resulting in decreased chomotaxis and activation of immune cells because of decrease in the generation of anaphylatoxins, C3a and C5a. Thus, TcaCRT may play additional roles early during infection by contributing to suppression of inflammatory responses. It is possible that parasites modulate concentration of these molecules within the host to regulate the balance of Th1/Th2 responses. The timing of production and dose of these antigens may be important in the cross-regulation of host Th1/Th2 responses.

8.3 Summary

The work presented in this thesis showed that *T. carassii* uses a variety of mechanisms to evade or modulate goldfish immune responses. *T. carassii*-infection induced a parallel expression of both pro- and anti-inflammatory cytokines in goldfish. The profile of cytokine induced in *T. carassii*-infected fish suggests that a balance of Th1 and Th2 responses may play roles in controlling trypanosome infection in fish.

T. carassii ES products contain several immunogenic molecules, among which are Tcahsp70, TcaCRT and Tcagp63. I showed that rTcahsp70 can stimulate inflammatory responses while rTcagp63 may down-regulate inflammatory responses. These observations lend further credibility to assertion that a balance Th1/Th2 responses are important in controlling *T. carassii* infection. Also, TcaCRT a surface expressed molecule may be partly responsible for inefficient lysis of parasite by classical complement pathway. Recombinant TcaCRT bound and inhibited C1q-dependent lysis. It is clear that *T. carassii* uses a variety of strategies to persist in the host and ensure continuous transmission.

This host-parasite model system presents some challenges such as lack of complete genome information to accelerate identification of parasite genes that may be involved in immune evasion as well as host factors that may play roles in induction of protective immunity. Thus, considerable time and effort was required to generate reagents such as recombinant proteins and antibodies. Additionally, inability to culture the leech stage parasites in Canada prevents the use of largescale comparative proteomic and transcriptomic studies to identify parasite evasion molecules that are differentially regulated. Nevetheless, this model system presents an opportunity to study protective immune responses in ectothermic vertebrates and results obtained using this system will provide valuable information into an host-parasite interaction that occurs naturally. More research using this model system should be encouraged to advance our knowledge of immune evasion strategies used by this protozoan of ectothermic vertebrates.



T. carassii infection of goldfish



8.4 References

- Bai X., X. Wu, X. Wang, Z. Guan, F. Gao, J. Yu, L. Yu, B. Tang, X. Liu, Y. Song, X. Wang, B. Radu, P. Boireau, F. Wang, M. Liu. 2011. Regulation of cytokine expression in murine macrophages stimulated by excretory/secretory products from *Trichinella spiralis in vitro*. Molecular and Cellular Biochemistry DOI 10.1007/s11010-011-1046-4.
- Bakhiet, M., T. Olsson, J. Mhlanga, P. Buscher, N. Lycke, P.H. van der Meide, and K. Kristensson. 1996. Human and rodent IFN-γ as a growth factor for *Trypanosoma brucei brucei*. European Journal of Immunology 26: 1359-1364.
- Bakhiet, M., T. Olsson, C. Edlund, B. Hojeberg, K. Holmberg, J. Lorentzen, and K. Kristensson. 1993. *Trypanosoma brucei brucei* derived factors that triggers CD8⁺ lymphocytes to interferon gamma secretion: purification, characterization and protective effects *in vivo* by treatment with monoclonal antibodies against the factor. Scandinavian Journal of Immunology **37**: 165-178
- Bakhiet, M., T. Olsson, P. van der Meide, and K. Kristensson. 1990. Depletion of CD8+ T cells suppresses growth of *Trypanosoma brucei brucei* and interferon-gamma production in infected rats. Clinical and Experimental Immunology 81: 195-199.
- Barr, S. C., D. Rose, and J. M. Jaynes. 1995. Activity of lytic peptides against intracellular *Trypanosoma cruzi* amastigotes in vitro and parasitemias in mice. Journal of Parasitology 81: 974-978.
- Baruch, D. I., B. L. Pasloske, H. B. Singh, X. Bi, X. C. Ma, M. Feldman, T. F. Taraschi, and R. J. Howard. 1995. Cloning the *P. falciparum* gene encoding PfEMP1, a malarial variant antigen and adherence receptor on the surface of parasitized human erythrocytes. Cell 82: 77–87.
- Behr, C., J. L. Sarthou, J. F. Trape, M. H. Dat, J. C. Michel, G. Aribot, A. Dieye, J. M. Claverie, P. Druhihle, and P. Dubois. 1992. Antibodies and reactive T cells against the malaria heat shock protein Pf72/Hsp70-1 and

derived peptides in individuals continuosly exposed to *Plasmodium falciparum*. Journal of Immunology **149:** 3321-3330.

- Bernet, J., J. Mullick, A. K. Singh, A. Sahu. 2003. Viral mimicry of the complement system. Journal of Biosciences 28: 249–264.
- Bian, K., Y. Harari, M. Zhong, M. Lai, G. Castro, N. Weisbrodt, and F. Murad. 2001. Down-regulation of inducible nitric-oxide synthase (NOS-2) during parasite-induced gut inflammation: a path to identify a selective NOS-2 inhibitor. Molecular Pharmacology 59: 939–947.
- Biggs, B. A., L. Gooze. K. Wycherley, W. Wollish, B. Southwell, J. H. Leech, and G. V. Brown. 1991. Antigenic variation in *Plasmodium falciparum*. Proceedings of the National Academy of Science USA 88: 9171-9174.
- Blaxter, M. L., A. P. Page, W. Rudin, and R. M. Maizels. 1992. Nematode surface coats: actively evading immunity. Parasitology Today 8: 243–247.
- Blom A. M. 2004. Strategies developed by bacteria and virus for protection from the human complement system. Scandinavian Journal of Clinical Laboratory Investigation 64: 479–496.
- 13. Borst, P. 2002. Antigenic variation and allelic exclusion. Cell 109: 5-8.
- Bull, P. C., and K. Marsh. 2002. The role of antibodies to *Plasmodium falciparum*-infected-erythrocyte surface antigens in naturally acquired immunity to malaria. Trends in Microbiology. **10:** 55–58.
- Buscaglia, C. A., V. A. Campo, A. C. Frasch, and J. M. Di Noia. 2006. *Trypanosoma cruzi* surface mucins: host-dependent coat diversity. Nature Review Microbiology 4: 229-236.
- Cancela, M., C. Carmona, S. Rossi, B. Frangione, F. Goni, P. Berasain. 2004. Purification, characterization, and immunolocalization of paramyosin from the adult stage of *Fasciola hepatica*. Parasitology Research **92:** 441–448.
- Chang, C., O. Pleguezuelos, Y. Zhang, J. Zou, P. Nic, and C. J. Secombes. 2006. Two cathelicidin genes are present in both rainbow trout (*Oncorhynchus mykiss*) and atlantic salmon (*Salmo salar*). Antimicrobial

Agents and Chemotherapy 50: 185-195.

- Chang, C., O. Pleguezuelos, Y. Zhang, J. Zou, and C. J. Secombes. 2005. Identification of a novel cathelicidin gene in the rainbow trout *Oncorhynchus mykiss*. Infection and Immunity **73**: 5053-5064.
- Clegg, J. A., S. R. Smithers, and R. J. Terry. 1971. Acquisition of human antigens by *Schistosoma mansoni* during cultivation *in vitro*. Nature 232: 653–654.
- Cole, A. M., R. O. Darouiche, D. Legarda, N. Connell, and G. Diamond.
 2000. Characterization of a fish antimicrobial pepetide: gene expression, sub-cellular localization, and spectrum of activity. Antimicrobial Agents and Chemotherapy 44: 2039-2045.
- Contreras, I., M. A. Gomez, O. Nguyen, M. T. Shio, and R. W. McMaster.
 2010. *Leishmania*-induced inactivation of the macrophage transcription factor AP-1 is mediated by the parasite metalloprotease GP63. PLos Pathogen 6: e1001148.
- Cooper, N. R. 1991. Complement evasion strategies of microorganisms. Immunology Today 12: 327–331.
- 23. Corley, L.S., and M. R. Strand. 2003. Evasion of encapsulation by the polyembryonic parasitoid *Copidosoma floridanum* is mediated by a polar body-derived extraembryonic membrane. Journal of Invertebrate Pathology 83: 86–89.
- 24. Cotton, J. A., J. K. Beatty, and A. G. Buret. 2011. Host parasite interactions and pathophysiology in *Giardia* infections. International Journal of Parasitology 41: 925-933.
- 25. Cuevas, I. C., J. J. Cazzulo, and D. O. Sanchez. 2003. gp63 homologues in *Trypanosoma cruzi*: surface antigens with metalloprotease activity and a possible role in host cell infection. Infection and Immunity **71**: 5739-5749.
- Damian, R.T. 1997. Parasite immune evasion and exploitation: reflections and projections. Parasitology 115: S169–S175.
- 27. Damian, R. T., N. D. Greene, and W. J. Hubbard. 1973. Occurrence of mouse alpha-2-macroglobulin antigenic determinants on *Schistosoma*

mansoni adults, with evidence on their nature. Journal of Parasitology. **59**: 64–73.

- Deng, J., D. Gold, P. T. LoVerde, and Z. Fishelson. 2003. Inhibition of the complement membrane attack complex by *Schistosoma mansoni* paramyosin. Infection and Immunity **71:** 6402–6410.
- Doyle, P. S., Y. M. Zhou, I. Hsieh, D.C. Greenbaum, J. H. McKerrow and J. C. Engel. 2011. The *Trypanosoma cruzi* protease Cruzain mediates immune evasion. PLoS Pathogen 7: e1002139.
- 30. El-Sayed, N. M., and J. E. Donelson. 1997. African trypanosomes have differentially expressed genes encoding homologues of the *Leishmania* GP63 surface protease. Journal of Biological Chemistry 272: 26742-26748.
- 31. Ferreira, V., C. Valck, G. Sanchez, A. Gingras, S. Tzima, M. C. Molina, R. Sim, W. Schwaeble, and A. Ferreira. 2004. The classical activation pathway of the human complement system is specifically inhibited by calreticulin from *Trypanosoma cruzi*. Journal of Immunology **172**: 3042-3050.
- 32. Finlay, B. B., and G. McFadden. 2006. Anti-immunology: evasion of the host immune system by bacterial and viral pathogens. Cell **124**: 767–782.
- Folgueira, C., and J. M. Requena. 2007. A postgenomic view of the heat shock proteins in kinetoplastids. FEMS Microbiology Review 31: 359-377.
- Fraser, S. A., R. Karimi, M. Michalak, and D. Hudig. 2000. Perforin lytic activity is controlled by calreticulin. Journal of Immunology 164: 4150-4155
- 35. Gomez, M. A., I. Contreras, M. Halle, M. L. Tremblay, R. W. McMaster, and M. Olivier. 2009. *Leishmania* GP63 alters host signalling through cleavage-activated protein tryrosine phosphatases. Science Signal 2: ra58.
- 36. Grandgenett, P. M., K. Otsu, H. R. Wilson, M. E. Wilson, and J. E. Donelson. 2007. A function for a specific zinc metalloprotease of African trypanosomes. PLoS Pathogens 3: 1432-1445.

- Gregory, D. J., and M. Olivier. 2005. Subversion of host cell signalling by the protozoan parasite *Leishmania*. Parasitology 130: S27-S35
- Gruden-Movsesijan, A., N. Ilic, M. Colic, I. Majstorovic, S. Vasilev, I. Radovic, and L. J. Sofronic-Milosavljevic. 2011. The impact of *Trichinella spiralis* excretory-secretory products on dendritic cells. Comparative Immunology, Microbiology and Infectious Diseases 34: 429-439.
- 39. Gruszynski, A. E., F. J. van Deursen, M. C. Albareda, A. Best, K. Chaudhary, L. J. Cliffe, et al. 2006. Regulation of surface coat exchange by differentiating African trypanosomes. Molecular and Biochemical Parasitology 147: 211-223.
- 40. Haines, L. R., R. E. W. Hancock, and T. W. Pearson. 2003. Cationic antimicrobial peptide killing of African trypanosomes and *Sodalis* glossinidus, a bacterial symbiont of the insect vector of sleeping sickness. Vector-borne and Zoonotic Diseases 3: 175-186.
- 41. Hancock, R. E. W., and G. Diamond. 2000. The role cationic antimicrobial peptides in innate host defences. Trends in Microbiology **8:** 402-410
- 42. Handunnetti, S. M., K. N. Mendis, and P. H. David. 1987. Antigenic variation of cloned *Plasmodium fragile* in its natural host Macaca sinica. Sequential appearance of successive variant antigenic types. Experimental Medicine 165: 1269-1283.
- 43. Horta, M. F., F. J. Ramalho-Pinto, and M. Fatima. 1991. Role of human decay-accelerating factor in the evasion of *Schistosoma mansoni* from the complement-mediated killing in vitro. Journal of Experimental Medicine 174:1399–1406.
- 44. Inal, J. M. 2004. Parasite interaction with host complement: beyond attack regulation. Trends Parasitology 20: 407–412.
- 45. Jaffe, C. L., and D. M. Dwyer. 2003. Extracellular release of the surface metalloprotease gp63, from *Leishmania* and insect trypanosomatids. Parasitology Research **91**: 229–237.

- 46. Kaur, J., T. Kaur, and S. Kaur. 2011. Studies on the protective efficacy and immunogenicity of Hsp70 and Hsp 83 based vaccine formulations in *Leishmania donovani* infected BALB/c mice. Acta Tropica **119**: 50-56.
- 47. Kaur, T., Sobti, R. C., and S. Kaur. 2011. Cocktail of gp63 and Hsp70 induces protection against *Leishmania donovani* in BALB/c mice. Parasite Immunology **33**: 95-103.
- 48. Kemp, W. M., S. C. Merritt, M. S. Bogucki, J. G. Rosier, and J. R. Seed. 1977. Evidence for adsorption of heterospecific host immunoglobulin on the tegument of *Schistosoma mansoni*. Journal of Immunology **119**: 1849– 1854.
- 49. Kemp, W. M., R. T. Damian, N. D. Greene, and W. B. Lushbaugh. 1976. Immunocytochemical localization of mouse alpha 2-macroglobulin like antigenic determinants on *Schistosoma mansoni* adults. Journal of Parasitology **62:**413–419.
- Kotwal, G. J., and B. Moss. 1988. Vaccinia virus encodes a secretory polypeptide structurally related to complement control proteins. Nature 335: 176–178.
- Kraiczy, P., and R. Wurzner. Complement escape of human pathogenic bacteria by acquisition of complement regulators. Molecular Immunology 43: 31–44.
- Krettli, A. U., P. Weisz-Carrington, and R. S. Nussenzweig. 1979.
 Membrane- bound antibodies to bloodstream *Trypanosoma cruzi* in mice: strain differences in susceptibility to complement-mediated lysis. Clinical Experimental Immunology **37:** 416-423.
- Kyes, S., P. Horrocks, and C. Newbold. 2001. Antigenic variation at the infected red cell surface in malaria. Annual Review Microbiology 55: 673-707.
- 54. Laclette, J. P., P. J. Skelly, M. T. Merchant, C. B. Shoemaker. 1995. Aldehyde fixation dramatically alters the immunolocalization pattern of paramyosin in platyhelminth parasites. Experimental Parasitology 81: 140–143.

- LaCount, D. J., A. E. Gruszynski, P. M. Grandgenett, J. D. Bangs, and J. E. Donelson. 2003. Expression and function of the *Trypanosoma brucei* major surface protease (GP63) genes. Journal of Biological Chemistry 278: 24658-24664.
- Lambris, J. D., D. Ricklin, and B. V. Geisbrecht. 2008. Complement evasion by human pathogens. Nature Reviews Microbiology 6: 132-142.
- 57. Lawrence, C. E., J. C. Paterson, X. Q. Wei, F. Y. Liew, P. Garside, and M. W. Kennedy. 2000. Nitric oxide mediates intestinal pathology but not immune expulsion during *Trichinella spiralis* infection in mice. Journal of Immunology **164:** 4229–4243.
- 58. Magez, S., M. Geuskens, A. Beschin, H. del Favero, H. Verschueren, R. Lucas, E. Pays, and P. De Baetselier. 1997. Specific uptake of tumor necrosis factor-α is involved in growth control of *Trypanosoma brucei*. Journal of Cell Biology **137**: 715-727.
- Maizels, R. M. 2009. Parasite immunomodulation and polymorphisms of the immune system. Journal of Biology 8: 62.
- McGwire, B. S., C. L. Olson, B. F. Tack, and D. M. Engman. 2003. Killing of African trypanosomes by antimicrobial peptides. Journal of Infectious Diseases 188: 146-152.
- 61. McGwire, B. S., W. A. O'Connell. K. P. Chang, and D. M. Engman. 2002. Extracellular release of the glycosylphosphatidylinositol (GPI)-linked *Leishmania* surface metalloprotease, gp63, is independent of GPI phospholipolysis: implications for parasite virulence. Journal of Biological Chemistry 277: 8802-8809.
- McLean, S.A., C. D. Pearson, and R. S. Phillips. 1982. *Plasmodium chabaudi:* antigenic variation during recrudeseent parasitaemias in mice. Experimental Parasitology 54: 296-302.
- Mowatt, M. R., A. Aggarwal, and T. E. Nash. 1991. Carboxyterminal sequence conservation among variant-specific surface proteins of *Giardia lamblia*. Molecular and Biochemical Parasitology. 49: 215–227.

- 64. Muhlschlegel, F., L. Sygulla, P. Frosch, P. Massetti, and M. Frosch. 1993.
 Paramyosin of *Echinococcus granulosus*: cDNA sequence and characterization of a tegumental antigen. Parasitology Research **79:** 660– 666.
- 65. Nash, T.E. 2002. Surface antigenic variation in *Giardia lamblia*. Molecular Microbiology 45: 585–590.
- 66. Nash, T. E., H. D. Lujan, M. R. Mowatt, and J. T. Conrad. 2001. Variantspecific surface protein switching in *Giardia lamblia*. Infection and Immunity 69: 1922–1923.
- 67. Nonaka, M., and F. Yoshizaki. 2004. Evolution of the complement system. Molecular Immunology 40: 897–902.
- Norris, K. A. 1998. Stable transfection of *Trypanosoma cruzi* epimastigotes with the trypomastigote-specific complement regulatory protein cDNA confers complement resistance. Infection and Immunity 66: 2460-2465.
- Norris, K. A., J. E. Schrimpf, and M. J. Szabo. 1997. Identification of the gene family encoding the 160-kilodalton *Trypanosoma cruzi* complement regulatory protein. Infection and Immunity 65: 349-357.
- Okuyama, Y., J. Cho, Y. Nakajima, K. Homma, K. Sekimizu, and S. Natori. 2004. Binding between azurocidin and calreticulin: its involvement in the activation of peripheral monocytes. Journal of Biochemistry 135: 171-177.
- Oladiran, A., D. Beauparlant, and M. Belosevic. 2011. The expression analysis of inflammatory and antimicrobial genes in the goldfish (*Carassius auratus* L.) infected with *Trypanosoma carassii*. Fish and Shellfish Immunology **31:** 606-613.
- Oladiran, A., and M. Belosevic. 2010. *Trypanosoma carassii* calreticulin binds host complement component C1q and inhibits classical complement pathway-mediated lysis. Developmental and Comparative Immunology 34: 396-405.
- 73. Oladiran, A., and M. Belosevic. 2009. Trypanosoma carassii hsp70

increases expression of inflammatory cytokines and chemokines in macrophages of the goldfish (*Carassius auratus* L.). Developmental and Comparative Immunology **33:** 1128–1136.

- 74. Olivares-Fontt, E., P. De Baetselier, C. Heirman, K. Thielemans, R. Lucas, and B. Vray. 1998. Effects of granulocyte-macrophage colony-stimulating factor and TNFα on *Trypanosoma cruzi* trypomastigotes. Infection and Immunity **66**: 2722-2727.
- 75. Olivier, M., D. J. Gregory, and G. Forget. 2005. Subversion mechanisms by which *Leishmania* parasites can escape the host immune response: a signalling point of view. Clinical Microbiology Review **18**: 293-305.
- 76. Oren, Z., and Y. Shai. 1996. A class of highly potent antibacterial peptides derived from paraxin a pore forming peptide isolated from Moses sole fish *Pardachirus marmoratus*. European Journal of Biochemistry 237: 303-310.
- 77. Parizade, M., R. Arnon, P. J. Lachmann, and Z. Fishelson. 1994.
 Functional and antigenic similarities between a 94-kD protein of *Schistosoma mansoni* (SCIP-1) and human CD59. Journal Experimental Medicine 179: 1625–1636.
- 78. Pimenta, P. F., P. P. da Silva, and T. Nash. 1991. Variant surface antigens of *Giardia lamblia* are associated with the presence of a thick cell coat: thin section and label fracture immunocytochemistry survey. Infection and Immunity **59**: 3989–3996.
- 79. Plouffe, D. A., and M. Belosevic. 2004. Enzyme treatment of *Trypanosoma danilewskyi* (Laveran &Mesnil) increases its susceptibility to lysis by the alternative complement pathway of goldfish, *Carassius auratus* (L.). Journal of Fish Disease 27: 277-285.
- Prucca, C. G., and H. D. Lujan. 2009. Antigenic variation in *Giardia lamblia*. Cellular Microbiology 11: 1706-1715.
- Reeder, J. C., and G. V. Brown. 1996. Antigenic variation and immune evasion in Plasmodium falciparum malaria. Immunology and Cell Biology 74: 546-554.

- Reiner, S. L., and R. M. Locksley. 1995. The regulation of immunity to Leishmania major. Annual Review of Immunology 13: 151–177.
- Roberts, D. J., A. G. Craig, A. R. Berendt, R. Pinches, G. Nash, K. Marsh, and C. I. Newbold. 1992. Rapid switching to multiple antigenic and adhesive phenotypes in malaria. Nature 357: 689-692.
- 84. Rzepecka, J., S. Rausch, C. Klotz, C. Schnoller, T. Kornprobst, J. Hagen, R. Ignatius, R. Lucius, and S. Hartmann. 2009. Calreticulin from the intestinal nematode *Heligmosomoides polygyrus* is a Th2-skewing protein and interacts with murine scavenger receptor-A. Molecular Immunology 46: 1109-1119.
- Sacks, D., and A. Sher. 2002. Evasion of innate immunity by parasitic protozoa. Nature Immunology 3: 1041–1047.
- 86. Saeij, J. P. J., W. B. Van Muiswinkel, A. Groeneveld, and G. F. Wiegertjes. 2002. Immune modulation by fish kinetoplastid parasites: a role for nitric oxide. Parasitology 124: 77-86.
- 87. Santoro, F., M. A. Ouaissi, J. Pestel, and A. Capron. 1980. Interaction between *Schistosoma mansoni* and the complement system: binding of C1q to schistosomula. Journal of Immunology **124**: 2886–2891.
- 88. Schmid-Hempel, P. 2009. Immune defence, parasite evasion strategies and their relevance for 'macroscopic phenomena' such as virulence.
 Philosophical Transactions of The Royal Society of London. 364: 85-98.
- 89. Schmid-Hempel, P. 2008. Parasite immune evasion: a momentous molecular war. Trends in Ecology and Evolution **23**: 318-326.
- 90. Segura, M., Z. Su, C. Piccirillo, and M. M. Stevenson. 2007. Impairment of dendritic cell function by excretory-secretory products: A potential mechanism for nematode-induced immunosuppression. European Journal of Immunology **37:** 1887-1904.
- Shelby, K.S., O. A. Adeyeye, B. M. Okot-Kotber, and B. A. Webb. 2000. Parasitism-linked block of host plasma melanization. Journal of Invertebrate Pathology 75: 218–225.
- 92. Sher, A., B. F. Hall, and M. A. Vadas. 1978. Acquisition of murine major

histocompatibility complex gene products by schistosomula of *Schistosoma mansoni*. Journal of Experimental Medicine. **148:**46–57.

- 93. Simpson, A. J., D. Singer, T. F. McCutchan, D. L. Sacks, and A. Sher. 1983. Evidence that schistosome MHC antigens are not synthesized by the parasite but are acquired from the host as intact glycoproteins. Journal of Immunology 131: 962–965.
- Tarleton, R. L., and W. M. Kemp. 1981. Demonstration of IgG-Fc and C3 receptors on adult *Schistosoma mansoni*. Journal Immunology **126**: 379– 384.
- 95. Taylor, J. E., and G. Rudenko. 2006. Switching trypanosome coats: What's in the wardrobe? Trends in Genetics **22:** 614-620.
- 96. Torpier, G., A. Capron, and M. A. Ouaissi. 1979. Receptor for IgG(Fc) and human β2-microglobulin on *S. mansoni* schistosomula. Nature 278: 447–449.
- 97. Vanderplasschen A, E. Mathew, M. Hollinshead, R. B. Sim, and G. L. Smith. 1998. Extracellular enveloped vaccinia virus is resistant to complement because of incorporation of host complement control proteins into its envelope. Proceedings of National Academy of Science USA 95: 7544–7549.
- 98. van der Woude, M.W., and A. J. Baumler. 2004. Phase and antigenic variation in bacteria. Clinical Microbiology Review 17: 581–611.
- Walport, M. J. 2001. Complement. Second of two parts. New England Journal of Medicine 344: 1140–1144.
- 100. Walport, M. J. 2001. Complement. First of two parts. New England Journal Medicine 344: 1058–1066.
- 101. Winter, J. A., O. R. Davies, A. P. Brown, M. C. Garnett, S. Stolnik and D. I. Pritchard. 2005. The assessment of hookworm calreticulin as a potential vaccine for necatoriasis. Parasite Immunology 27: 139-146.
- 102. Yong, W. K., and P. K. Das. 1983. Acquisition of host proteins by the tegument of *Schistosoma mansoni* recovered from rats. Zeitschrift fur Parasitenkunde. 69: 53–60.

- 103. Zhang, Z., J. Yang, J. Wei, Y. Yang, X. Chen, X. Zhao, Y. Gu, S. Cui, and X. Zhu. 2011. *Trichinella spiralis* paramyosin binds to C8 and C9 and protects the tissue-dwelling nematode from being attacked by host complement. PLoS Neglected Tropical Disease 5: e1225.
- 104. Zipfel, P. F., R. Wurzner, and C. Skerka. 2007. Complement evasion of pathogens: common strategies are shared by diverse organisms. Molecular Immunology 44: 3850–3857.